K13034 September 05, 2013

Center for Devices and Radiological Health

Laser Lipo Ltd Heath House, Crockham Hill Edenbridge United Kingdom, TN8 6ST Telephone: +44 844 980 1820 e-mail: ian@strawberry-laser.com



510[k] Summary

as required by section 807.92(c)

Owner's Name SEP 0.6 2013 Laser Lipo Ltd Address: Heath House Crockham Hill Edenbridge Kent TN8 6ST United Kingdom Tel: 011 44 844 980 1820 Mobile: 011 44 777 445 9611 Fax: 011 44 1732 866 231 Contact Person: Ian Cobley at Laser Lipo Ltd Tel: 011 44 844 980 1820 Mobile: 011 44 777 445 9611 Fax: 011 44 844 980 1820 E-mail: Т ian@strawberry-laser.com Date this summary was prepared: January 31, 2013

Classification name: Low Level laser system for aesthetic use, 21 CFR 878.5400 79 OLI

Common/Usual Name: Low Level laser system for aesthetic use

Proprietary Name: Laser Lipo Ltd will manufacture two devices:

- The Strawberry low level laser system model ILO, and,
- The Strawberry & Cream low level laser system model SC

Establishment Registration Number:

The Strawberry low level laser system and Strawberry & Cream low level laser system will be manufactured by: Laser Lipo Ltd

Heath House Crockham Hill Edenbridge Kent TN8 6ST United Kingdom 011 44 844 980 1820 011 44 844 980 1820

Establishment Registration Number: To be applied for following clearance of this submission.

Substantial Equivalence: The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is substantially equivalent in design, use and materials to the: Chromogenex Technologies Ltd I-Lipo System – K111501

• They are made of the similar materials

Telephone:

Fax:

- They have similar indications for use that are achieved through the same technology
- They are assembled from similar components
- They have similar user input and output interfaces and
- Have similar features.

In particular both the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd i-Lipo System:

- use laser diodes
 - o of approximately 660 nm wavelength, and
 - o less than 50 mW output
- incorporated into
 - multiple multi-diodes "paddles" attached to the subject with special straps
 - Strawberry 4 to 10 paddles with 6 diodes each i-Lipo 4 paddles with 9 diodes each
 - and both have two probes with one diode each
- have a similar control unit
 - LCD subject display
 - o membrane key pad inputs

- incorporated into display as a touch screen for strawberry and cream
 o microprocessor controlled through software
 - multi range sever supply (Ell and US voltages and f
 - multi-range power supply (EU and US voltages and frequencies)
- have a similar range of connecting cables
- have the same laser protection goggles
 - o different name printed on outside of frame

Description of Product:

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system consists of a control unit, various connection leads up to 10 paddles, 2 probes, various "Velcro" attachment straps and other accessories. (A full list of system contents is included at the end of this section.)

A paddle is a device containing six cold red laser emitting diodes which is designed to be placed on the skin. The system can operate using 4, 6, 8 or 10 paddles that are connected to the control unit.

A probe is a device containing one cold red laser emitting diode which is designed to be placed on the skin to treat specific smaller areas of fat, where a usual flat paddle won't ergonomically fit.

The control unit is an electrically powered unit (100-240v, 50-60Hz auto-ranging), enabled by a main switch and key switch. Once enabled it is controlled using a button (Strawberry) or touchscreen (Strawberry and Cream) interface. The output of the diode (six per paddle) is limited to 40mW + -15% by a power limiting PCB from the central processing unit.

When the laser paddles are placed on the skin, the cold red laser beams penetrate the skin just deep enough to reach the layers of fat. When the light hits the fat cells, a rapid chain of events takes place. Firstly, pores form on the cells causing them to spill out. The water, Glycerol and fatty acids move into the interstitial space beneath the fatty layer in the skin. Then further water, fatty acids and Glycerol spill out. The adipocyte cells are therefore reduced in size.

Intended use:

The Low level Laser model Strawberry and Strawberry and Cream can be used for the non-invasive temporary reduction in waist circumference by the disruption of adipocyte cells within the fat layer for the release of fat and lipids from these cells for non-invasive aesthetic use.

Performance Data:

Laser wavelength and output has been demonstrated to be capable and substantially equivalent to the predicate device (Chromogenex i-Lipo, K 111051).

Test	Méan	Quantity	Sample) Standard Deviation	Capability) (Cp
Laser diode power output: Paddle Laser diode output is set to 40mw +/-15%.	39.35	1140	0.81	2.47
Probe Laser diode output is set to 40mw +/-15%.	39.02	40	0.72	3.49
EU portable appliance testing	100% pass	20	-	
 Visual Inspection: external components, cables, outer casings and the General condition of all associated parts. 	100% pass	20		100% inspection
Assembly tests	100% pass	20		100% inspection
Functionality tests	100% pass	20		100% inspection

Double blind clinical study: Laser Lipo Ltd, at the request of the CDRH, had a full double blind clinical study carried out using a total of 35 subjects. Of these, 22 subjects received active laser treatments, and the remaining 13 subjects received placebo treatments. (See section 20.)

Conclusions: At the conclusion of the study, it was clear that 95% of the actively treated subjects achieved, or met, the success criteria. The criteria was set as achieving a temporary reduction around the subjects abdomen (at the height of the iliac crest) of 1.6in (4cm). The average recorded loss was 3.68in (9.35cm) with the greatest loss at 5.6in (14.22cm). Out of the 22 treated subject one failed to met the criteria with a reduction of only 1.2in (3cm). Thus proving the efficacy of the Strawberry and Strawberry & Cream inch loss device.

Not one of the placebo subjects achieved more than 1.3in (3.3cm).

No adverse effects were experienced by any of the trial subjects.



DEPARTMENT OF HEALTH & HUMAN SERVICES

Public Health Service

Food and Drug Administration 10903 New Hampshire Avenue Document Control Center – WO66-G609 Silver Spring, MD 20993-0002

September 6, 2013

Laser Lipo Ltd Mr. Ian Cobley Heath House, Crockham Hill Edenbridge, Kent TN8 6ST United Kingdom

Re: K130341

Trade/Device Name: Laser Lipo Ltd Strawberry Low Level Laser system and Strawberry and Cream Low level Laser system Regulation Number: 21 CFR 878.4810 Regulation Name: Laser surgical instrument for use in general and plastic surgery and in dermatology Regulatory Class: Class II Product Code: OLI Dated: July 26, 2013 Received: July 29, 2019

Dear Mr. Cobley:

We have reviewed your Section 510(k) premarket notification of intent to market the device referenced above and have determined the device is substantially equivalent (for the indications for use stated in the enclosure) to legally marketed predicate devices marketed in interstate commerce prior to May 28, 1976, the enactment date of the Medical Device Amendments, or to devices that have been reclassified in accordance with the provisions of the Federal Food, Drug, and Cosmetic Act (Act) that do not require approval of a premarket approval application (PMA). You may, therefore, market the device, subject to the general controls provisions of the Act. The general controls provisions of the Act include requirements for annual registration, listing of devices, good manufacturing practice, labeling, and prohibitions against misbranding and adulteration. Please note: CDRH does not evaluate information related to contract liability warranties. We remind you; however, that device labeling must be truthful and not misleading.

If your device is classified (see above) into either class II (Special Controls) or class III (PMA), it may be subject to additional controls. Existing major regulations affecting your device can be found in the Code of Federal Regulations, Title 21, Parts 800 to 898. In addition, FDA may publish further announcements concerning your device in the Federal Register.

Please be advised that FDA's issuance of a substantial equivalence determination does not mean that FDA has made a determination that your device complies with other requirements of the Act or any Federal statutes and regulations administered by other Federal agencies. You must Page 2 – Mr. Ian Cobley

comply with all the Act's requirements, including, but not limited to: registration and listing (21 CFR Part 807); labeling (21 CFR Part 801); medical device reporting (reporting of medical device-related adverse events) (21 CFR 803); good manufacturing practice requirements as set forth in the quality systems (QS) regulation (21 CFR Part 820); and if applicable, the electronic product radiation control provisions (Sections 531-542 of the Act); 21 CFR 1000-1050.

If you desire specific advice for your device on our labeling regulation (21 CFR Part 801), please contact the Division of Small Manufacturers, International and Consumer Assistance at its toll-free number (800) 638-2041 or (301) 796-7100 or at its Internet address

http://www.fda.gov/MedicalDevices/ResourcesforYou/Industry/default.htm. Also, please note the regulation entitled, "Misbranding by reference to premarket notification" (21CFR Part 807.97). For questions regarding the reporting of adverse events under the MDR regulation (21 CFR Part 803), please go to

http://www.fda.gov/MedicalDevices/Safety/ReportaProblem/default.htm for the CDRH's Office of Surveillance and Biometrics/Division of Postmarket Surveillance.

You may obtain other general information on your responsibilities under the Act from the Division of Small Manufacturers, International and Consumer Assistance at its toll-free number (800) 638-2041 or (301) 796-7100 or at its Internet address <u>http://www.fda.gov/MedicalDevices/ResourcesforYou/Industry/default.htm</u>.

Sincerely yours,

Mark N. Melkerson -S

Mark N. Melkerson Acting Director Division of Surgical Devices Office of Device Evaluation Center for Devices and Radiological Health

Enclosure

Indications for Use

510(k) Number (if known): K130341

Device Name: Laser Lipo Ltd Strawberry Low Level Laser system and Strawberry and Cream Low level Laser system

Indications for Use: The Low level Laser model Strawberry and Strawberry and Cream can be used for the non-invasive temporary reduction in waist circumference by the disruption of adipocyte cells within the fat layer for the release of fat and lipids from these cells for non-invasive aesthetic use.

Prescription Use ____X___

AND/OR

Over-The-Counter Use

(Part 21 CFR 801 Subpart D)

(21 CFR 801 Subpart C)

(PLEASE DO NOT WRITE BELOW THIS LINE-CONTINUE ON ANOTHER PAGE IF NEEDED)

Concurrence of CDRH, Office of Device Evaluation (ODE)

Neil R Ogden 2013.09.06 11:33:42 -04'00'

(Division Sign-Off) for MXM

Division of Surgical Devices

510(k) Number <u>__K130341</u>



ť,

DEPARTMENT OF HEALTH & HUMAN SERVICES

Public Health Service

Food and Drug Administration 10903 New Hampshire Avenue Document Control Center – WO66-G609 Silver Spring, MD 20993-0002

1

September 6, 2013

Laser Lipo Ltd Mr. Ian Cobley Heath House, Crockham Hill Edenbridge, Kent TN8 6ST United Kingdom

Re: K130341

Trade/Device Name: Laser Lipo Ltd Strawberry Low Level Laser system and Strawberry and Cream Low level Laser system Regulation Number: 21 CFR 878.4810 Regulation Name: Laser surgical instrument for use in general and plastic surgery and in dermatology Regulatory Class: Class II Product Code: OLI Dated: July 26, 2013 Received: July 29, 2019

Dear Mr. Cobley:

We have reviewed your Section 510(k) premarket notification of intent to market the device referenced above and have determined the device is substantially equivalent (for the indications for use stated in the enclosure) to legally marketed predicate devices marketed in interstate commerce prior to May 28, 1976, the enactment date of the Medical Device Amendments, or to devices that have been reclassified in accordance with the provisions of the Federal Food, Drug, and Cosmetic Act (Act) that do not require approval of a premarket approval application (PMA). You may, therefore, market the device, subject to the general controls provisions of the Act. The general controls provisions of the Act include requirements for annual registration, listing of devices, good manufacturing practice, labeling, and prohibitions against misbranding and adulteration. Please note: CDRH does not evaluate information related to contract liability warranties. We remind you; however, that device labeling must be truthful and not misleading.

If your device is classified (see above) into either class II (Special Controls) or class III (PMA), it may be subject to additional controls. Existing major regulations affecting your device can be found in the Code of Federal Regulations, Title 21, Parts 800 to 898. In addition, FDA may publish further announcements concerning your device in the Federal Register.

Please be advised that FDA's issuance of a substantial equivalence determination does not mean that FDA has made a determination that your device complies with other requirements of the Act or any Federal statutes and regulations administered by other Federal agencies. You must

Page 2 – Mr. Ian Cobley

comply with all the Act's requirements, including, but not limited to: registration and listing (21 CFR Part 807); labeling (21 CFR Part 801); medical device reporting (reporting of medical device-related adverse events) (21 CFR 803); good manufacturing practice requirements as set forth in the quality systems (QS) regulation (21 CFR Part 820); and if applicable, the electronic product radiation control provisions (Sections 531-542 of the Act); 21 CFR 1000-1050.

If you desire specific advice for your device on our labeling regulation (21 CFR Part 801), please contact the Division of Small Manufacturers, International and Consumer Assistance at its toll-free number (800) 638-2041 or (301) 796-7100 or at its Internet address

http://www.fda.gov/MedicalDevices/ResourcesforYou/Industry/default.htm. Also, please note the regulation entitled, "Misbranding by reference to premarket notification" (21CFR Part 807.97). For questions regarding the reporting of adverse events under the MDR regulation (21 CFR Part 803), please go to

http://www.fda.gov/MedicalDevices/Safety/ReportaProblem/default.htm for the CDRH's Office of Surveillance and Biometrics/Division of Postmarket Surveillance.

You may obtain other general information on your responsibilities under the Act from the Division of Small Manufacturers, International and Consumer Assistance at its toll-free number (800) 638-2041 or (301) 796-7100 or at its Internet address http://www.fda.gov/MedicalDevices/ResourcesforYou/Industry/default.htm.

Sincerely yours,

Mark N. Melkerson -S

Mark N. Melkerson Acting Director Division of Surgical Devices Office of Device Evaluation Center for Devices and Radiological Health

Enclosure

Page 3 – Mr. Ian Cobley

Concurrence & Template History Page [THIS PAGE IS INCLUDED IN IMAGE COPY ONLY]

Full Submission Number:

For Office of Compliance Contact Information:

http://insideportlets.fda.gov:9010/portal/page?_pageid=197.415881&_dad=portal&_schema=PORTAL&org=318

For Office of Surveillance and Biometrics Contact Information:

http://insideportlets.fda.gov:9010/portal/page?_pageid=197.415881&_dad=portal&_schema=PORTA1.&org=423

Digital	Signature Concurrence Table
Reviewer Sign-Off	Sankar P. Basu
Branch Chief Sign-Off	Neil R.P. Ogden
Division Sign-Off	Mark N. Melkerson -S 2013.09.06 15:57:32 -04'00'

f/t: SPB:dlm:9/6/13

Template Name: K1(A) - SE after 1996

Template History:

Date of Update	By	Description of Update
7/27/09	Brandi Stuart	Added Updates to Boiler Table
8/7/09	Brandi Stuart	Updated HFZ Table
1/11/10	Diane Garcia	Liability/Warranty sentence added at bottom of 1 st page
10/4/11	M. McCabe Janicki	Removed IFU sheet and placed in Forms
9/25/12	Edwena Jones	Added digital signature format
12/12/12	M. McCabe Janicki	Added an extra line between letter signature block and the word "Enclosure". Also, added a missing digit in 4-digit extension on letterhead zip code: "002" should be "0002".
4/2/2013	M. McCabe Janicki	Edited sentence that starts "If you desire specific advice for your device on our labeling regulation (21 CFR Part 801)" Replaced broken Compliance link with general link to DSMICA.
4/12/2013	Margaret McCabe Janicki	Fixed a typo: Paragraph 1, final sentence, "We remind you, however; that device labeling must be truthful" Replaced incorrect semicolon with a comma.

ſ,

Indications for Use

510(k) Number (if known): K130341

Device Name: Laser Lipo Ltd Strawberry Low Level Laser system and Strawberry and Cream Low level Laser system

Indications for Use: The Low level Laser model Strawberry and Strawberry and Cream can be used for the non-invasive temporary reduction in waist circumference by the disruption of adipocyte cells within the fat layer for the release of fat and lipids from these cells for non-invasive aesthetic use.

Prescription Use X

AND/OR

Over-The-Counter Use _____

(Part 21 CFR 801 Subpart D)

(21 CFR 801 Subpart C)

(PLEASE DO NOT WRITE BELOW THIS LINE-CONTINUE ON ANOTHER PAGE IF NEEDED)

Concurrence of CDRH, Office of Device Evaluation (ODE)

Neil R Ogden 2013.09.06 11:33:42 -04'00'

(Division Sign-Off) for MXM

Division of Surgical Devices

510(k) Number __K130341 _____

February 15, 2013

3. 510(k) Cover Letter

K130341

FDA CDRH DMC

Center for Devices and Radiological Health

FEB 1 9 2013

February 15, 2013

Received

Laser Lipo Ltd Heath House Crockham Hill Edenbridge United Kingdom, TN8 6ST Telephone: +44 844 980 1820 e-mail: ian@strawberry-laser.com



February 15, 2013.

Food and Drug Administration Center for Devices and Radiological Health Document Mail Center - WO66-G609 10903 New Hampshire Avenue Silver Spring Maryland 20993-0002 United States of America

Ref: 510(k) Premarket Notification

Dear Document Control Clerk:

Pursuant to the requirements of section 510 (k) of the Food, Drug and Cosmetic Act, notification is made to manufacture and market the following new medical device:

Classification Name:

Low Level laser system for aesthetic use 79 OLI

Classification Regulation: 21 CFR 878.5400

Common/Usual Name:

Low level laser system for aesthetic use

Page 3.2

February 15, 2013

Proprietary Name:

Laser Lipo Ltd will manufacture two devices:

- the Strawberry low level laser system model ILO, and,
- the Strawberry & Cream low level laser system model SC

Establishment Registration Number:

The Strawberry low level laser system and Strawberry & Cream low level laser system will be manufactured by:

Laser Lipo Ltd Heath House Crockham Hill Edenbridge Kent TN8 6ST United Kingdom Telephone: 011 44 844 980 1820 Fax: 011 44 844 980 1820 Establishment Registration Number: To be applied for following clearance of this submission.

Classification:

The FDA has classified: Low level laser system for aesthetic use (79 OLI) as a Class II medical device under 21 CFR 878.5400.

Performance Standards:

None established under 514, however FDA has published "Class II Special Controls Guidance Document: Low Level Laser System for Aesthetic Use".

The following international consensus standards, recognized by FDA, have been complied with:

IEC 62304:2006 – software life-cycle IEC 60601-1:2006 – electrical safety IEC 60601-1-2:2007 - electromagnetic compatibility IEC 60601-2-22:1996 – laser products IEC 60825-1:2007 – laser products ISO 14971:2009 – risk management ISO 10993-1:2009 - biocompatibility

The following additional standards have also been complied with, although not recognized by the FDA:

ISO 13485: 2003 – quality management system EN 980:2008 – European symbols (explanatory text included in US labeling) EN 1041:2008 – requirements for information (supplemented by FDA guidance listed below)

The following Guidance notes have been used extensively in the preparation of this submission:

- Class II Special Controls Guidance Document: Low Level
 Laser System for Aesthetic Use
- Guidance for the Content of Premarket Submissions for Software Contained in Medical Devices
- Format for Traditional and Abbreviated 510(k)s
- Laser Products Conformance with IEC 60825-1 and IEC 60601-2-22

Labeling and Promotional Materials:

Draft labels, labeling (including the User Manual) and promotional materials for the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are contained in Section 13.

Substantial Equivalence:

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is substantially equivalent in design, use and materials to the:

Chromogenex Technologies Ltd I-Lipo System - K111501

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is made of the similar materials as the Chromogenex Technologies Ltd I-Lipo System – K111501.

The Laser Lipo Ltd Strawberry and Strawberry & Cream low level laser systems has similar indications for use

The Low level Laser model Strawberry and Strawberry and Cream can be used for the non-invasive aesthetic treatment for the temporary reduction in waist circumference.

to the Chromogenex Technologies Ltd I-Lipo System.

February 15, 2013

The Chromogenex Technologies Limited i-lipoTm Low Level Laser System is indicated for non-invasive aesthetic treatment for the temporary reduction in circumference of the waist.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is made of the same materials as the Chromogenex Technologies Ltd I-Lipo System – K111501.

- thermoplastic control unit enclosure
 - proprietary components and assemblies used within the unit
- thermoplastic paddles and probes
- nylon covered "velcro" fixed straps
- nylon/polycarbonate goggles

Both the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd i-Lipo System:

- use laser diodes
 - \circ of approximately 660 nm wavelength, and
 - less than 50 mW output
- incorporated into
 - various laser diode cluster probes and paddles attached to the subject with special straps
 Strawberry 4 to 10 paddles with 6 diodes each
 i-Lipo 4 paddles with 9 diodes each
 - I-LIPO 4 paddies with and diada and
 - \circ and two probes with one diode each
- have a similar control unit
 - LCD subject display
 - o membrane key pad inputs
 - incorporated into display as a touch screen for strawberry and cream
 - o microprocessor controlled through software
 - multi-range power supply (EU and US voltages and frequencies)
- have a similar range of connecting cables
- have the same two pairs of laser protection goggles
 - different name printed on outside of frame

The Laser Lipo Ltd Strawberry low level laser system, Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd I-Lipo System have the same use achieved through the same technology, are assembled from similar components made from similar materials, have similar user input and output interfaces and have similar features.

All systems use laser emitting diodes of the similar wavelengths and powers: the Laser Lipo systems have the potential to use more paddles but with less

diodes in each paddle. All systems have two cluster probes with a single laser diode in each. In all systems the paddles and cluster probes are secured to the subject with nylon encased rubber straps secured with "velcro".

All control units have plastic enclosures, discrete power supply, microprocessor controlled electronics, controls, alarms and similar control software.

Laser Lipo Ltd concluded that the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are substantially equivalent to the Chromogenex Technologies Ltd I-Lipo System, K111501

For a demonstration of equivalence, please see Section 12.

Promotional Material giving details of the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and Chromogenex Technologies Ltd I-Lipo System is compared in Section 12.

Previous Submission:

Laser Lipo Limited have previously submitted the Strawberry and Strawberry & Cream Low Level Laser system under K122354 and PRE Submission Q120359

Description of Product:

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system consists of a control unit, various connection leads up to 10 paddles, 2 cluster probes, various "Velcro" attachment straps and other accessories. (A full list of system contents is included at the end of this section.)

A paddle is a device containing six cold red laser emitting diodes which is designed to be placed on the skin. The system can operate using 4, 6, 8 or 10 paddles that are connected to the control unit.

A probe is a device containing one cold red laser emitting diode which is designed to be placed on the skin to treat specific smaller areas of fat, where a usual flat paddle won't ergonomically fit.

The control unit is an electrically powered unit (100-240v, 50-60Hz autoranging), enabled by a main switch and key switch. Once enabled it is controlled using a button (Strawberry) or touch screen (Strawberry and Cream) interface. The output of the diode (six per paddle) is limited to 40mW +/- 15% by a power limiting PCB from the central processing unit.

When the laser paddles are placed on the skin, the cold red laser beams penetrate the skin just deep enough to reach the layers of fat. When the light hits the fat cells, a rapid chain of events takes place. Firstly, pores form on the cells causing them to spill out. The water, Glycerol and fatty acids move into the interstitial space beneath the fatty layer in the skin. Then further water, fatty acids and Glycerol spill out. The adipocyte cells are therefore reduced in size.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system contain the same control electronics and control software however the Strawberry and Cream has a menu driven touch screen interface while the Strawberry model has a liquid crystal display and (up, down, left, right and center/enter) push buttons. The outputs and functionality of the Strawberry and Strawberry and Cream models are identical, however displays, display drivers, interface electronics and software and power supply transformed are different. Full details are contained in Section 11.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is shipped in dedicated packaging to prevent damage and protected by a cardboard box.

The system as shipped contains:

- o 1 low level laser system unit, either
 - Strawberry model, or
 - Strawberry & Cream model
- o 1 user manual
- o 1 power lead
- Paddle leads, for example for a 10 paddle system
 - 1 long paddle lead for connection to the device
 - 8 paddle standard leads
 - 1 long paddle lead
- o 2 cluster Probes
- 1 Pair of goggles
- 2 Replacement fuses
- o 2 Keys

variants of the system w	The available with 4, 0, 0 and 10 paddles.
number of paddles	number of paddle leads
3 standard and 1 end	3 standard
5 standard and 1 end	5 standard
7 standard and 1 end	7 standard
9 standard and 1 end	8 standard and 1 long (to connect two groups of five paddles)
Table 1 - Variants	

Variants of the system will be available with 4, 6, 8 and 10 paddles.

A full list of the proposed variants and a schematic drawing showing the key design features are included in Section 11.

Engineering Drawings and Photographs of the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system can be found in Section 11.

Photographs, photo drawings and block diagrams showing the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd I-Lipo System are contained in Section 11 and comparison to the predicate Chromogenex Technologies Ltd I-Lipo System can be seen in Section 12.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system will be packaged in a cardboard box together with all components (instructions booklet, cables, paddles), software and accessories (diodes, goggles, fuse, straps and lead) within the scope of the 510(k). A shelf-life has been determined for 5 years based on the laser diodes.

Electrical Safety and Electromagnetic Compatibility is demonstrated through compliance with IEC 60601 part 1, part 1-2 and part 2-22 full test reports from a third party laboratory are included in Section 17.

The Guidance also states that there should be compliance with the IEC 60601-1-4 standard. Laser Lipo Ltd believes that compliance with IEC 62304:2006 life-cycle standard achieves at least the same level of assurance in the quality of the software used in the device.

A usability file detailing application of IEC 60601-1-6:2008 is contained in Section 11.

Table 2 - Product Release Testing Summary

Gives the results of bench testing of the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system which can be seen in section 18 and is confirmed by clinical reports in section 20.

Test	Mean	Quantity	Sample Standard Deviation	Capability, Cp
Laser diode power output: Paddle Laser diode output is set to 40mw +/-15%.	39.35	1140	0.81	2.47
Probe Laser diode output is set to $40 \text{mw} + -15\%$.	40.02	40	0.72	3.49
EU portable appliance testing	100% pass	20		
 Visual Inspection: external components, cables, outer casings and the General condition of all associated parts. 	100% pass	20		100% inspection
Assembly tests	100% pass	20		100% inspection
Functionality tests	100% pass	20		100% inspection

Table 2 - Product Release Testing Summary

Biocompatibility:

Biocompatibility assessment on the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system have been conducted following the ISO 10993-1 (FDA Recognized Consensus Standard), Biological Evaluation of Medical Devices, Part 1: Evaluation and Testing. Please see section 15.

Sterilization:

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are supplied non-sterile.

Cleaning instructions are contained in the user manual included with each system.

Safety and Effectiveness:

A summary of safety and effectiveness information ("510(k) Summary" per 21 CFR 807.92) upon which an equivalence determination has been made is included as Section 5.

Indications for Use Statement:

A single sheet statement of "indications for use" as required by FDA for submissions after January 01, 1996, is included in Section 4.

Truthful and Accurate Statement:

The premarket notification "truthful and accurate" statement (per 21 CFR 807.87(k)), signed by the Group Director Ian Cobley of Laser Lipo Ltd is included as Section 6.

E Copy Statement:

The E Copy provided within our submission is an exact duplicate of the paper copy.

Owner:

This Pre-Market Notification is owned by Laser Lipo Ltd.

February 15, 2013

Center for Devices and Radiological Health

Submitter:

This Pre-Market Notification is submitted by Ian Cobley Group Director For and on behalf of: Laser Lipo Ltd

Contact Persons:

The address and telephone number for **<u>all</u>** correspondence is:

Ian Cobley Laser Lipo Ltd Heath House Crockham Hill Edenbridge Kent TN8 6ST United Kingdom Tel: 011 44 844 980 1820 Mobile: 011 44 777 445 9611 Fax: 011 44 844 980 1820 e-mail: ian@strawberry-laser.com

As per your letter dated February 14, 2013 please find a replacement E copy of our submission.

If I can be of any further assistance, please do not hesitate to contact me.

Sincerely,

Ian Cobley Group Director Laser Lipo Ltd

1. Medical Device User Fee Cover Sheet (Form FDA 3601)

Form Approved: OMB No. 0910-511 Expiration Date: February 28, 2013. See Instructions for OMB Statement.

DEPARTMENT OF HEALTH AND HUMAN SERVICES FOOD AND DRUG ADMINISTRATION MEDICAL DEVICE USER FEE COVER SHEET	PAYMENT IDENTIFICATION NUMBER: (b)(4) Trade Secret Write the Payment Identification number on Statesson, Product
A completed cover sheet must accompany each original application courier, please include a copy of this completed form with payment. http://www.fda.gov/oc/mdufma/coversheet.html	or supplement subject to fees. If payment is sent by U.S. mail or Payment and mailing instructions can be found at:
 COMPANY NAME AND ADDRESS (include name, street address, city state, country, and post office code) 	2. CONTACT NAME lan Cobley
	2.1 E-MAIL ADDRESS
LASER LIPOLID Heath House	ian@strawberry-laser.com
Crockham Hill	2.2 TELEPHONE NUMBER (Include Area code)
Edenbridge Kent TN8 6ST	+44844-9801820
1.1 EMPLOYER IDENTIFICATION NUMBER (EIN)	2.3 FACSIMILE (FAX) NUMBER (Include Area code)
3. TYPE OF PREMARKET APPLICATION (Select one of the follow	ing in each column; if you are unsure, please refer to the application
descriptions at the following web site: http://www.fda.gov/oc/mdufma	
Select an application type:	3.1 Select a center
[X] Premarket notification(510(k)); except for third party	
[] 513(g) Request for Information	2.2. Select and of the types below
[] Biologics License Application (BLA)	5.2 Select one of the types below
[] Premarket Approval Application (PMA)	
[] Modular PMA	Li Efficacy (BLA)
[] Product Development Protocol (PDP)	[] Panel Track (PMA_PMR_PDP)
[] Premarket Report (PMR)	[] Real-Time (PMA_PMR_PDP)
[] Annual Fee for Periodic Reporting (AFR)	[1180-day (PMA, PMR, PDP)
[] SU-Day Notice	() (or copy () and () and () and () and ()
 4. ARE YOU A SMALL BUSINESS? (See the instructions for more [] YES, I meet the small business criteria and have submitted the re qualifying documents to FDA 4.1 If Yes, please enter your Small Business Decision Number: 	equired [X] NO, I am not a small business
5. FDA WILL NOT ACCEPT YOUR SUBMISSION IF YOUR COMP THAT IS DUE TO FDA. HAS YOUR COMPANY PAID ALL ESTABL	ANY HAS NOT PAID AN ESTABLISHMENT REGISTRATION FEE ISHMENT REGISTRATION FEES THAT ARE DUE TO FDA?
MYES (All of our establishments have registered and paid the fee, 30 days of FDA's approval/clearance of this device.)	or this is our first device, and we will register and pay the fee within
[] NO (If "NO," FDA will not accept your submission until you have http://www.fda.gov/cdrh/mdufma for additional information)	paid all fees due to FDA. This submission will not be processed; see
6. IS THIS PREMARKET APPLICATION COVERED BY ANY OF T APPLICABLE EXCEPTION.	THE FOLLOWING USER FEE EXCEPTIONS? IF SO, CHECK THE
[] This application is the first PMA submitted by a qualified small buincluding any affiliates	usiness, [] The sole purpose of the application is to support conditions of use for a pediatric population
[] This biologics application is submitted under section 351 of the F Health Service Act for a product licensed for further manufacturing i	Public government entity for a device that is not to be distributed commercially
7. IS THIS A SUPPLEMENT TO A PREMARKET APPLICATION F PEDIATRIC POPULATION THAT NOW PROPOSES CONDITION subject to the fee that applies for an original premarket approval ap	OR WHICH FEES WERE WAIVED DUE TO SOLE USE IN A OF USE FOR ANY ADULT POPULATION? (If so, the application is plication (PMA).
PAPERWORK REDUCTION ACT STATEMENT Public reporting burden for this collection of information is estimated instructions, searching existing data sources, gathering and maintai information. Send comments regarding this burden estimate or any reducing this burden, to the address below.	d to average 18 minutes per response, including the time for reviewing ining the data needed, and completing and reviewing the collection of other aspect of this collection of information, including suggestions for
Department of Health and Human Services, Food and Drug Admini Floor Rockville, MD 20850	stration, Office of Chief Information Officer, 1350 Piccard Drive, 4th
[Please do NOT return this form to the above address, except as it	pendins to comments on the burden estimate.j
8. USER FEE PAYMENT AMOUNT SUBMITTED FOR THIS PREI	MARKET APPLICATION 14-Dec-2012

Form FDA 3601 (01/2007)

2. CDRH Premarket Review Submission Cover Sheet

	DEPARTMENT OF HEALTH AN FOOD AND DRUG AD	ND HUMAN SERV	ICES Form Approv OMB No. 901 Expiration Da			val 110-0120 ate: December 31, 2013	
CDRH PRE	MARKET REVIEW S	UBMISSION	COVER	SHEET		See OMB St	tatement on page 5.
Date of Submission 17 December 2	.012 User Fee Payment ID	Number t Process - Pro	duct Specs	FDA S	Submissio	on Document	t Number (if known)
SECTION A PMA Original Submission Premarket Report Modular Submission Amendment Report Report Amendment Licensing Agreement	PMA & HDE Supplement Regular (180 day) Special Panel Track (PMA Only) 30-day Supplement 30-day Notice 135-day Supplement Real-time Review Amendment to PMA &HDE Supplement Other	TYPE OF S PDP Original PDP Notice of Cor	UBMISSIO npletion to PDP	ON 510(k) Original Submiss Traditional Special Abbreviated section I, Pa Additional Inform Third Party		ion: Complete le 5) ation	Meeting Pre-510(K) Meeting Pre-IDE Meeting Pre-PMA Meeting Pre-PDP Meeting Day 100 Meeting Agreement Meeting Determination Meeting Other (specify):
IDE Original Submission Amendment Supplement	Humanitarian Device Exemption (HDE) Original Submission Amendment Supplement Report Report Report Amendment	Class II Exempt	ion Petition nission ormation	Evaluation of Automatic Class III Designation (De Novo) Original Submission Additional Information		itomatic nation) ion ation	Other Submission 513(g) Other (describe submission):
Have you used or cited Sta	ndards in your submission?	Yes [No (If	Yes, please con	nplete S	ection I, Pag	e 5)
SECTION B Company / Institution Name Laser Lipo Limit	sub ed	MITTER, APPLI	CANT OR S Establishme To be ap	PONSOR ent Registration oplied for fo	Number	r (if known) ng clearai	nce of this 510[k]
Division Name (if applicable)			Phone Number (including area code) 011 44 844 980 1820				
Street Address Heath House, C	rockham Hill		FAX Number (including area code) 011 44 844 980 1820				
_{City} Edenbridge			State / ProvinceZIP/PostalKentTN8 6		al Code 6ST	Country U.K.	
Contact Name Ian Cobley							
Contact Title Group Director			Contact E-mail Address ian@strawberry-laser.com				
SECTION C Company / Institution Name		ESPONDENT (e.	g., consulta Establishme	ant, if differer ent Registration	nt from n Numbe	above) er (if known)	
Division Name (if applicable)			Phone Number (including area code)				
Street Address			FAX Number (including area code)				
City			State / Provi	nce 2	ZIP/Posta	al Code	Country
Contact Name			<u> </u>	I			1
Contact Title			Contact E-m	ail Address			

SECTION D1 RE	ASON FOR APPLICATION - PMA, PDP, OR H	IDE
 New Device Withdrawal Additional or Expanded Indications Request for Extension Post-approval Study Protocol Request for Applicant Hold Request for Removal of Applicant Hold Request to Remove or Add Manufacturing Site 	 Change in design, component, or specification: Software / Hardware Color Additive Material Specifications Other (specify below) 	Location change: Manufacturer Sterilizer Packager
Process change: Manufacturing Sterilization Packaging Other (specify below) Response to FDA correspondence:	Labeling change: Indications Instructions Performance Shelf Life Trade Name Other (specify below)	
Other Reason (specify)		
SECTION D2	REASON FOR APPLICATION - IDE	Repose to FDA Letter Concerning:
 New Indication Addition of Institution Expansion / Extension of Study IRB Certification Termination of Study Withdrawal of Application Unanticipated Adverse Effect Notification of Emergency Use Compassionate Use Request Treatment IDE Continued Access 	Correspondent / Applicant Design / Device Informed Consent Manufacturer Manufacturing Process Protocol - Feasibility Protocol - Other Sponsor Report submission: Current Investigator Annual Progress Report Site Waiver Report Final	 Conditional Approval Deemed Approved Deficient Final Report Deficient Progress Report Deficient Investigator Report Disapproval Request Extension of Time to Respond to FDA Request Meeting Request Hearing
Other Reason (specify):		
SECTION D3	REASON FOR SUBMISSION - 510(k)	
New Device	Additional or Expanded Indications	Change in Technology
Other Reason (specify):	1	

S	ECTION E	to w	/hich	ADD substantial equivale		DNAI is cla	- INFC	ORMATIO	ON ON 51	0(K) SUB	MIS	SSION	IS	9	Summarv of.	or statement concerning.
1	OLI	2			3	3 4							S	afety and ef	fectiveness information		
5		6			7	,		8							∑ 510 (k) :	summary attached	
Int	Information on devices to which substantial equivalence is claimed <i>(if known</i>)																
	510/k) Number Tride or Dreprietens or Medel Name Mensfectures																
	570(A) 14					Trau				vanie					Wan	
1	K111501				1	I-L	_ipo	Syster	m				1	Chro	or	nogene	ex
							·	,						lecr	٦r	lologie	s Ltd
2					2								2				
3					3								3				
4					4					4							
5					5								5				
6					6								6				
6				PRODUCT											0		
S		r els	accifi	PRODUCT	INI		VIATIC	DN - APPI	LICATIO		J ALL	Al	PLIC	ATION	5		
I		r	ssiii	stem for ae	ctl	heti	ic us	e									
-			J y.		50	i cu	c us										
	Trade or Proprietary or	Mo	del	Name for This Devic	e								Mode	Number	r		
1	Strawberry											1	ILC)			
2	Strawberry 8	કે (Cre	eam								2	SC	SC			
3												3					
4												4					
												-					
5												5					
FE 1	DA document numbers of	ofall	pric	or related submission	IS (re	egard	ess of e	outcome)	4				5				6
ĸ	(122354	2			ľ								ľ				0
7		8			9				10				11				12
Da	ata Included in Submissi	on		Laborat	ory	Testin	ng		Animal Tria	s		[Hur	man Trial	s		
S	ECTION G	0		PRODUCT	LA:	SSIF		ION - API	PLICATIO	DN.	TO AL	LA	PPLI	CATIO	NS		
C	Product Code C.F.R. Section (<i>if applicable</i>) Device Class																
CI	assification Panel														Inc	lossified	
G	Seneral and p	as	sti	c surgery d	ev	ices	5					5105	5 111		/IIC	lassilieu	
In	Indications (from labeling)																
Т	The Low level Laser model Strawberry and Strawberry and Cream can be used for the																
n	on-invasive t	en	np	orary reduc	tio	on i	n wa	aist cir	cumfe	re	nce	by	' the	e disr	u	ption o	f adipocyte
С	ells within the	e f	at	layer for th	e	rele	ease	of fat	and li	pi	ds fr	on	n th	ese c	e	lls for ı	non-invasive
а	esthetic use.																

<i>Note:</i> Submission of this or 2891a Device Establis	information does not affect the nee hment Registration form.	ed to submit a 2891	FDA Document Number (if known)					
SECTION H	MANUFACTURING / PACK	AGING / STERILIZ	ATION SITES RELATING TO	A SUBMISSION				
Original	FDA Establishment Registration To be applied for follo clearance of this 510[Number Dwing [k]	Manufacturer Contract Sterilizer Contract Manufacturer Repackager / Relabeler					
Laser Lipo Lin	nited		To be applied for follow	^r ving clearance of this 510[k]				
Division Name (if applica	ble)		Phone Number (including area cod 011 44 844 980 18	^{le)} 20				
Street Address Heath House,	Crockham Hill		FAX Number (including area code) 011 44 1732 866 2	31				
^{City} Edenbridge			State / Province Kent	ZIP/Postal CodeCountryTN8 6STU.K.				
Contact Name Ian Cobley		Contact Title Group Direc	tor	Contact E-mail Address ian@strawberry- laser.com				
	EDA Establishment Registration	Number						
Original	T DA Establionment Registration		Manufacturer	Contract Sterilizer				
Add Delete			Contract Manufacturer	Repackager / Relabeler				
Company / Institution Na	ne		Establishment Registration Number					
Division Name (if applica	ble)		Phone Number (including area code)					
Street Address			FAX Number (including area code)					
City			State / Province	ZIP/Postal Code Country				
Contact Name		Contact Title		Contact E-mail Address				
	EDA Establishment Devictor	Number						
Original	T DA ESTADIISHIITIENT REGISTRATION	NUMBER	Manufacturer	Contract Sterilizer				
Add Delete			Contract Manufacturer	Repackager / Relabeler				
Company / Institution Na	me		Establishment Registration Number					
Division Name (<i>if applicable</i>)			Phone Number (including area code) ()					
Street Address			FAX Number (including area code) ()					
City			State / Province	ZIP/Postal Code Country				
Contact Name		Contact Title		Contact E-mail Address				

SEC	TIONI		UTILIZATION OF STANDARDS							
Note: state	Complete this sectior ment.	n if your application or	submission cites standards or includes a "Declaration of Confo	rmity to a Recognized	Standard"					
	Standards No.	Standards Organization	Standards Title	Version	Date					
1	62304	IEC	Medical Device Software, Software Life-cycle Processes		2006					
	Standards No.	Standards Organization	Standards Title	Version	Date					
2	60601	IEC	Medical Electrical Equipment, General Requirements for Safety	part 1	2006					
	Standards No.	Standards Organization	Standards Title	Version	Date					
3	60601	IEC	Medical Electrical Equipment, General Requirements for safety and essential performance, Collateral standard, Electromagnetic compatibility and tests	parts 1-2	2007					
	Standards No.	Standards Organization	Standards Title	Version	Date					
4	60601	IEC	Medical electrical equipment, Particular Requirements for safety. Specification for Diagnosis and Therapeutic Laser Equipment.	parts 2-22	1996					
	Standards No.	Standards Organization	Standards Title	Version	Date					
5	60825	IEC	Safety of laser products, Equipment classification and Requirements	part 1	2007					
	Standards No.	Standards Organization	Standards Title	Version	Date					
6	14971	ISO	Application of Risk Management to Medical Devices		2009					

Department of Health and Human Services Food and Drug Administration Office of the Chief Information Officer 1350 Piccard Drive, Room 400. Rockville, MD 20850

Biological evaluation of Medical

Devices, Evaluation and Testing

Please include any additional standards to be cited on a separate page.

Public reporting burden for this collection of information is estimated to average 0.5 hour per response, including the time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing reviewing the collection of information. Send comments regarding this burden estimate or any

An agency may not conduct or sponsor, and a person is not required to respond to, a collection of information unless it displays a currently valid OMB control number.

other aspect of this collection of information, including suggestions for reducing this burden to:

Standards

ISO

Organization

Standards Title

Standards No.

10993

7

Date

2009

Version

part 1

UTILIZATION OF STANDARDS

Note state	: Complete this secti ment.	ion if your application	or submission cites standards or includes a "Declaration of C	onformity to a Recogi	nized Standard"		
	Standards No.	Standards Organization	Standards Title	Version	Date		
8	13485	ISO	Quality Management Systems, Requirements for Regulatory Purposes		2003		
9	Standards No.	Standards Organization	Standards Title	Version	Date		
	Standards No.	Standards Organization	Standards Title	Version	Date		
10							
44	Standards No.	Standards Organization	Standards Title	Version	Date		
11							
	Standards No.	Standards Organization	Standards Title	Version	Date		
12							
	Standards No.	Standards Organization	Standards Title	Version	Date		
13							
	Standards No.	Standards Organization	Standards Title	Version	Date		
14							
	l	Please	e include any additional standards to be cited on a separa	ite page.			
Public reporting burden for this collection of information is estimated to average 0.5 hour per response, including the time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing reviewing the collection of information. Send comments regarding this burden estimate or any other aspect of this collection of information, including suggestions for reducing this burden to:							

Department of Health and Human Services Food and Drug Administration Office of the Chief Information Officer 1350 Piccard Drive, Room 400. Rockville, MD 20850

An agency may not conduct or sponsor, and a person is not required to respond to, a collection of information unless it displays a currently valid OMB control number.

3. 510(k) Cover Letter

Laser Lipo Ltd Heath House Crockham Hill Edenbridge United Kingdom, TN8 6ST Telephone: +44 844 980 1820 e-mail: ian@strawberry-laser.com



Thursday, 31 2013.

Food and Drug Administration Center for Devices and Radiological Health Document Mail Center - WO66-G609 10903 New Hampshire Avenue Silver Spring Maryland 20993-0002 United States of America

Ref: 510(k) Premarket Notification

Dear Document Control Clerk:

Pursuant to the requirements of section 510 (k) of the Food, Drug and Cosmetic Act, notification is made to manufacture and market the following new medical device:

Classification Name:

Low Level laser system for aesthetic use 79 OLI

Classification Regulation:

21 CFR 878.5400

Common/Usual Name:

Low level laser system for aesthetic use

Proprietary Name:

Laser Lipo Ltd will manufacture two devices:

- the Strawberry low level laser system model ILO, and,
- the Strawberry & Cream low level laser system model SC

Establishment Registration Number:

The Strawberry low level laser system and Strawberry & Cream low level laser system will be manufactured by:

Laser Lipo Ltd Heath House Crockham Hill Edenbridge Kent TN8 6ST United Kingdom Telephone: 011 44 844 980 1820 Fax: 011 44 844 980 1820 Establishment Registration Number: To be applied for following clearance of this submission.

Classification:

The FDA has classified: Low level laser system for aesthetic use (79 OLI) as a Class II medical device under 21 CFR 878.5400.

Performance Standards:

None established under 514, however FDA has published "Class II Special Controls Guidance Document: Low Level Laser System for Aesthetic Use".

The following international consensus standards, recognized by FDA, have been complied with:

IEC 62304:2006 – software life-cycle IEC 60601-1:2006 – electrical safety IEC 60601-1-2:2007 - electromagnetic compatibility IEC 60601-2-22:1996 – laser products IEC 60825-1:2007 – laser products ISO 14971:2009 – risk management ISO 10993-1:2009 - biocompatibility The following additional standards have also been complied with, although not recognized by the FDA:

ISO 13485: 2003 – quality management system EN 980:2008 – European symbols (explanatory text included in US labeling) EN 1041:2008 – requirements for information (supplemented by FDA guidance listed below)

The following Guidance notes have been used extensively in the preparation of this submission:

- Class II Special Controls Guidance Document: Low Level
 Laser System for Aesthetic Use
- Guidance for the Content of Premarket Submissions for Software Contained in Medical Devices
- Format for Traditional and Abbreviated 510(k)s
- Laser Products Conformance with IEC 60825-1 and IEC 60601-2-22

Labeling and Promotional Materials:

Draft labels, labeling (including the User Manual) and promotional materials for the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are contained in Section 13.

Substantial Equivalence:

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is substantially equivalent in design, use and materials to the:

Chromogenex Technologies Ltd I-Lipo System – K111501

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is made of the similar materials as the Chromogenex Technologies Ltd I-Lipo System – K111501.

The Laser Lipo Ltd Strawberry and Strawberry & Cream low level laser systems has similar indications for use

The Low level Laser model Strawberry and Strawberry and Cream can be used for the non-invasive aesthetic treatment for the temporary reduction in waist circumference.

to the Chromogenex Technologies Ltd I-Lipo System.

The Chromogenex Technologies Limited i-lipoTm Low Level Laser System is indicated for non-invasive aesthetic treatment for the temporary reduction in circumference of the waist.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is made of the same materials as the Chromogenex Technologies Ltd I-Lipo System – K111501.

- thermoplastic control unit enclosure
 - proprietary components and assemblies used within the unit
- thermoplastic paddles and probes
- nylon covered "velcro" fixed straps
- nylon/polycarbonate goggles

Both the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd i-Lipo System:

- use laser diodes
 - \circ of approximately 660 nm wavelength, and
 - less than 50 mW output
- incorporated into
 - various laser diode cluster probes and paddles attached to the subject with special straps Strawberry 4 to 10 paddles with 6 diodes each
 - i-Lipo 4 paddles with 9 diodes each
 - $\circ~$ and two probes with one diode each
- have a similar control unit
 - LCD subject display
 - o membrane key pad inputs
 - incorporated into display as a touch screen for strawberry and cream
 - microprocessor controlled through software
 - multi-range power supply (EU and US voltages and frequencies)
- have a similar range of connecting cables
- have the same two pairs of laser protection goggles
 - o different name printed on outside of frame

The Laser Lipo Ltd Strawberry low level laser system, Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd I-Lipo System have the same use achieved through the same technology, are assembled from similar components made from similar materials, have similar user input and output interfaces and have similar features.

All systems use laser emitting diodes of the similar wavelengths and powers: the Laser Lipo systems have the potential to use more paddles but with less
diodes in each paddle. All systems have two cluster probes with a single laser diode in each. In all systems the paddles and cluster probes are secured to the subject with nylon encased rubber straps secured with "velcro".

All control units have plastic enclosures, discrete power supply, microprocessor controlled electronics, controls, alarms and similar control software.

Laser Lipo Ltd concluded that the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are substantially equivalent to the Chromogenex Technologies Ltd I-Lipo System, K111501

For a demonstration of equivalence, please see Section 12.

Promotional Material giving details of the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and Chromogenex Technologies Ltd I-Lipo System is compared in Section 12.

Previous Submission:

Laser Lipo Limited have previously submitted the Strawberry and Strawberry & Cream Low Level Laser system under K122354 and PRE Submission Q120359

Description of Product:

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system consists of a control unit, various connection leads up to 10 paddles, 2 cluster probes, various "Velcro" attachment straps and other accessories. (A full list of system contents is included at the end of this section.)

A paddle is a device containing six cold red laser emitting diodes which is designed to be placed on the skin. The system can operate using 4, 6, 8 or 10 paddles that are connected to the control unit.

A probe is a device containing one cold red laser emitting diode which is designed to be placed on the skin to treat specific smaller areas of fat, where a usual flat paddle won't ergonomically fit.

The control unit is an electrically powered unit (100-240v, 50-60Hz autoranging), enabled by a main switch and key switch. Once enabled it is controlled using a button (Strawberry) or touchscreen (Strawberry and Cream) interface. The output of the diode (six per paddle) is limited to 40mW +/-15% by a power limiting PCB from the central processing unit.

When the laser paddles are placed on the skin, the cold red laser beams penetrate the skin just deep enough to reach the layers of fat. When the light hits the fat cells, a rapid chain of events takes place. Firstly, pores form on the cells causing them to spill out. The water, Glycerol and fatty acids move into the interstitial space beneath the fatty layer in the skin. Then further water, fatty acids and Glycerol spill out. The adipocyte cells are therefore reduced in size.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system contain the same control electronics and control software however the Strawberry and Cream has a menu driven touch screen interface while the Strawberry model has a liquid crystal display and (up, down, left, right and center/enter) push buttons. The outputs and functionality of the Strawberry and Strawberry and Cream models are identical, however displays, display drivers, interface electronics and software and power supply transformed are different. Full details are contained in Section 11.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is shipped in dedicated packaging to prevent damage and protected by a cardboard box.

The system as shipped contains:

- o 1 low level laser system unit, either
 - Strawberry model, or
 - Strawberry & Cream model
- o 1 user manual
- o 1 power lead
- Paddle leads, for example for a 10 paddle system
 - 1 long paddle lead for connection to the device
 - 8 paddle standard leads
 - 1 long paddle lead
- 2 cluster Probes
- 1 Pair of goggles
- 2 Replacement fuses
- o 2 Keys

variance of the sy	Stelli will be availab	1 with $4, 0, 0$		paulies.
number of paddles	numb	er of paddle lead	S	
3 standard and 1 end	3 stan	dard		
5 standard and 1 end	5 stan	dard		
7 standard and 1 end	7 stan	dard		
9 standard and 1 end	8 stan	dard and 1 long (to	o connect	two groups of five paddles)

Variants of the system will be available with 4, 6, 8 and 10 paddles.

Table 1 - Variants

A full list of the proposed variants and a schematic drawing showing the key design features are included in Section 11.

Engineering Drawings and Photographs of the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system can be found in Section 11.

Photographs, photo drawings and block diagrams showing the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd I-Lipo System are contained in Section 11 and comparison to the predicate Chromogenex Technologies Ltd I-Lipo System can be seen in Section 12.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system will be packaged in a cardboard box together with all components (instructions booklet, cables, paddles), software and accessories (diodes, goggles, fuse, straps and lead) within the scope of the 510(k). A shelf-life has been determined for 5 years based on the laser diodes.

Electrical Safety and Electromagnetic Compatibility is demonstrated through compliance with IEC 60601 part 1, part 1-2 and part 2-22 full test reports from a third party laboratory are included in Section 17.

The Guidance also states that there should be compliance with the IEC 60601-1-4 standard. Laser Lipo Ltd believes that compliance with IEC 62304:2006 life-cycle standard achieves at least the same level of assurance in the quality of the software used in the device.

A usability file detailing application of IEC 60601-1-6:2008 is contained in Section 11.

Table 2 - Product Release Testing Summary

Gives the results of bench testing of the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system which can be seen in section 18 and is confirmed by clinical reports in section 20.

Test	Mean	Quantity	Sample Standard Deviation	Capability, Cp
Laser diode power output: Paddle Laser diode output is set to 40mw +/-15%.	39.35	1140	0.81	2.47
Probe Laser diode output is set to 40mw +/-15%.	40.02	40	0.72	3.49
EU portable appliance testing	100% pass	20		
 Visual Inspection: external components, cables, outer casings and the General condition of all associated parts. 	100% pass	20		100% inspection
Assembly tests	100% pass	20		100% inspection
Functionality tests	100% pass	20		100% inspection

 Table 2 - Product Release Testing Summary

Biocompatibility:

Biocompatibility assessment on the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system have been conducted following the ISO 10993-1 (FDA Recognized Consensus Standard), Biological Evaluation of Medical Devices, Part 1: Evaluation and Testing. Please see section 15.

Sterilization:

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are supplied non-sterile.

Cleaning instructions are contained in the user manual included with each system.

Safety and Effectiveness:

A summary of safety and effectiveness information ("510(k) Summary" per 21 CFR 807.92) upon which an equivalence determination has been made is included as Section 5.

Indications for Use Statement:

A single sheet statement of "indications for use" as required by FDA for submissions after January 01, 1996, is included in Section 4.

Truthful and Accurate Statement:

The premarket notification "truthful and accurate" statement (per 21 CFR 807.87(k)), signed by the Group Director Ian Cobley of Laser Lipo Ltd is included as Section 6.

E Copy Statement:

The E Copy provided within our submission is an exact duplicate of the paper copy.

Owner:

This Pre-Market Notification is owned by Laser Lipo Ltd.

Submitter:

This Pre-Market Notification is submitted by Ian Cobley Group Director For and on behalf of: Laser Lipo Ltd

Contact Persons:

The address and telephone number for **<u>all</u>** correspondence is:

Ian Cobley Laser Lipo Ltd Heath House Crockham Hill Edenbridge Kent TN8 6ST United Kingdom Tel: 011 44 844 980 1820 Mobile: 011 44 777 445 9611 Fax: 011 44 844 980 1820 e-mail: ian@strawberry-laser.com

If I can be of any further assistance, please do not hesitate to contact me.

Sincerely,

Ian Cobley Group Director Laser Lipo Ltd

4. Indications for Use Statement

Indications for Use

510(k) Number (if known):

Device Name: Laser Lipo Ltd Strawberry Low Level Laser system and Strawberry and Cream Low level Laser system

Indications for Use:

The Low level Laser model Strawberry and Strawberry and Cream can be used for the non-invasive temporary reduction in waist circumference by the disruption of adipocyte cells within the fat layer for the release of fat and lipids from these cells for non-invasive aesthetic use.

Prescription Use _ (Part 21 CFR 801 Subpart D) AND/OR

Over-The-Counter Use (21 CFR 801 Subpart C)

(PLEASE DO NOT WRITE BELOW THIS LINE-CONTINUE ON ANOTHER PAGE OF NEEDED)

Concurrence of CDRH, Office of Device Evaluation (ODE)

Page _1_ of _1_

5. 510(k) Summary

Laser Lipo Ltd Heath House, Crockham Hill Edenbridge United Kingdom, TN8 6ST Telephone: +44 844 980 1820 e-mail: ian@strawberry-laser.com



510[k] Summary

as required by section 807.92(c)

Owner's Name Laser Lipo Ltd	Addre	ess:		Heath House Crockham Hill Edenbridge
	Tel: Mobile Fax:	e:		Kent TN8 65T United Kingdom 011 44 844 980 1820 011 44 777 445 9611 011 44 1732 866 231
Contact Person:	Tel: Mobil Fax:	e:		Ian Cobley at Laser Lipo Ltd 011 44 844 980 1820 011 44 777 445 9611 011 44 844 980 1820
	E-ma	il:	I	ian@strawberry-laser.com
Date this summary was	s prepa	ared:		January 31, 2013
Classification name:		Low L 21 CF 79 OI	₋evel la R 878 _I	aser system for aesthetic use, .5400
Common/Usual Nam	e:	Low L	evel la	aser system for aesthetic use
Proprietary Name:		Laser	Lipo L	td will manufacture two devices:

- The Strawberry low level laser system model ILO, and,
- The Strawberry & Cream low level laser system model SC

Establishment Registration Number:

The Strawberry low level laser system and Strawberry & Cream low level laser system will be manufactured by: Laser Lipo Ltd Heath House Crockham Hill Edenbridge Kent TN8 6ST United Kingdom Telephone: Fax: 011 44 844 980 1820

Establishment Registration Number: To be applied for following clearance of this submission.

Substantial Equivalence: The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is substantially equivalent in design, use and materials to the: Chromogenex Technologies Ltd I-Lipo System – K111501

- They are made of the similar materials
- They have similar indications for use that are achieved through the same technology
- They are assembled from similar components
- They have similar user input and output interfaces and
- Have similar features.

In particular both the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd i-Lipo System:

- use laser diodes
 - o of approximately 660 nm wavelength, and
 - less than 50 mW output
- incorporated into
 - multiple multi-diodes "paddles" attached to the subject with special straps
 - Strawberry 4 to 10 paddles with 6 diodes each
 - i-Lipo 4 paddles with 9 diodes each
 - \circ $\,$ and both have two probes with one diode each
- have a similar control unit
 - LCD subject display
 - membrane key pad inputs

- incorporated into display as a touch screen for strawberry and cream

 microprocessor controlled through software
 - multi-range power supply (EU and US voltages and frequencies)
- have a similar range of connecting cables
- have the same laser protection goggles
 - o different name printed on outside of frame

Description of Product:

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system consists of a control unit, various connection leads up to 10 paddles, 2 probes, various "Velcro" attachment straps and other accessories. (A full list of system contents is included at the end of this section.)

A paddle is a device containing six cold red laser emitting diodes which is designed to be placed on the skin. The system can operate using 4, 6, 8 or 10 paddles that are connected to the control unit.

A probe is a device containing one cold red laser emitting diode which is designed to be placed on the skin to treat specific smaller areas of fat, where a usual flat paddle won't ergonomically fit.

The control unit is an electrically powered unit (100-240v, 50-60Hz auto-ranging), enabled by a main switch and key switch. Once enabled it is controlled using a button (Strawberry) or touchscreen (Strawberry and Cream) interface. The output of the diode (six per paddle) is limited to 40mW +/-15% by a power limiting PCB from the central processing unit.

When the laser paddles are placed on the skin, the cold red laser beams penetrate the skin just deep enough to reach the layers of fat. When the light hits the fat cells, a rapid chain of events takes place. Firstly, pores form on the cells causing them to spill out. The water, Glycerol and fatty acids move into the interstitial space beneath the fatty layer in the skin. Then further water, fatty acids and Glycerol spill out. The adipocyte cells are therefore reduced in size.

Intended use:

The Low level Laser model Strawberry and Strawberry and Cream can be used for the non-invasive temporary reduction in waist circumference by the disruption of adipocyte cells within the fat layer for the release of fat and lipids from these cells for non-invasive aesthetic use.

Performance Data

Laser wavelength and output has been demonstrated to be capable and substantially equivalent to the predicate device (Chromogenex i-Lipo, K 111051).

Test	Mean	Quantity	Sample Standard Deviation	Capability, Cp
Laser diode power output: Paddle Laser diode output is set to 40mw +/-15%.	39.35	1140	0.81	2.47
Probe Laser diode output is set to 40mw +/-15%.	39.02	40	0.72	3.49
EU portable appliance testing	100% pass	20		
 Visual Inspection: external components, cables, outer casings and the General condition of all associated parts. 	100% pass	20		100% inspection
Assembly tests	100% pass	20		100% inspection
Functionality tests	100% pass	20		100% inspection

Laser Lipo Ltd believes these demonstrate substantial equivalence, but have included clinical papers confirming similar performance from devices legally marketed within the European Union.

January 31, 2013

6. Truthful and Accuracy Statement

6 Truthful and Accuracy Statement

Center for Devices and Radiological Health

January 31, 2013

Laser Lipo Ltd Heath House, Crockham Hill Edenbridge United Kingdom, TN8 6ST Telephone: +44 844 980 1820 e-mail: ian@strawberry-laser.com



January 31, 2013

I certify that, in my capacity as Group Director of Laser Lipo Ltd, I believe to the best of my knowledge, that all data and information submitted in the premarket notification are truthful and accurate and that no material fact has been omitted.

-. ~ F

(Signature)

Ian Cobley

(Typed Name)

January 31, 2013

(Date)

(Premarket Notification [510(k)] Number)

*For a new submission, leave the 510(k) number blank. Must be signed by a responsible person of the firm required to submit the premarket notification [e.g., not a consultant for the 510(k) submitter].

6 Truthful and Accuracy Statement

7. Class III Summary and Certification

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system products are class II medical devices hence no class III summary or certification is required.

8. Financial Certification or Disclosure Statement

No clinical trial data is presented in this submission and hence no Financial Certification or Disclosure Statement is required.

9. Declarations of Conformity and Summary Reports

SECTION 9 TABLE OF CONTENTS

9. Dec	larations of Conformity and Summary Reports	9.1
9.1.	Introduction	9.2
9.2.	FDA Guidance Documents	9.3
9.3.	FDA Recognized Consensus Standards	9.4
9.4.	FDA Guidance Cross-Reference Tables	9.5
9.4	.1. FDA Guidance to IEC 62304 Software Report	9.5
9.4	.2. Low Level Laser Guidance to this Submission	
9.5.	FDA Standard Forms, FDA 3654	9.9

9.1. Introduction

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system have been "CE Marked" for European distribution under the monitoring of the European Notified Body SGS; conformity to international and European standards is an important aspect of this process.

Where these international consensus standards have been recognized by the FDA, we have included the relevant test reports but we have also considered both the extent of FDA recognition and FDA's general and product specific guidance.

While we appreciate FDA does not recognize ISO 13485 (Quality System Standard for Medical Devices) and that this standard differs from FDA's current Good Manufacturing Practices (Quality System Regulation 21 CFR 820), we ask you to note that the Laser Lipo Quality management System has been accredited under this standard by their Notified Body SGS as part of the CE Marking process.

9.2. FDA Guidance Documents

The submission format and section numbering system has been based on FDA Guidance for the "Format of Traditional and Abbreviated 510[k]s Guidance", dated August 12, 2005.

The product specific guidance "Class II Special Controls Guidance Document: Low Level Laser System for Aesthetic Use" has been used extensively in the preparation of this submission

In the development of the software for the Strawberry and Strawberry and Cream Low Level Laser systems, Laser Lipo has followed the IEC 62304 LifeCycle model and we have included a cross-reference table to the FDA Document "Guidance for the Content of Premarket Submissions for Software Contained in Medical Devices".

The FDA Guidance "Laser Products - Conformance with IEC 60825-1 and IEC 60601-2-22" has also been considered in the compilation of this submission and IEC 60601-2-22 has been considered together with electrical safety and electromagnetic compatibility in Section 17.

9.3. FDA Recognized Consensus Standards

IEC 62304:2006 Medical Device Software, Software Life-cycle Processes FDA Recognition List Number: 13-8 Effective Date: 2006

IEC 60601-1:2006 Medical Electrical Equipment, General Requirements for Safety

FDA Recognition List Number: 5-4

The Laser Lipo products were originally tested to the second edition standard but have been retested to the third edition standard which we understand FDA now accepts as demonstrating electrical safety. The full test report is included in Section 16.

IEC 60601-1-2:2007 Medical Electrical Equipment, General Requirements for safety and essential performance, Collateral standard, Electromagnetic compatibility and tests FDA Recognition List Number: 5-28 Effective Date: 2001

IEC 60601-2-22:1996 Medical electrical equipment, Particular Requirements for safety. Specification for Diagnosis and Therapeutic Laser Equipment. FDA Recognition List Number: 12-197 Effective Date: 1995

IEC 60825-1:2007 Safety of laser products, Equipment classification and Requirements FDA Recognition List Number: 12-168 Effective Date: 2007

ISO 14971:2009 Application of Risk Management to Medical Devices FDA Recognition List Number: 5-70 Effective Date: 04/10/2007

ISO 10993-1:2009 Biological evaluation of Medical Devices, Evaluation and Testing FDA Recognition List Number: 2-152 Effective Date: 01/2006

9.4. FDA Guidance Cross-Reference Tables

9.4.1. FDA Guidance to IEC 62304 Software Report

FDA Software in Submissions Guidance Item	Specifics: FDA Requirements for Moderate level of concern in Guidance for the Content of Premarket Submissions for Software Contained in Medical Devices	Location, in the Software Development Report ,Section 16.3, unless otherwise stated.	Location in Additional Software Report, section 16.4
Level of concern	Guidance for the Content of Premarket Submissions for Software Contained in Medical Devices	Section 16.1.2 of the 510k	N/A
Software Description	A summary overview of the features and software operating environment.	9.0 Development Plan >Background >Development Requirements (development process, programming environment, Strawberry, Strawberry & Cream)	N/A
Device Hazard Analysis	Tabular description of identified hardware and software hazards, including severity assessment and mitigations.	6.0 Software Classification 3.0 Introduction >Risk Management 4.0 Purpose	16.4.1
Software Requirements Specification (SRS)	The complete SRS document.	7.0 Requirements 9.0 Development Plan >Development requirements	16.4.2

FDA Software in Submissions Guidance Item	Specifics: FDA Requirements for Moderate level of concern in Guidance for the Content of Premarket Submissions for Software Contained in Medical Devices	Location, in the Software Development Report ,Section 16.3, unless otherwise stated.	Location in Additional Software Report, section 16.4
Architecture Design	Detailed depiction of	10.0 Software	16.4.3
Chart	functional units and	Performance and	
	software modules. May	Functional	
	include state diagrams	Requirements	
	as well as flow charts.	13.0 Annex A: Software	
		Listing for Strawberry	
		14.0 Annex B: Software	
		Listing for Strawberry &	
		Cream	
Software Design	Software design	10.0 Software	16.4.4
Specification (SDS)	specification	Performance and	
	document.	Functional	
		Requirements	
Traceability Analysis	Traceability among requirements, specifications, identified hazards and mitigations, and Verification and Validation testing.	9.0 Development plan >Verification Planning >validation 10.0 Software Performance and Functional Requirements >Software Verification	N/A
Software Development	Summary of software	4.0 Purpose	N/A
Environment	life cycle	7.0 Requirements	
Description	development plan,	9.0 Development Plan	
	including a	12.0 Software	
	summary of the	Modification and	
	configuration	Maintenance Plan	
	management and		
	maintenance		
	activities.		
Verification and	Description of V&V	5.0 Scope	16.4.5
Validation	activities at	9.0 Development Plan	
Documentation	the unit, integration,	>Verification Planning	
	and	>Validation	
	system level. System	10.0 Software	
	level test	Performance and	
	protocol, including	Functional	
	pass/fail	Requirements	

FDA Software in Submissions Guidance Item	Specifics: FDA Requirements for Moderate level of concern in Guidance for the Content of Premarket Submissions for Software Contained in Medical Devices	Location, in the Software Development Report ,Section 16.3, unless otherwise stated.	Location in Additional Software Report, section 16.4
	criteria, and tests results.	>Software Verification	
Revision Level History	Revision history log, including release version number and date.	2.0 Document Revision History 3.0 Introduction >Revision 11.0 Software Release 12.0 Software Modification and Maintenance Plan	16.4.6
Unresolved Anomalies (Bugs or Defects)	List of remaining software anomalies, annotated with an explanation of the impact on safety or effectiveness, including operator usage and human factors.	12.0 Software Modification and Maintenance Plan	N/A

9.4.2. Low Level Laser Guidance to this Submission

Low Level Laser Guidance Section	Item	Location in this 510[k]
 Background Introduction and Scope 		2. Cover Sheet 3. Cover Letter etc
4. Device Description	Device Components	11. Device Description
····	Photographs or Drawing	11. Device Description
	Comparison to Predicate Device	12. SE Discussion
5. Risks to Health	Ocular Injury	11. Description (Goggles) 12. SE Discussion
	Electric Shock	16. IEC 60601 Testing
	Unintended Cell Damage	12. SE Discussion
	Use Error	13. Proposed labeling
6. Bench Testing	To design/performance specification	18. Performance Testing
	Laser Power Output	18. Performance Testing
	Targeting	18. Performance Testing
	Failure Simulation	18. Performance Testing
7. Software Validation	Software in Subs. Guidance	See table above section 9.4.1
	IEC 60601-1-4	17. IEC 62304 Report
8. Clinical Testing	Alternatives	12. SE Discussion 16. IEC Safety Testing 18. Bench Testing
9. Biocompatibility	ISO 10993-1	15. Biocompatibility
10. EMC	IEC 60601-1-2	16. IEC 60601-1-2 Testing
11. Elec and Mech Safety	IEC 60601-1	16. IEC 60601-1
12. Labeling:		13. Proposed Labeling
_	Device User Manual	13. Draft User Manual
	Directions for Use	13. Draft User Manual
	Indications for Use	4. IFU Statement 12. SE Discussion 13. Draft User Manual
	Contraindications	13. Draft User Manual
	Storage Conditions	13. Draft User Manual
	Warnings	13. Draft User Manual
	Precautions	13. Draft User Manual

9.5. FDA Standard Forms, FDA 3654

FDA Standards Forms (FDA #3654) for each of the recognized consensus standards follow:

- ISO 10993
- ISO 14971
- IEC 60601-1-2
- IEC 60601-1-1
- IEC 60601-2-22
- IEC 60825
- IEC 62304

Form Approved: OMB No: 0910-0120; Expiration Date: 8/31/10

hen submitting a 510(indard referenced in th	(k) tha he 51	at refer- 0(k).
		000000
Ye	98	No
	3	
	152	
tandard identified	כ	
luded in the	ב	Ø
standard as it	3	
	ו	
	כ	Ø
[SIS) ^{\$} ?]	
E	כ	Ø
ystem for Aesthetic U	2 2 50	
n conformance assessment ont includes intermation on rent of the device. on sheet (SIS) is additional "DA recognizes the standard gowiscripts/odm/cfdocs/cfSb H Guidance Documents can html	to this all stan informs d. Four andard be fou	idards ston nd at s ² nd at
	SIS) ⁵ ?	SIS) ⁵ ?

	EXTENT OF STANDARD CC SUMMARY REPORT	NFORMANCE
STANDARD TITLE		1
150 10993-1:2009	Biological evaluation of Medical Devices, Evalua	ation and Lesting
	CONFORMANCE WITH STAND	ARD SECTIONS*
SECTION NUMBER	SECTION TIFLE	
TYPE OF DEVIATION	OR OPTION SELECTED *	
DESCRIPTION		
JUSTIFICATION		
SECTION NUMBER	SECTION TITLE	CONFORMANCE?
TYPE OF DEVIATION	OR OPTION SELECTED *	
DESCRIPTION		
JUSTIFICATION		
SECTION NUMBER	SECTION TITLE	CONFORMANCE?
TYPE OF DEVIATION	OR ORTION SEI SOTED *	Yes No NA
THE OF DEVIATION	OR OF HOW BELEGIED	
DESCRIPTION		
JUSTIFICATION		
 For completeness an explanation is n to be described an options selected w tion' on the report Types of deviations information sheet (list all sections of the standard and indicate whether weeded under " justification." Some standards included d adequately justified as appropriate for the subject in then following a standard is required under " type of or More than one page may be necessary. Is can include an exclusion of a section in the standard SIS), a deviation to adapt the standard to the device.	conformance is met. If a section is not applicable (N/A) e options, so similar to deviations, the option chosen nats device. Explanation of all deviations or description of deviation or option selected," * description* and * jostifi d, a deviation brought out by the FDA supplemental or any adaptation of a section.
	Paperwork Reduction Act	Statement
Public reporti time for revie completing as aspect of this	ng burden for this collection of information is estima wing instructions, searching existing data sources, ga ad reviewing the collection of information. Send corr collection of information, including suggestions for	ated to average 1 hour per response, including the athering and maintaining the data needed, and iments regarding this burden estimate or any other reducing this burden, to:
	Center for Devices and Radiolo 1350 Piccard Drive Rockville, MD 20850	gical Health
An agency	may not conduct or sponsor, and a person is not req unless it displays a currently valid (uired to respond to, a collection of information IMB control number.
ORM FDA 3654 (9/	07) Page 2	

Form Approved, OMB No. 0910-0120; Expiration Date: 8/31/10

Department of Health Food and Drug STANDARDS DATA F (To be filled in	and Human Services Administration REPORT FOR 510(k)s by applicant)		
This report and the Summary Report Table are to be comp ences a national or international standard. A separate repo	pleted by the applicant when submitting a 5 rt is required for each standard referenced i	10(k) ti n the 5	hat refer- 10(k).
TYPE OF 510(K) SUBMISSION	Abbreviated		
STANDARD TITLE 1 ISO 14971:2009 Application of Risk Management to Medica	al Devices		
Please answer the following questions		Yes	No
Is this standard recognized by FDA ² ?			
FDA Recognition number ³		5-70	
Was a third party laboratory responsible for testing conform in the 510(k)?	nity of the device to this standard identified		
Is a summary report ⁴ describing the extent of conformance 510(k)? If no, complete a summary report table.	e of the standard used included in the		Ø
Does the test data for this device demonstrate conformity to pertains to this device?	o the requirements of this standard as it	Ø	
Does this standard include acceptance criteria? If no, include the results of testing in the 510(k).			
Does this standard include more than one option or selection If yes, report options selected in the summary report table.	on of tests?		
Were there any deviations or adaptations made in the use of If yes, were deviations in accordance with the FDA supplem	of the standard? nental information sheet (SIS) ⁵ ?		N
Were deviations or adaptations made beyond what is speci If yes, report these deviations or adaptations in the summa	fied in the FDA SIS? ry report table.		
Were there any exclusions from the standard? If yes, report these exclusions in the summary report table.			Ø
Is there an FDA guidance ⁶ that is associated with this stand If yes, was the guidance document followed in preparation Title of guidance. Guidance for the Content of Premarket S	lard? of this 510k? ubmissions for Software Contained in Medic	2 2 al Dev	
 ¹ The formatting convention for the stle is: [SDD] [numeric identifier] [Bite of standard] [date of publication] ² Authority [21 U.S.C. 360d], www.ttla.gov/cdm/sidsprog.html ³ http://www.accessedata.fds.gov/scripts/cdm/cfdocs/cfStandards/ search.cfm ⁴ The summary report should include: any adaptations used to adapt to the device under review (for example, alternative test methods); choices made when options or a selection of methods are described, deviations from the standard; requirements not applicable to the device; and the name and address of the test laboratory or 	certification body involved in conformance assessm standard. The summary report includes information utilized during the development of the device. ⁶ The supplemental information sheet (SIS) is additio which is necessary before FDA recognizes the stan http://www.accessdata.fda.gov/scripts/cdm/cfdocs/ search.cfm ⁶ The online search for CDRH Guidance Documents www.fda.gov/cdm/guidance.html	ient to th on all st dard. Fo cfStanda can be %	s anderds nation nund at rds: xund at

	EXTENT OF STANDARD CONFOR SUMMARY REPORT TABLE	MANCE
STANDARD TITLE ISO 14971:2009 A	oplication of Risk Management to Medical Devices	
	CONFORMANCE WITH STANDARD SE	CTIONS*
SECTION NUMBER	SECTION TITLE	
TYPE OF DEVIATION	DR OPTION SELECTED *	L
DESCRIPTION		ţ,
JUSTIFICATION		
SECTION NUMBER	SECTION TITLE	CONFORMANCE?
TYPE OF DEVIATION	OR OPTION SELECTED *	
DESCRIPTION		
JUSTIFICATION		
SECTION NUMBER	SECTION TITLE	
TYPE OF DEVIATION	DR OPTION SELECTED *	
DESCRIPTION		
JUSTIFICATION		
 * For completeness an explanation is n to be described an options selected w tion[*] on the report * Types of deviations information sheet (ist all sections of the standard and indicate whether conform eeded under " justification." Some standards include option d adequately justified as appropriate for the subject device. I then following a standard is required under " type of deviation . More than one page may be necessary. can include an exclusion of a section in the standard, a dev SIS), a deviation to adapt the standard to the device, or any	nance is met. If a section is not applicable (N/A) is, so similar to deviations, the option chosen near Explanation of all deviations or description of in or option selected," * description* and * jostifi viation brought out by the FDA supplemental adaptation of a section.
	Paperwork Reduction Act Staten	oent
Public reporti time for revie completing ar aspect of this	ng burden for this collection of information is estimated to a wing instructions, searching existing data sources, gathering d reviewing the collection of information. Send comments r collection of information, including suggestions for reducing	average 1 hour per response, including the and maintaining the data needed, and regarding this burden estimate or any other g this burden, to:
	Center for Devices and Radiological He 1350 Piccard Drive Rockville, MD 20850	calth
An agency	may not conduct or sponsor, and a person is not required to unless it displays a currently valid OMB con	respond to, a collection of information ntrol number.

FORM FDA 3654 (9/07)

Page 2

Form Approved: OMB No. 0910-0120; Expiration Date: 8/31/10

Department of Health Food and Drug STANDARDS DATA F (To be filled in	and Human Services Administration REPORT FOR 510(k)s In by applicant)		
This report and the Summary Report Table are to be comp ences a national or international standard. A separate repo	pleted by the applicant when submitting a 5 ort is required for each standard referenced	10(k) ti in the 5	hat refer- 10(k).
TYPE OF 510(K) SUBMISSION	Abbreviated		
STANDARD TITLE IEC 60601-1-2:2007 Medical Electrical Equipment, General	Requirements for safety and essential perfo	ormanc	e
Please answer the following questions		Yes	No
Is this standard recognized by FDA ² ?			
FDA Recognition number ³		∮ 5-28	
Was a third party laboratory responsible for testing conformity of the device to this standard identified in the 510(k)?			
Is a summary report ⁴ describing the extent of conformance of the standard used included in the 510(k)?			
Does the test data for this device demonstrate conformity to the requirements of this standard as it pertains to this device?			
Does this standard include acceptance criteria? If no, include the results of testing in the 510(k).			
Does this standard include more than one option or selection of tests?			
Were there any deviations or adaptations made in the use of the standard? If yes, were deviations in accordance with the FDA supplemental information sheet (SIS) ⁵ ?			
Were deviations or adaptations made beyond what is specified in the FDA SIS? If yes, report these deviations or adaptations in the summary report table.			
Were there any exclusions from the standard? If yes, report these exclusions in the summary report table.			N
Is there an FDA guidance ⁶ that is associated with this standard? If yes, was the guidance document followed in preparation of this 510k? Title of guidance: Class II Special Controls Guidance Document: Low Level Laser System for Aestheti		2 2 Use	
 ¹ The formatting convention for the title is: (SDO) [numeric identifier] [title of standard] [date of publication] ² Authority [21 U.S.C. 360d], www.tda.gov/cdrh/stdsprog.html ⁹ http://www.accessdata.fda.gov/scripts/cdrh/cfdocs/cfStandards/ search.cfm ⁴ The summary report should include: any adaptations used to adapt to the device under review (for example, alternative test methods); choices made when options or a selection of methods are described; deviations from the standard; requirements not applicatile to the device; and the name and address of the test laboratory or 	certification body Involved in conformance assess standard. The summary report includes information utilized during the development of the device. The supplemental information sheet (SIS) is addition which is necessary before FDA recognizes the star http://www.accessdata.fda.gov/scripts/cdrh/cfdiocs/ search.cfm The online search for CDRH Guidance Documents www.fda.gov/cdrh/guidance.html	ient to thi on all sta insi inform idard. Fo ofStandar can be to	is andards nation rund at rds/
JKM FDA 3654 (9/07) Pag	ge 1	EC Origina I	2912445-1898

9 Declarations of Conformity and Summary Reports

	EXTENT OF STANDARD SUMMARY REPO	CONFORMANCE
STANDARD TITLE IEC 60601-1-2:200	7 Medical Electrical Equipment, General Req	irements for safety and essential performance
	CONFORMANCE WITH STAN	IDARD SECTIONS*
SECTION NUMBER	SECTION TITLE	
TYPE OF DEVIATION	OR OPTION SELECTED *	
DESCRIPTION		
JUSTIFICATION		
SECTION NUMBER	SECTION TITLE	
TYPE OF DEVIATION	OR OPTION SELECTED *	had too had too had too
DESCRIPTION		
JUSTIFICATION		
SECTION NUMBER	SECTION TITLE	CONFORMANCE?
TYPE OF DEVIATION	OR OPTION SELECTED *	
DESCRIPTION		
JUSTIFICATION		
 For completeness an explanation is n to be described an options selected w tion⁻ on the report Types of deviations information sheet (ist all sections of the standard and indicate wheth eeded under " justification." Some standards incl d adequately justified as appropriate for the subje hen following a standard is required under " type - More than one page may be necessary. Is can include an exclusion of a section in the stan SIS), a deviation to adapt the standard to the devi	er conformance is met. If a section is not applicable (N/A) ude options, so similar to deviations, the option chosen near ct device. Explanation of all deviations or description of of deviation or option selected," " description" and " jostifi sard, a deviation brought out by the FDA supplemental ce, or any adaptation of a section.
2. 	Paperwork Reduction	Act Statement
Public reporti time for revie completing ar aspect of this	ng burden for this collection of information is esti wing instructions, searching existing data sources and reviewing the collection of information. Send c collection of information, including suggestions f	mated to average 1 hour per response, including the gathering and maintaining the data needed, and omments regarding this burden estimate or any other or reducing this burden, to:
	Center for Devices and Radio 1350 Piccard Drive Rockville, MD 20850	ological Health
An agency	may not conduct or sponsor, and a person is not unless it displays a currently vali	required to respond to, a collection of information d OMB control number.
FORM FDA 3654 (9/	07) Page 2	

Form Approved: OMB No. 0910-0120; Expiration Data: 8/31/10

Department of Health Food and Drug STANDARDS DATA F (To be filled in	and Human Services Administration REPORT FOR 510(k)s In by applicant)		
This report and the Summary Report Table are to be comp ences a national or international standard. A separate report	pleted by the applicant when submitting a 5 rt is required for each standard referenced i	10(k) th n the 5	nat refer- 10(k).
TYPE OF 516(K) SUBMISSION	Abbreviated		
STANDARD TITLE ' IEC 60601-1:2006 Medical Electrical Equipment, General R	equirements for Safety		
Please answer the following questions		Yes	No
Is this standard recognized by FDA ² ?			
FDA Recognition number ³		5-4	
Was a third party laboratory responsible for testing conform in the 510(k)?	nity of the device to this standard identified		
Is a summary report ⁺ describing the extent of conformance of the standard used included in the 510(k)?			
Does the test data for this device demonstrate conformity to the requirements of this standard as it pertains to this device?			
Does this standard include acceptance criteria? If no, include the results of testing in the 510(k).			
Does this standard include more than one option or selection of tests?			
Were there any deviations or adaptations made in the use of the standard? If yes, were deviations in accordance with the FDA supplemental information sheet (SIS) ⁵ ?			
Were deviations or adaptations made beyond what is specified in the FDA SIS? If yes, report these deviations or adaptations in the summary report table.			
Were there any exclusions from the standard?			
Is there an FDA guidance ⁶ that is associated with this standard? If yes, was the guidance document followed in preparation of this 510k?		2 2 Use	
 ¹ The formatting convention for the title is: (SDO) [numeric identifier] [title of standard] [date of publication] ² Authonity [21 U.S.C. 360d], www.fda.gov/cdrtv/stdsprog.html ³ http://www.accessedata.fda.gov/scripts/cdrtv/ctrlocs/cfStandards/ search.cfm ⁴ The summary report should include: any adaptations used to adapt to the device under review (for example, alternative test methods); choices made when options or a selection of methods are described; deviations from the standard, requirements not applicable to the device; and the name and address of the fest laboratory or 	certification body involved in conformance assessm standard. The summary report includes information utilized during the development of the device ¹⁴ The supplemental information sheet (SIS) is additio which is necessary before FDA recognizes the stan http://www.accessdata.ida.gow/scripts/cdr/r/cfdocs/ search.cfm ¹⁶ The online search for CDRH Guidance Documents www.fda.gov/cdr/riguidance.html	ent to th on all sti nal inforr dard. Fo clStandar can be fo	s inderds nation und at dis/
ORM FDA 3654 (9/07) Pag	ge 1 .	SC Gutin (91546-109

	EXTENT OF STANDAR SUMMARY REF	D CONFORMANCE
STANDARD TITLE IEC 60601-1:2006	Medical Electrical Equipment, General Rec	puirements for Safety
	CONFORMANCE WITH S	TANDARD SECTIONS*
SECTION NUMBER	SECTION TITLE	
TYPE OF DEVIATION	OR OPTION SELECTED *	
DESCRIPTION		
JUSTIFICATION		
SECTION NUMBER	SECTION TITLE	CONFORMANCE?
TYPE OF DEVIATION	OR OPTION SELECTED*	
DESCRIPTION		
JUSTIFICATION		
SECTION NUMBER	SECTION TITLE	CONFORMANCE?
TYPE OF DEVIATION	OR OPTION SELECTED*	
DESCRIPTION		
JUSTIFICATION		
* For completeness an explanation is n to be described an options selected w tion* on the report * Types of deviations information sheet (ist all sections of the standard and indicate wh eeded under " justification." Some standards d adequately justified as appropriate for the su hen following a standard is required under " ty . More than one page may be necessary. s can include an exclusion of a section in the si SIS), a deviation to adapt the standard to the o	ether conformance is met. If a section is not applicable (N/A) include options, so similar to deviations, the option chosen nets bject device. Explanation of all deviations or description of pe of deviation or option selected," " description" and " jostifi tendard, a deviation brought out by the FDA supplemental device, or any adaptation of a section.
	Paperwork Reducti	on Act Statement
Public reporti time for revie completing ar aspect of this	ng burden for this collection of information is wing instructions, searching existing data sour id reviewing the collection of information. Ser collection of information, including suggestion	estimated to average 1 hour per response, including the ces, gathering and maintaining the data needed, and id comments regarding this burden estimate or any other ns for reducing this burden, to:
	Center for Devices and R 1350 Piccard Drive Rockville, MD 20850	adiological Health
An agency	may not conduct or sponsor, and a person is unless it displays a currently	tot required to respond to, a collection of information valid OMB control number.
ORM FDA 3654 (9/	07) Page	2
Form Approved: OMB No. 0910-0120; Expiration Date: 8/31/10

510(k) th in the 5	hat refer- 10(k),
or Diagn	iosis and
Yes	No
<u>⊭</u> 12-197	75
P	
Ø	
R	
NN	
nent to thi i ori all str onal inform ndard. Fo icfStandar can be fo	s andards nation und at rds/ xund at
	ment to the nor all still to the formed and. For out of the formed and and are the formed and a scan be for the formed and a scan be scan be scan be formed

	EXTENT OF STAND/ SUMMARY R	ARD CONFORMANCE EPORT TABLE	
STANDARD TITLE IEC 60601-2-22:19	96 Medical electrical equipment, Particu	lar Requirements for safety. Speci	fication for Diagnosis and
	CONFORMANCE WITH	STANDARD SECTIONS*	7
SECTION NUMBER	SECTION TITLE		CONFORMANCE?
TYPE OF DEVIATION	OR OPTION SELECTED *		
DESCRIPTION			
JUSTIFICATION			2
SECTION NUMBER	SECTION TITLE		
TYPE OF DEVIATION	OR OPTION SELECTED *		
DESCRIPTION			
JUSTIFICATION			
SECTION NUMBER	SECTION TITLE		CONFORMANCE?
TYPE OF DEVIATION	OR OPTION SELECTED*		
DESCRIPTION			
JUSTIFICATION			12
* For completeness an explanation is n to be described an options selected w tion* on the report * Types of deviations	ist all sections of the standard and indicate eeded under " justification." Some standard d adequately justified as appropriate for the hen following a standard is required under " . More than one page may be necessary. s can include an exclusion of a section in the	whether conformance is met. If a sect is include options, so similar to deviat subject device. Explanation of all dev type of deviation or option selected," a standard, a deviation brought out by	tion is not applicable (N/A) lons, the option chosen nets lations or description of " description" and " justifi the FDA supplemental
information sheet (SIS), a deviation to adapt the standard to th	e device, or any adaptation of a section	an.
	Paperwork Redu	ction Act Statement	
Public reporti time for revie completing ar aspect of this	ng burden for this collection of information wing instructions, searching existing data so to reviewing the collection of information. S collection of information, including suggest	is estimated to average 1 hour per res surces, gathering and maintaining the send comments regarding this burden tions for reducing this burden, to:	ponse, including the data needed, and estimate or any other
	Center for Devices and 1350 Piccard Drive Rockville, MD 20850	Radiological Health	
An agency	may not conduct or sponsor, and a person t unless it displays a current	is not required to respond to, a collect ly valid OMB control number.	tion of information

Page 2

Form Approved: OMB No: 0910-0120;	Expiration	Date: 8/31/10
Department of Health and Human Services Food and Drug Administration STANDARDS DATA REPORT FOR 510(k)s (To be filled in by applicant)		
This report and the Summary Report Table are to be completed by the applicant when submitting a ences a national or international standard. A separate report is required for each standard reference	i 510(k) ti d in the 5	nat refer- 10(k).
TYPE OF 510(K) SUBMISSION		
STANDARD TITLE ' IEC 60825-1:2007 Safety of laser products, Equipment classification and Requirements		
Please answer the following questions	Yes	No
Is this standard recognized by FDA ² ?		
FDA Recognition number ³	#12-16	3
Was a third party laboratory responsible for testing conformity of the device to this standard identified in the 510(k)?		Ø
Is a summary report ⁴ describing the extent of conformance of the standard used included in the 510(k)? If no, complete a summary report table.		Ø
Does the test data for this device demonstrate conformity to the requirements of this standard as it pertains to this device?		
Does this standard include acceptance criteria? If no, include the results of testing in the 510(k).		
Does this standard include more than one option or selection of tests? If yes, report options selected in the summary report table.	. 🗆	Ø
Were there any deviations or adaptations made in the use of the standard? If yes, were deviations in accordance with the FDA supplemental information sheet (SIS) ⁵ ?		N
Were deviations or adaptations made beyond what is specified in the FDA SIS? If yes, report these deviations or adaptations in the summary report table.	. 🗆	N
Were there any exclusions from the standard? If yes, report these exclusions in the summary report table.	. 🗆	Ø
Is there an FDA guidance ⁶ that is associated with this standard? If yes, was the guidance document followed in preparation of this 510k? Title of guidance: Laser Products – Conformance with IEC 60825-1 and IEC 60601-2-22;	NN	
 ¹ The formatting convention for the title is: [SDD] [numeric identifier] [Itile of standard] [dete of publication] ² Authority [21 U.S.C. 360d], www.title.gov/cdrivisidsprog.html ³ http://www.accessedata.fds.gov/scripts/cdrivic/docs/cfStandards/ search.cfm ⁴ The summary report should include: any adaptations used to adapt to the device under review (for example, alternative test methods); choices made when options or a selection of methods are described, deviations from the standard, requirements not applicable to the device; and the name and address of the test laboratory or ⁶ The summary report for CDRH Guidance html ⁶ The online search for CDRH Guidance html 	isment to th ion on all sti itional inforr tandard. Fo cs/cfStandar its can be &	ls. anderds nation und at rdsi und at

Page 1

PSC Displaysis (SH) 445-1998 - 147

	EXTENT OF STANDARD SUMMARY REPO	CONFORMANCE RT TABLE
STANDARD TITLE IEC 60825-1:2007	Safety of laser products, Equipment classifica	tion and Requirements
	CONFORMANCE WITH STAM	IDARD SECTIONS*
SECTION NUMBER	SECTION TITLE	CONFORMANCE?
TYPE OF DEVIATION	DR OPTION SELECTED *	
DESCRIPTION		,
JUSTIFICATION		
SECTION NUMBER	SECTION TITLE	CONFORMANCE?
TYPE OF DEVIATION	OR OPTION SELECTED *	
DESCRIPTION		
JUSTIFICATION		
SECTION NUMBER	SECTION TITLE	CONFORMANCE?
TYPE OF DEVIATION	I OR OPTION SELECTED *	
DESCRIPTION		
JUSTIFICATION		
 For completeness I an explanation is n to be described an options selected wi tion" on the report Types of deviations information sheet (ist all sections of the standard and indicate wheth seded under " justification." Some standards incl d adequately justified as appropriate for the subje sen following a standard is required under " type i More than one page may be necessary. can include an exclusion of a section in the stand SIS), a deviation to adapt the standard to the devi	er conformance is met. If a section is not applicable (N/A) ude options, so similar to deviations, the option chosen nets ct device. Explanation of all deviations or description of af deviation or option selected," "description" and " joelifi fard, a deviation brought out by the FDA supplemental ce, or any adaptation of a section.
	Paperwork Reduction	Act Statement
Public reporti time for revie completing ar aspect of this	ng burden for this collection of information is esti wing instructions, searching existing data sources, d reviewing the collection of information. Send c collection of information, including suggestions f	mated to average 1 hour per response, including the gathering and maintaining the data needed, and omments regarding this burden estimate or any other or reducing this burden, to:
	Center for Devices and Radio 1350 Piccard Drive Rockville, MD 20850	logical Health
An agency	may not conduct or sponsor, and a person is not unless it displays a curvently vali	required to respond to, a collection of information d OMB control number.

Page 2

Form Approved: OMB No. 0910-0120; Expiration Date: 8/31/10

and Human Services Administration EPORT FOR 510(k)s by applicant)		
leted by the applicant when submitting a 5 t is required for each standard referenced i	10(k) tř n the 5	nat refer- 10(k).
Abbreviated		
le Processes		
	Yes	No
	13-8	
ity of the device to this standard identified		
of the standard used included in the		
the requirements of this standard as it		
in of tests?		Ø
f the standard? ental information sheet (SIS) ⁵ ?		
ied in the FDA SIS? y report table.		
ard? of this 510k?		
Ibmissions for Software Contained in Medic	al Dev	ces
certification body involved in conformance assessm standard. The summary report includes information utilized during the development of the device. ⁶ The supplemental information sheet (SIS) is addition which is necessary before FDA recognizes the stan http://www.accessidata.fda.gov/scripts/cdr/rcfdocs/ search.cfm ⁹ The online search for CDRH Guidance Documents www.fda.gov/cdr/rguidance.html	ent to th on all str dard. Fo d'Standar can be fo	s anderds nation und at dsi und at
	and Human Services Administration EPORT FOR 510(k)s by applicant) leted by the applicant when submitting a 5 t is required for each standard referenced i	and Human Services Administration EPORT FOR 510(k)s by applicant) leted by the applicant when submitting a 510(k) th t is required for each standard referenced in the 5 Abbreviated Abbreviated Abbreviated Abbrevi

	EXTENT OF STANDARD CO SUMMARY REPORT	ONFORMANCE TABLE
STANDARD TITLE IEC 62304:2006 M	edical Device Software, Software Life-cycle Pro	cesses
	CONFORMANCE WITH STAND	ARD SECTIONS*
SECTION NUMBER	SECTION TITLE	CONFORMANCE?
TYPE OF DEVIATION	OR OPTION SELECTED *	
DESCRIPTION		
JUSTIFICATION		
SECTION NUMBER	SECTION TITLE	CONFORMANCE?
TYPE OF DEVIATION	OR OPTION SELECTED *	
DESCRIPTION		
JUSTIFICATION		
SECTION NUMBER	SECTION TITLE	CONFORMANCE?
TYPE OF DEVIATION	OR OPTION SELECTED *	
DESCRIPTION		
JUSTIFICATION		
 For completeness I an explanation is n to be described an options selected w tion[*] on the report Types of deviations information sheet (ist all sections of the standard and indicate whether eeded under " justification." Some standards includ d adequately justified as appropriate for the subject her following a standard is required under " type of . More than one page may be necessary. can include an exclusion of a section in the standa SIS), a deviation to adapt the standard to the device	conformance is met. If a section is not applicable (N/A) e options, so similar to deviations, the option chosen nates device. Explanation of all deviations or description of deviation or option selected," " description" and " jostifi rd, a deviation brought out by the FDA supplemental or any adaptation of a section.
Public reporti time for revie completing ar aspect of this	Paperwork Reduction Ac ng burden for this collection of information is estim wing instructions, searching existing data sources, g ad reviewing the collection of information. Send cor collection of information, including suggestions for	t Statement ated to average 1 hour per response, including the athering and maintaining the data needed, and ments regarding this burden estimate or any other reducing this burden, to:
	Center for Devices and Radiolo 1350 Piccard Drive Rockville, MD 20850	gical Health
An agency	may not conduct or sponsor, and a person is not re unless it displays a currently valid	quived to respond to, a collection of information OMB control number:

Page 2

10. Executive Summary

SECTION 10 TABLE OF CONTENTS

10.	Exe	ecutive Summary	10-1
10.	1.	Description,	10-2
10.	2.	Substantial Equivalence Tables	10-2
10.	3.	Summary of Performance Testing	10-4

10.1. Description

Laser Lipo Ltd manufacture two devices:

- the 'Strawberry' with a liquid crystal display and membrane keypad and
- the 'Strawberry & Cream' with a touch screen display

These two low level laser systems are for the non-invasive temporary reduction in waist circumference by the disruption of adipocyte cells, within the fat layer, for the release of fat and lipids from these cells. This is done by anatomic redistribution, not weight loss. (21 CFR 878.5400 - 79 OLI)

The 'Strawberry' and the 'Strawberry & Cream' consist of a control unit, connection leads with up to 10 multi diode laser "paddles", 2 cluster probes, various "Velcro" attachment straps and other accessories.

The control units are electrically powered units (100-240v, 50-60Hz auto-ranging), enabled by a main switch and key switch. Once powered "ON", they are controlled using membrane buttons (Strawberry) and touch screen (Strawberry & Cream) interface. The output of the diodes is limited to 40mW +/-15% by a power limiting PCB from the central processing unit.

When the laser paddles are placed on the skin, the cold red laser beams permeate the skin just deep enough to reach the layers of fat. When the light hits the fat cells, a rapid chain of events takes place. Firstly, pores form on the cells causing them to spill out. That is, the water, Glycerol and fatty acids move into the interstitial spaces beneath the fatty layer in the skin. Then further water, fatty acids and Glycerol spill out. The adipocyte cells with reduced content are therefore reduced in size.

	Subject Device	Subject Device	Predicate Device
	Strawberry	Strawberry & Cream	I-Lipo
510(k)	THIS 510(k)	THIS 510(k)	K111501
Intended use	Indicated for non-invasive temporary reduction in waist circumference by the disruption of adipocyte cells within the fat layer for the release of fat and lipids from these cells for non- invasive aesthetic use.	same	indicated for Non- invasive treatment for the temporary reduction in Circumference of the waist.

10.2. Substantial Equivalence Tables

	Subject Device	Subject Device	Predicate Device
	Strawberry	Strawberry & Cream	I-Lipo
510(k)	THIS 510(k)	THIS 510(k)	K111501
Enclosure	Plastic	Same	Same
Display	LCD	Same	Same
Laser type	Class 3b	Same	Same
Membrane Keypad	Yes	Touch screen	Same as Strawberry
Electronic technology	Yes	Same	Same
Fuses	Provided	Same	Same
Control electronics	Yes	Same	Same
Emergency stop	Provided	Same	Same
Key switch	Provided	Same	Same
Speaker	Provided	Same	Same
Adjustment	Keypad	Touch screen	Same as
		interface	Strawberry
Power supply	100-240 v ac 50-60 Hz	Same	Similar 240 v ac 50 Hz 120 v ac 60 Hz
Transformer	Provided	Same	Same
Energy source	Laser Diode from 660 nm nominal	Same	Same
Energy output	up to 40 mW (50 mW probes)	Same	Similar up to 50 mW
Cooling requirements	Air cooled	Same	same
Paddle size	15.0 x 4.5 cm	Same	Similar 13 x 8.4 cm
Probe size	6 x 2.4 x 4.3cm	Same	Similar 2.7 x 3.6 x 2.6 cm
Environment Requirements	10°C to 30°C Non-condensing humidity's below 75%	Same	Same
Laser Output: wavelength output "paddles" laser diodes per "paddle" "cluster probes" laser diodes per "cluster probe"	660 ± 15 nm up to 62 x 50 mW 4 , 6, 8 or 10 6 2 1	Same Same Same Same Same Same	650 - 690 nm up to 38 x 50 mW 4 9 2 1
Laser class Laser Diode	IIIb GaAlAs	Same Same	IIIb GaAlAs
Safety Features: Prevention of unauthorized use	Key switch	Same	Interlock
"Emergency" Stop	"Stop" button	Same	Same

10.3. Summary of Performance Testing

Laser wavelength and output has been demonstrated to be capable and substantially equivalent to the predicate device (Chromogenex i-Lipo K111051).

Test	Mean	Quantity	Sample Standard Deviation	Capability, Cp
Laser diode power output: Paddle Laser diode output is set to 40mw +/-15%.	39.35	1140	0.81	2.47
Probe Laser diode output is set to 45mw +/-15%.	45.02	40	0.72	3.49
EU portable appliance testing	100% pass	20		
 Visual Inspection: external components, cables, outer casings and the general condition of all associated parts. 	100% pass	20		100% inspection
Assembly tests	100% pass	20		100% inspection
Functionality tests	100% pass	20		100% inspection

Laser Lipo Ltd has demonstrated substantial equivalence and have included a recent clinical paper conforming similar performance from devices legally marketed within the European Union.

11. Device Description

SECTION 11 TABLE OF CONTENTS

11. Devic	e Description	11.1
11.1. Su	ummary Device Description	11.2
11.1.1.	System Overview	11.2
11.1.2.	Control Unit Overview	11.4
11.1.3.	Packaging	11.5
11.1.4.	Shelf-Life Overview	11.5
11.1.5.	Laser	11.6
11.1.6.	System Control	11.6
11.1.7.	Laser Paddles	11.7
11.1.8.	Cluster Probes	11.11
11.1.9.	User safety Precautions	11.12
11.1.1(D. Power Lead	11.12
11.1.1	1. Paddle leads	11.13
11.1.12	2. Cluster Probe lead	11.13
11.1.13	3. Casing Screws	11.14
$11.1.1^{2}$	4. 1 Amp Fuse	11.14
11.1.15	5. Straps	11.15
11.1.16	6. Goggles	11.16
11.1.17	7. Keys	11.18
11.1.18	3. Presentation Box	11.19
11.2. Us	sability	11.21
11.4. H	lardware	11.28
11.3. M	echanical Drawings	11.53
11.4. Pł	notographs of Exploded Assemblies	11.56
11.5. El	ectrical Layout Drawings	11.60
11.6. Tr	aining and Documentation	11.63
11.6.1.	Consent Form and Medical Questionnaire	11.63
11.6.2.	Treatment Forms	11.64

11.1. Summary Device Description

11.1.1. System Overview

The Laser Lipo Ltd 'Strawberry' low level laser system and 'Strawberry & Cream' low level laser system both consists of a control unit, multiple connection leads with up to 10 paddles, 2 cluster probes, various "Velcro" attachment straps and other accessories. (A full list of system contents is included at the end of this section.)

Each paddle is a device containing six cold red laser emitting diodes which are designed to be placed on the skin. The system can operate using 4, 6, 8 or 10 paddles that are connected to the control unit. These paddles are identical for both the 'Strawberry' and Strawberry & Cream' systems.

The control unit is an electrically powered unit (100-240v, 50-60Hz autoranging), enabled by a main switch and key switch. Once enabled it is controlled using a button (Strawberry) or touchscreen (Strawberry and Cream) interface. The output of the diode (six per paddle) is limited to 40mW +/- 15% by a power limiting PCB from the central processing unit.

The control unit is contained within a plastic enclosure weighing approximately 14 lbs. The dimensions of the unit are approximately $8'' \times 10'' \times 12''$.

When the laser paddles are placed on the skin, the cold red laser beams permeates the skin just deep enough to reach the layers of fat. When the light reaches the fat cells, a rapid chain of events takes place. Firstly, pores form on the cells causing them to spill out water, Glycerol and fatty acids. These elements move into the interstitial space beneath the fatty layer in the skin. Due to these elements leaving the adipocyte cells, they are reduced in size.

The Laser Lipo Ltd 'Strawberry' system and 'Strawberry & Cream' low level laser systems contain the same control electronics and control software, however the 'Strawberry & Cream' has a menu controlled by a touch screen interface, whilst the 'Strawberry' model has a liquid crystal display and (up, down, left, right and center/enter) push buttons. The outputs and functionality of the 'Strawberry' and 'Strawberry & Cream' models are identical; however displays, display drivers, interface electronics and software differ. Full details are contained in Section 11.1.2.

The Laser Lipo Ltd ships both systems in dedicated packaging to prevent damage in transit. The systems are protected by cardboard boxes and polystyrene.

Each system is shipped with the following contents:

- o 1 low level laser system unit, either
 - Strawberry model, or
 - Strawberry & Cream model
- o 1 user manual
- o 1 power lead
- Paddle leads, for example for a 10 paddle system
 - 1 long paddle lead for connection to the device
 - 8 paddle standard leads
 - 1 long paddle lead
- o 2 Cluster Probes
- o 1 Pair of goggles
- o 2 Replacement fuses
- o 2 Keys

Variants of the system will be available with 4, 6, 8 and 10 paddles.

number of paddles	number of paddle leads
3 standard and 1 end	3 standard
5 standard and 1 end	5 standard
7 standard and 1 end	7 standard
9 standard and 1 end	8 standard and 1 long
	(to connect two groups of five paddles)

Table 1 - Variants

Photographs are used in the following sections as a visual aide.

11.1.2. Control Unit Overview



Figure 11.1.2 a) - Strawberry Low Level Laser Control Unit



Figure 11.1.2 b) - Strawberry & Cream Low Level Laser Control Unit

The 'Strawberry' and the 'Strawberry & Cream' low level laser control units contain:

- Power supply of 100-240V ac 50-60Hz
- Key switch
- Emergency stop
- LCD display or Touch screen display
- Keypad
- 32-bits EISC with ISP Flash, USB and ADPCM Engine
- Mother board for S&C
- Fuse
- Digital PCB
- Wires
- Speaker
- Transformer
- Ribbon cable
- Hexagon bolts, nuts and washers
- Upper and lower case enclosure

Engineering Drawings of the 'Strawberry' and the 'Strawberry & Cream' low level laser systems can be found in Section 11.5.

IEC 60601 electrical safety and electromagnetic compatibility test reports and certificates can be seen in Section 17.

The Software Development Report following the IEC 62304 Life-cycle methodology is contained in Section 16.

11.1.3. Packaging

The Laser Lipo Ltd 'Strawberry' system and 'Strawberry & Cream' low level laser systems will be packaged in a cardboard case with polystyrene inserts.

11.1.4. Shelf-Life Overview

A shelf-life has been determined for 5 years based on the laser diodes. Bench test reports are contained in Section 18.

11.1.5. Laser

The Laser Lipo Ltd 'Strawberry' and 'Strawberry & Cream' low level laser systems are non-invasive class 3b cold red laser and uses 2 different laser LED configuration which are:

- 1. 660nm 40mW (+/- 15%), x 6 ea LD for paddles
- 2. 660nm 40mW (+/- 15%), x 1 ea LD for cluster probes

Both models have a GAAIAS & GAA1LnP Diode Source and use a continuous wave diode driver.

11.1.6. System Control



Figure 11.1.6 a)

The 'Strawberry' low level laser system is controlled using a membrane keypad as the user interface to set the treatment time.



Figure 11.1.6 b)

The 'Strawberry & Cream' uses a touch screen user interface to select treatment time.

January 31, 2013



Figure 11.1.6 c)

The i-Lipo LCD display screen with push buttons to set treatment time.

11.1.7. Laser Paddles

Laser Paddle Comparison between the Strawberry and i-Lipo paddles

Although there is a difference in the number of paddles, they actually equate to the same in terms of treatment power / energy applied to the skin.

i-lipo treatment example;

An abdomen treatment with the i-Lipo pads involves two 10 minute treatments. The size of the pads only covers half of an average abdomen. The pads therefore need to be placed on two areas to cover the abdomen.



Fig. 11.1.7 a). i_lipo pad treating right side of the abdomen for 10 mins, then left side of the abdomen for 10 mins.

Image above show subjects right side being treated for 10 minutes. This is then followed with a second treatment of 10 minutes, with the pads placed on the subjects left side.

Therefore a treatment with the i-Lipo pads amount to two 10 minute treatments or 20 minutes of laser exposure time.

Strawberry treatment example;

The Strawberry paddles cover the entire abdomen area in one 10 minute treatment, without the need to move the paddles.



Fig. 11.1.7 b) Strawberry paddles placed for a treatment.

Physical size and number of diode comparison:



Fig. 11.1.7 c) Image of Strawberry paddles (x10) shown on the top, with i-Lipo pads (x4) shown beneath.



The i-Lipo laser treatment pads contain nine (9) laser diodes with a power output of 40mW and cover an active area of 84mm x 130mm.

When the four (4) laser pads are placed side by side the total skin area covered is 336mm x 130mm X 2 = 873.6 cm² for the entire treatment.

The pads are placed twice on the abdomen in order to treat the full abdomen area.

Fig. 11.1.7 d)

Joules are defined as Watts multiplied by the time in seconds, so Joules (J) = Watts x time (sec). Total energy given in Watts is calculated over an area as joules per centimeter squared (J/cm²).



The Strawberry paddles contain six (6) laser diodes with a power output of up to 40mW and cover an active area of 45mm x 150mm.

When 10 paddles are placed side by side the total skin area covered is 450 mm x 150 mm = 675 cm² for one entire abdomen treatment.

Fig. 11.1.7 e). Strawberry paddle

Energy emitted per abdomen treatment is calculated as follows:

Single i-Lipo pad	40mW x 9 diodes	= 360mW or 0.36W
4 pads	4 laser paddles (0.36W x 4)	= 1.44w

The total energy emitted during a single treatment $= 1.44W \times 1200sec = 1,728 J.$

Coverage of (4 pads x 2 placements) 8 pads area of 873.6 cm²

The energy that reaches the skin is therefore 1,728/873.6 cm² = 1.98 J/cm²

1x Strawberry pade	dle 40mW x 6 diodes	= 240mW or 0.24W
10 paddles	10 paddles (0.24W x 10)	= 2.40W.

The total energy emitted during a single treatment = $2.4W \times 600sec = 1,440 \text{ J}$. Coverage of (10 paddles each 450mm x 150mm is 675 cm²

The energy that reaches the skin is therefore $1,440 / 675 \text{SqCm} = 2.13 \text{ J/cm}^{2}$.

(N.B. Calculations are based upon individual diode output being 40mW +/- 15%).

Optional number of paddles supplied with the 'Strawberry' and 'Strawberry & Cream' low level laser systems.

The 'Strawberry' and the 'Strawberry & Cream' low level laser systems have been designed to operate with up to 10 laser paddles. The systems can be purchased with 4, 6, 8 and 10 paddles.

The reason for this option is a financial one. Naturally a system with 4 paddles is sold at a lower price than a 6, 8 or 10 paddle system.

If a practitioner were to purchase a 4 paddle system, as with the i-Lipo system, it would be necessary to move the paddles a number of times to carry out an abdomen treatment. This results in a number of 10 minute treatments being necessary to complete a full abdomen treatment.

The treatment and results will be the same between treatments carried out with 4, 6, 8 or 10 paddles but will have taken longer to achieve. As a result practitioners will be able to treat fewer subjects in a working day.

The system is therefore designed to be upgradeable by purchasing additional paddles. This means that as demand for treatments increases, the practitioner can purchase additional paddles to reduce the treatment times down.



Fig. 11.1.7 f). Strawberry 6 paddle system fitted to abdomen.



Fig. 11.1.7 g). Strawberry 10 paddles on abdomen.

The images below show (Fig.11.1.7 e) a 4 paddle system with probes and (Fig. 11.1.7.f) shows a set of 10 Strawberry paddles)



Fig. 11.1.7 e). Strawberry 4 paddle system.



Fig. 11.1.7 f). Strawberry 10 paddles.

11.1.8. Cluster Probes

The Cluster probes used with the i-Lipo system and the Strawberry Inch Loss systems are different in design (see Fig. 11.1.8 a) but substantially equivalent in power output and usage.

These probes are placed on the skin, in areas that are too small to fit laser paddles. The laser diodes are the same as those used in the larger laser paddles. They are used to assist the overall abdomen treatments to increase the treatment area.



Fig. 11.1.8 a) Strawberry Cluster probe on the left, i-Lipo cluster probe on the right.



Strawberry Cluster probe 40mW power output.

Fig. 11.1.8 b) Strawberry Cluster probe



i-Lipo cluster probe 40mW power output.

Fig. 11.1.8 c) i-Lipo Cluster probe

The cluster probes are used in conjunction with the laser paddles. Once the laser paddles have been attached to the Velcro belt and connected to the base unit, the probes are then connected to the base unit and positioned on to the skin.



There is one laser diode in each cluster probe and the cluster probes are supplied as a pair.

Fig. 11.1.8 d) i-Lipo Cluster probe

The 'Strawberry' and the 'Strawberry & Cream' low level laser systems are manufactured with 2 probes. The dimensions of the probes are approximately 2" x 1" x $1\frac{1}{2}$ ".

11.1.9. User safety Precautions

Prior to the device being switched on, the technician and subject need to wear protective goggles, see Section 11.1.16. Moreover a notice should be displayed on the door to the room, see section 13, showing the Laser Symbol and stating "Laser light, to avoid exposure to beam 3b laser product".

11.1.10. Power Lead



Fig. 11.1.10 a) US Power Lead

11.1.11. Paddle leads

The Laser Lipo Ltd 'Strawberry' and the 'Strawberry & Cream' low level laser systems are manufactured with 3 types of paddle lead:

- 1 long lead for the connection between the unit and the first paddle
- 3, 5, 7 or 8 standard for the interconnection between paddles
- 1 longer interconnection lead between the 2 middle paddles when two straps are used.

number and type of paddles	number and type of paddle leads
3 standard and 1 end	3 standard
5 standard and 1 end	5 standard
7 standard and 1 end	7 standard
9 standard and 1 end	8 standard and 1 longer size to connect two sets of paddles



Fig. 11.1.11 a) Paddle Leads

11.1.12. Cluster Probe lead

The two cluster probes are connected to a dual wire "Y" type lead which connects with a single plug to the socket on the control unit.



Fig. 11.1.12 a) Probe "Y" type lead

11.1.13. Casing Screws

The top and bottom control unit moldings are held together with six casing screws.



Fig. 11.1.13. a) Casing Screws (set of 6)

11.1.14. 1 Amp Fuse



Fig. 11.1.14 a) Fuse - Fitting in lower housing

The 'Strawberry' and the 'Strawberry & Cream' low level laser systems are shipped with two spare fuses. Additional fuses are available as accessories.

The fuse is a Quick Acting F LBC Fuse, 1A 5x20mm and ROHS Compliant.

January 31, 2013

11.1.15. Straps

The 'Strawberry' and the 'Strawberry & Cream' low level laser systems are manufactured with 3 different length straps:

- Large (135 cmx17 cm)
- Medium (75 cm x 17 cm)
- Small (52 cm x17 cm)



Fig. 11.1.15 a) Velcro straps used to hold the paddles around the body.

The 'Strawberry' Velcro st	traps are made of the following materials:
External surface material:	Nylon
Internal material:	Neoprene (80%) + Nylon (20%)
Fitment:	Velcro pads on Nylon surface

11.1.16. Goggles

The 'Strawberry' and the 'Strawberry & Cream' low level laser systems are supplied with one pair of goggles in a protective case (for the subject) with a cleaning cloth (see Figure 11-2 - Photographs of goggles supplied as part of the system



Fig. 11.1.16 a) Laser protection goggles supplied with cleaning cloth.

The goggles are made of nylon and the lenses are made of Polycarbonate, with a luminous transmittance of 16% and designed as laser protection goggles for the appropriate wavelength.

The goggles are the same model from the same manufacturer as the goggles supplied with the predicate device. (see certification Fig.11.1.16 b)

Each pair of goggles is supplied with information regarding the glasses. (See 11.1.16. b)

January 31, 2013





EC-type examination certificate

NOIR

C7045.4NOIR

Applicant / manufacturer

NOIR Laser Company 6155 Pontiac Trail SOUTH LYON, MI 48178 USA

Identification of the manufacturer

Product type

Model name

Standard(s) / technical rules

Test report

Material and optical properties

Marking

Laser protection filter

ML3 4base + 8base DIN EN 207 : 2009 Annex II of the PPE-Directive 89/686/EEC

10313-ECS-07

Polycarbonate Luminous transmittance D65 16%

190 – 315 D LB7 + IR LB4 >315 – 395 D LB5 + IRM LB6 630 – 660 DIR LB3 >660 – 670 DIR LB2 780 – 920 DIR LB2 800 – 915 DIR LB3 NOIR CE

Herewith, ECS certifies that the named model complies with the basic requirements for health and safety as they are provided by the European Directive for Personal Protective Equipment 89/666/EEC. This certificate is based both / either on the test results as they are summarized in the named test reports, and/or on the technical documentation as it is delivered by the manufacturer. The applicant / manufacturer agrees to the General Business Rules of the ECS GmbH and to additional agreements as they are named in the application for conformity assessment.

The eye-protection device is to be marked as assigned. Either / both the frame and / or the ocular, spectacle, goggle or shield must be signed, as appropriate. If different marking has been assessed, the lowest marking must be applied, respectively. The validity of this EC-type examination will expire, if the manufacturer modifies the safety-relevant properties of this product with comparison to the tested one or if the requirements in the standards or technical rules will be revised and/or tightened. Name, address and identification number 1883 of the notified body ECS must be indicated in the information brochure of this product.

ECS GmbH Notified Body 1883 13/10/10

Tchaler Ing Herbert Schäfer



ECS GmbH – European Certification Service Augerischulz und Persönliche Schutzausrüstung Laserschulz und Optische Messtechnik

Fig. 11.1.16 b) Laser Goggle Type Certificate



Fig. 11.1.16.c) Information supplied with goggles

11.1.17. Keys

The 'Strawberry' and the 'Strawberry & Cream' low level laser systems are manufactured and supplied with 2 keys to help prevent accidental or unauthorized use.

The clinician must insert and turn the key in order to operate the unit.



Fig. 11.1.17 a) Keys

11.1.18. Presentation Box



Fig. 11.1.18 a) Cardboard Box with top tier removed to show control unit



Fig. 11.1.18 b) Top tier within cardboard box for cluster probes, paddles, leads etc



Fig. 11.1.18. c) Detail of individually packed items

11.2. Usability

The Following Usability file has been conducted using the 60601-1-6:2008 Standard

Doc No: UF/01/V1	Usability File	"STRAWBERRY"
Page: 1 of 4 Date : 03/04/2012	Strawberry	The fastest growing International Beauty & Aesthotic Brand
	Medical Device	1

EN 60601-1-6: 2008 Documentation of Compliance

Author / Reviewer: Lisa Ormrod Signed

- 1. Application Specification
- 1.1 Medical Purpose

The Strawberry medical device is a non invasive class 3b cold red laser, the device is used for the purpose of Pain relief, inch loss and for the temporary reduction in the waist circumference.

1.2 Subject Population

The Strawberry Medical device is for use on adults only, and must not be used on people that suffer from any conditions listed as contraindications. (See Medical Questionnaire Document No. FRM-MED/001/Issue1)

1.3 Treatment areas

The Strawberry Medical device can be used on any area of the body with the exception of the middle and upper facial areas, due to the vulnerability of the retina in the eye.

1.4 User Profile

Therapists and clinicians can use the Strawberry Medical Devices, providing they have received training from the manufacturer or appointed distributors. (See Training schedule Document No. FRM 025 and certificate profile)

1.5 Operational environment

Treatments should be carried out in a closed area to protect the subject's modesty. The area should not be exposed to any people that are not aware of the need for protective eye protection. (See Eye protection specification Page 3 User Manual)

1.6 Operating Principle

The laser paddles need to be in contact with the skin, prior to the lasers being turned on. Once the laser paddles are in contact with the skin, the laser beams will be of no danger in the immediate proximity, and the beams will penetrate to a depth of no more than 13mm into the body. The treatment duration is selected by the technician and will not exceed 10 minutes in any single area. (See Training schedule Document No. FRM 025).

Doc No: UF/01/V1	Usability File	"STRAWBERRY"
Page: 2 of 4	Strawberry	The fastest growing international Beauty & Acetholic Brand
Date : 03/04/2012	Medical Device	170

- 2. Operational procedures:
- 2.1 Check that the machine is connected to a 13amp wall socket and the switch is on.
- 2.2 Check that the laser paddles are correctly configured (See User Manual).
- 2.3 Ensure that the keys are in the back of the machine and press the power ON button, followed by turning the key to the ON position.
- 2.4 The machine will now carry out a 'self test function' and is then ready to be used.
- 2.5 Treatment protocols to follow those listed within the Training Schedule.
- 2.6 After use procedures are to carefully place the laser paddles so that they can not be damaged, turn off the power switches and after turning the key to the off position, the operator should remove the keys.
- 2.7 The machine is fitted with a standard 13amp fuse in the power lead and 2 additional 1amp fuses located at the back of the machine. These fuses are to protect against power surges or fluctuations.
- 3. Risk Analysis:
- 3.1 Intended Use/Intended Purpose
 - See 1.1
- 3.2 User Profile
- See 1.4
- 3.3 Things that could go wrong

Sources: Clinical Evaluation Report, Risk Analysis Report, Risk Management file.

3.3.1 During Normal use:

a) Goggles MUST remain on subject and therapist's eyes at all times.

- b) Therapist must stay in the room whilst treatment is being carried out.
- 3.3.2 Use Errors:
 - a) Dropping machine or parts
 - b) Exceeding user time (e.g. over 10 minutes in one area)

Doc No: UF/01/V1	Usability File	"STRAWBERRY"
Page: 3 of 4 Date : 03/04/2012	Strawberry	The fastest growing International Beauty & Acethotic Brand
	Medical Device	and the second se

3.3.4 Hygiene

- 3.4 Protection protocol
 - In the event that the operator or client's protective goggles should be dislodged the machine should be immediately switched off. This can be easily done by pressing the "LASER STOP" button clearly marked on the machine.
 - The goggles MUST BE WORN AT ALL TIMES by the subject and therapist while the machine is turned on for treatments.
- 3.7 Resulting Hazardous situations and harms
 - If the machine is dropped you could damage either the machine or laser paddles. This would be identified within the self check feature upon tuning on the machine.
 - Ensuring that machine is thoroughly cleaned and sterilised between each client.

3.8 Preliminary review of the user interface concept The user interface concept clearly identifies the potential risk of laser light to the eyes. The Risk assessment confirms this potential danger. Any person in possession of the keys to the device, without the proper training, could potentially do damage to their eyes or those around them. This is why the key system has been applied to this device.

- Removing goggles: Lasers are damaging hence why goggles must be worn, if clients and clinicians were to briefly look at the laser light it would cause blinking for a few seconds.
 - 4. Strawberry medical device:
 - 4.1 General
 - a) MEDICAL DEVICE STRAWBERRY MEDICAL DEVICE
 - b) Basis

Intended use possible use errors hazardous situations or harms related to use context of use preliminary use scenarios

Doc No: UF/01/V1 Page: 4 of 4 Date : 03/04/2012	Usability File Strawberry	"STRAWBERRY" The Sectors growing International Beauty & Acetholic Brand.
	Medical Device	

4.2 Use Scenarios

- Worst case scenario to provide a basis for validation with subject and not the user:
- a) Treatment potentially being carried out in an open, unprotected environment.
- b) Untrained operator not providing the necessary eye protection goggles.
- 4.3 User actions related to Primary operating functions. (See Point 2)
- 4.4 User Interface requirements for the primary operating functions. (See point 2)
- 4.5 User Interface requirements for those use scenarios that are most frequent or related to safety.
- a) The whole procedure shall be easy to understand by reading the user manual.
- b) Text: English and appropriate language for the country of use.
- c) Font size 8pt Arial at a minimum
- d) Symbols within the whole procedure should be based on international standards
- e) Clearly understandable visuals are used within the User Manual.
- 4.6 Requirements for determining whether the primary operating functions are easily recognisable by the user.

a) To protect against lack of understanding regarding the operating functions for the device, each potential operator undertakes a test to confirm that they have clearly understood the operating procedures, prior to receiving certification.

11.3. Systems Level Description

The 'Strawberry' and the 'Strawberry & Cream' low level laser systems are controlled by Digital PCBs, powered by a switched mode power supply (SMPS), input and output information are the keypad and LCD Display, and the energy is delivered to the paddles and cluster probes through a remote inter lock module or the PCB interface.



Main Board


Strawberry&Cream System Block Diagram



Center for Devices and Radiological Health

11.4. Hardware

11.4.1 Laser

Operational rules to provide a safe environment of operation for the laser and other safety rules have been developed for using the Laser Lipo Ltd Strawberry and Strawberry and Cream models at "The Strawberry Clinic" which is co-located with their facility.

These rules identify the potential hazards and control measures used.

Laser Lipo Ltd appreciates that such rules have to be developed for each facility having due regard to the environment of use and is committed to providing appropriate training either directly or through their distributors to potential users to help them use the product safely.

11.4.2. LCD Display

Details of the touch screen display used follow, however we must stress these are confidential to the manufacturer or the component.

(The remainder of this page is intentionally left blank.)

11.4.3. Subject Contact

Part Name	Details	Type of Subject Contact	Material
Enclosure for control unit	see section 11.1.2	The clinician touches the display for a few seconds	thermoplastic elastomer
strap	see section 11.1.15	The straps are pulled taught around the clients body and are kept on for the duration of the treatment time	Nylon
paddle	see section 11.1.7	Paddles are placed against the selected fatty area, pressed against the skin and held in place with Velcro straps.	thermoplastic elastomer
Cluster probe	see section Error! Reference source not found.	Cluster Probes are placed against the selected fatty area, and pressed against the skin.	thermoplastic elastomer
goggles	see section 11.1.16	goggles are kept on the clients face for the duration of the treatment	Plastic: Nylon Lenses: Polycarbonate
LCD Display	see section <mark>0</mark>	The clinician touches the display for a few seconds	LCM Module Model AT070TN83 V.1 by InnoLux

11.3. Mechanical Drawings

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system control unit is contained within a plastic enclosure weighing approximately 14 lbs. The dimensions of the unit are approximately 8" x 10" x 12".

Cut outs and holes are made in the plastic enclosure for the membrane keypad (except for the Strawberry & Cream low level laser system), the emergency stop, the key switch, the leads connections and the air cooling requirements.



11.4. Photographs of Exploded Assemblies

11 Device Description



January 31, 2013

STRAWBERRY CLUSTER PROBE



8



11 Device Description
11.6. Training and Documentation

Laser Lipo Ltd is committed to offering training and supports either directly or through their US distributors. Part of this process will be to provide a set of forms to facilitate recording of treatment and results. These forms will be developed in co-operation with distributors and other marketing partners.

We include the type of documentation currently in use in the Europe.

11.6.1. Consent Form and Medical Questionnaire

	STRAWBERRY" aser Lipo Inch Loss CONSENT FORM	I	L	STRAWBERRY" aser Lipo Inch Loss	
Title: (Mr/Mis/Ms/Miss)	GP Name & Surgery Name				
Client Name:	GP Contact Number:	-	Please list any / all medications the	at you are currently taking:	
Address:	Tel. Home:				
	Tel. Work:				
	Tel, Mobile:		Have you ever exper	tienced any of the following specific condi-	tions?
	Emp2 Address		(P)	hase circle where appropriate)	
	eman Address:	8	Contraction	NOVER .	
ost Code:	Age: [] Gende	r: [Male] [Female]	Ludoetes .	NONTES	
			Kan Line Berkhene	NOTES	
uly authorize the technicians of	to perform the Laser Lipo Inch L	ose procedure for the	May Liver Problema	NUTES	
apose of spot fat reduction and skin tighten	ing. I am aware that clinical results may va	ry depending on individual	Any Kidney Problems	NOMES	
stors, including medical history, patient con	npliance with pre/post treatment instruction	s and individual response to	sAuto immune disease	NOMES	
ament. I have been made aware that my d	set and the amount of expercise I do, will have	we a major enection the	Currently Pregnant	NOMES	
suits of my realments, it i do not make an	emort to address my diet and excercise	Fam aware that the	Gastric Olders	NOMES	
suits will not be not be recained.		122216	Any form of infection, fever or disease	NOPPES	
nderstand that treatment with the Laser Lip	o machine involves a course of 8 treatment	its. The fee structure has	Cardio Vascular Conditions	NOPPES	
en fully explained and I understand that I a one place. I am fully aware that should I wis	m required to pay for a course of treatment shito cancel the course the outstanding treat	s, prior to any procedures tment value is non	Any condition currently treated by a Medical practitioner	NOTYES	
fundable. The course cost is £	(client's initials)	and a standard in the st	Thyroid problems	NOMES	
			Any metal pins or plates	NO/YES	
ar to the demand for treatments, all 8 ap	pointments are scheduled in following t	te initial consultation, 1	Muscular / skeletal problems	NO/YES	
ve been made awars that all cancellation	na require a minimum of 24hrs notice. F	eilure to do so will result in	Digestive problems	NO/YES	
at treatment being deducted from my co	urse without a refund. I am aware that th	is may have a negative	Circulation problems	NO/YES	
lect on the overall results. Any changes	to the initial treatment dates will be sub	ect to availability.	Gynaecological problems	NOMES	
and so which the second s		and the second second	Immune system	NO/YES	
SCHOOL OF STOCHE LEVEL	11 10 88 85 6	82. 9 2 25	A.S.2 (LIFE STYLE QUESTIONS:	
entity that I have been fully informed of the	nature and purpose of the procedure, expe	cted outcomes and possible	Course have require periods	INAVES1	
mplications. I understand that no guarantee	e can be given as to the final result obtained	L I am fully aware that my	Do you work at a consoler?	NOVES	
nomon is of a cosmetic concern and that th	e decision to proceed is based solely on m	vexpressed desire to do so	The year and regular media?	NOVES	
nderstand that it is my personal responsibil	ity to inform the clinician of any changes to	my medical history during	Do you get in a humu?	NOVES	
a course of Laser Lipo treatment sessions a	and I confirm that should this occur I shall a	dvise the clinician of any	Do you exercise?	NOVES	
anges.			Do you suffer allemies?	NOVES	
			How would you mark your current stress	100017	
onsent to the taking of photographs and au	filorize their anonymous use for the purpos	es of medical audit.	Enter date of last visit to doctor:	INOMES	
turnered and branchester			Additional conditions not listed? (PI	ease list below):	
settify that I have been given the opportunity intents of this consent form.	to ask questions and that I have read and	fully understand the			
lient signature:					
ate//			Print name:	Signature:	30
átness:					
			Date: / /		

11.6.2. Treatment Forms



12. Substantial Equivalence Discussion

SECTION 12 TABLE OF CONTENTS

12. Substa	ntial Equivalence Discussion	12.1
12.1. Sul	ostantial Equivalence Summary	12.2
12.2. Tab	oular Comparison of Technical Specifications	12.4
12.3. Tat	oular Comparison of Features	12.6
12.4. Det	ailed Comparison to Predicate Devices	12.7
12.4.1.	Control Unit	12.7
12.4.2.	paddles	12.10
12.4.3.	probes	12.11
12.4.4.	Paddle lead	12.12
12.4.5.	Probe lead	
12.4.6.	Straps	12.14
12.4.7.	Goggles	12.15
12.4.8.	Display	12.16
12.4.9.	Treatment	12.17
12.4.10.	User manual	12.19
12.4.11.	Software	12.41

12.1. Substantial Equivalence Summary

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are substantially equivalent in design, use and materials to the:

Chromogenex Technologies Ltd I-Lipo System – K111501

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are made of the same materials as the Chromogenex Technologies Ltd I-Lipo System – K111501.

- thermoplastic control unit enclosure
 - o proprietary components and assemblies used within the unit
- thermoplastic paddles and cluster probes
- nylon covered "velcro" fixed straps
- nylon/polycarbonate goggles

Both the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd i-Lipo System:

- use laser diodes
 - o of approximately 660 nm wavelength, and
 - Up to 50 mW output
- incorporated into
 - multiple multi-diodes "paddles" attached to the patient with special straps
 Strawberry 4 to 10 paddles with 6 diodes each
 - Strawberry 4 to 10 paddles with 6 diodes each
 - i-Lipo 4 pads (paddles) with 9 diodes each
 - \circ $\,$ and two cluster probes with one diode each
- have a similar control unit
 - o lcd patient display
 - o membrane key pad inputs
 - incorporated into display as a touch screen for strawberry and cream
 - o microprocessor controlled through software
 - multi-range power supply (EU and US voltages and frequencies)
- have a similar range of connecting cables
- have the same laser protection goggles
 - o different name printed on outside of frame

The key parts of the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd i-Lipo system are compared photographically in section 12.4. The systems compared in section 12.4 are both the European Systems however the US labeling for the Laser Lipo systems from Section 13 has been used.

The Laser Lipo Ltd Strawberry low level laser system, Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd I-Lipo System have the same intended use achieved through the same technology, are assembled from similar components made from similar materials, have similar user input and output interfaces and have similar features.

All systems use laser emitting diodes of similar wavelengths and powers: the Laser Lipo systems have the potential to use more paddles but with less diodes in each paddle. All systems have two cluster probes with a single laser diode in each. In all systems the paddles and cluster probes are secured to the subject with nylon encased rubber straps secured with "velcro".

All control units have plastic enclosures, discrete power supply, microprocessor controlled electronics, controls, alarms and similar control software.

Laser Lipo Ltd concluded that the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are substantially equivalent to the Chromogenex Technologies Ltd I-Lipo System, K111501.

12.2. Tabular Comparison of Technical Specifications

Feature	Laser Lipo Strawberry Subject Device	Chromogenex i-Lipo Predicate Device
		K111501
Electrical Requirements:	100 240	120 and 240 welt activities
Voltage	100 - 2400	120 and 240 volt settings
fuee rating		50 and 50 Hz respectively
Photographs in the user manual	TA quick-blow luse	SA Ceramic Tuse
section 12.4.10		
Laser Output:		
wavelength	660 ± 15 nm	650 – 690 nm
output	Up to 62 x 40 mW (± 15%)	up to 38 x 50 mW
*paddles	4 , 6, 8 or 10	4
laser diodes per paddle	6	9
Photographs in section 12.4.2		
probes	2	2
laser diodes per probe	1	1
Photographs in section 12.4.3		
Laser class	IIIb	IIIb
Laser Diode	GaAlAs	
Physical Dimensions:		222 462 442
control unit dimensions	2/5 x 185 x 315 mm	320 x 460 x 440 mm
Control unit weight	бкд	то ка
Photographs in section 12.4.1		
Temperature	$50 - 90^{\circ} E (10 - 32^{\circ} C)$	$10 - 30^{\circ}$ C
Humidity	30-90 (10 - 32 C)	10 - 50 °C
IP rating		IPX0
Storage temperature	$50 - 90^{\circ} F (10 - 32^{\circ} C)$	$5 - 30^{\circ}$ C
Storage humidity	50 50 (10 52 6)	35 – 65% RH
Photographs in section 12.4.10		
Safety Features:		
Prevention of unauthorized use	Key switch	Interlock
"Emergency" Stop	"Stop" button on front	"Stop" button on front
	panel	panel
Photographs in section 12.4.1		

*Laser Lipo uses the term paddle, where i-lipo uses the term pad for the device that comes in to contact with the body.

Feature	Laser Lipo Strawberry	Chromogenex i-Lipo
	Subject Device	Predicate Device K111501
Indications for Use Statement	The Low level Laser model Strawberry and Strawberry and Cream can be used for the Non-invasive aesthetic treatment for the temporary reduction in	The Chromogenex Technologies Limited i-lipo Tm Low Level Laser System is indicated for Non-invasive aesthetic treatment for
Strawberry, see Section 4 FDA Database K111501	waist circumference.	the temporary reduction in Circumference of the waist.

12.3. Tabular Comparison of Features

	Subject Device	Subject Device	Predicate
	Churrenth a murr	Church a mark 0	
	Strawberry	Cream	1-цро
510(k)	THIS 510(k)	THIS 510(k)	K111501
Intended use	Temporary reduction in waist circumference by the disruption of adipocyte cells within the fat layer for the release of fat and lipids from these cells for non-invasive aesthetic use.	same	same
Enclosure	Plastic	same	same
Display	LCD	same	same
Laser type	Class 3b	same	same
Membrane Keypad	yes	touch screen	same as Strawberry
Electronic technology	yes	same	same
fuses	provided	same	same
control electronics	yes	same	same
emergency stop	provided	same	same
key switch	provided	same	same
Speaker	provided	same	same
adjustment	Keypad	touchscreen interface	same as Strawberry
power supply	100-240 v ac 50-60 Hz	same	similar 240 v ac 50 Hz 120 v ac 60 Hz
transformer	provided	same	same
Energy source	Laser Diode from 658 nm	same	same
Energy output	up to 50± mW	same	Same up to 50 mW
cooling requirements	Air cooled	same	same
paddle size	15.2 x 4.7 cm	same	similar 15 x 9.6 cm
Cluster probe size	6 x 2.4 x 4.3cm	same	similar 2.7 x 3.6 x 2.6 cm
Environment Requirements	10°C to 30°C Non-condensing humidity below 75% RH	same	same

12.4. Detailed Comparison to Predicate Devices

12.4.1. Control Unit



Figure 12-1 - Strawberry & Cream Low Level Laser Control Unit



Figure 12-2 - I-lipo Control Unit



Figure 12-3 - Strawberry & Cream Low Level Laser Control Unit back side



Figure 12-4 - I-lipo Control Unit back side



Figure 12-5 - Strawberry & Cream Low Level Laser Control Unit underneath



Figure 12-6 - I-lipo Control Unit underneath

12.4.2. Paddles



Figure 12-7 – Strawberry/Strawberry and Cream paddles



Figure 12-8 - I-lipo paddles

12.4.3. <u>Cluster probes</u>



Figure 12-9 – Strawberry/Strawberry and Cream cluster probes



Figure 12-10 - I-lipo cluster probes

12.4.4. <u>Paddle lead</u>



Figure 12-11 – Strawberry/Strawberry and Cream paddle lead



Figure 12-12 - I-lipo paddle lead

12.4.5. Cluster probe lead



Figure 12-13 – Strawberry/Strawberry and Cream cluster probe lead



Figure 12-14 - I-lipo cluster probe lead

12.4.6. Straps



Figure 12-16 – Strawberry/Strawberry and Cream straps



Figure 12-15 - I-lipo straps

12.4.7. <u>Goggles</u>

The goggles we have shown below are exactly the same model from the same supplier as those used by Laser Lipo Ltd. They have a different name printed on the frame.



Figure 12-17 - Goggles

12.4.8. <u>Display</u>



Figure 12-18 - Displays: Strawberry (left) and Strawberry and Cream (right)



Figure 12-19 - i-Lipo Display

12.4.9. Treatment



Figure 12-20 Strawberry (left) and Strawberry and Cream (right) treatment mode displays



Figure 12-21 i-Lipo treatment mode display



Figure 12-22 - Strawberry Abdomen treatment



Figure 12-23 - I-lipo Abdomen treatment

January 31, 2013

Strawberry and Strawberry Cream Low level Laser Systems Left hand side: Laser Lipo Ltd 12.4.10. <u>User manual</u>

Right hand side: Chromogenex i-Lipo K111501

User Manual

For the

- Strawberry low level laser system model ILO, and, . .
- Strawberry & Cream low level laser system model SC

User Manual. Abover A00.1784 A00.1200



"STRAWBERRY" Model



"STRAWBERRY" & Cream Model



CE Low Voltage Directive 200695/EC

Copyright Chromogenex** © 2004 All Rights Reser

Page 12.19 HLIPO USER MANUAL Doc

12 Substantial Equivalence Discussion

January 31, 2013

USER MANUAL INDEX

Contents

Ŧ	Indi	cations for Use
2	Con	traindications
3	Stol	age
4	Wa	mings
5	Pre	cautions
9	Pre	paration
	6.1	Receiving and Opening5
	6.2	Assembling Your System7
2	n	lerstanding Your System7
	7.1	Laser Theory
	7.2	How It Works
	7.3	Training
	7.4	Laser Warning Signs and Safety
	7.5	Laser Machine Symbols and Specification
00	lnst	ructions for Use 11
6	Ma	ntenance12
	9.1	Laser Servicing and Error Messages12
	9.2	Cleaning13
	9.3	Routine Maintenance
	9.4	Changing the Fuses

Olipo

chromogenex.

I S	kafety Warnings.	3
1.1	Optical Safety	4
1.2	Electromagnetic Compatibility	v
1.3	Electrical Safety	9
1.4	Labelling	1
2 B	Introduction	10
21	Listing of Controls	п
2	Technical Specifications	13
3 7	Transportation and Installation	14
3.1	Transport Damage	11
3.2	Environmental Transport & Storage Conditions	11
3.3	Installation	14
3.4	Fitting the Stand	14
4 S	System Operation	16
I.t	Switch On Procedure	16
4	Setting Treatment Time	17
43	Initial System Set - Up & Service Information	18
5 6	Seneral Maintenance	20
5.1	Visual Inspection	20
5.2	Cleaning and Disinfecting	20
5.3	Changing the Fuses	21
6 F	Fault Diagnostics / Servicing	22
6.1	Safe Disposal	8
7 T	Freatment Information	23

chromogenex.

5 Precautions

- Suitable protective eye-wear should be used at all times when the Low level Laser model Strawberry and Strawberry and Cream is in use
- suitable goggles are supplied with the Low level Laser model Strawberry and Strawberry and Cream
- Suitable "over-goggles" are available for those who wear prescription glasses
- Keep the spare key secure to prevent unauthorized use
- Remove the operating key and store securely when not in use
- Access to and viewing of the treatment area should be limited while the laser is in operation to those wearing suitable protective eye-wear.
- Always check the device and leads for damage before use and if unsure consult a qualified electrician.

Olipo

1 Safety Warning

The ChromogenexTM it is a low level laser system intended to be used for body contouring.

Personnel operating and maintaining the iLipo⁷¹¹ system should be familiar with the safety information provided in this section.

Any laser light emitting device can cause injury if used improperly. Use of controls, adjustments or the performance of procedures other than those specified herein may result in hazardous exposure to light radiation. No attempt should be made to operate the unit until the User Manual has been read and fully understood. In addition, a clear understanding of the biological effects of the interaction of laser light with tissue should be prerequisite to the use of this system.

When not in use the iLipo $^{\rm TM}$ should be protected from unqualified use by removal of the key from the keyswitch.

IT IS STRONGLY RECOMMENDED THAT ALL OPERATORS OF THE ILIpo™ ATTEND THE SUPPLIERS' APPROVED TRAINING COURSE PRIOR TO THE USE OF THIS SYSTEM. FURTHER INFORMATION ON CHROMOGENEXTW PRODUCTS CAN BE VIEWED AT:-

http//: www.chromogenex.com

HLIPO USER MANUAL DOC

Page 12.21

m

4 Warnings

The only adverse warning is that goggles must be worn throughout the duration of the treatment. As long as protocols are followed there is no danger. Exposure of a class 3b laser directly to the eye can cause damage to the retina. This is the only Warning.

The following Laser Warning Signs are included with the Low level Laser model Strawberry and Strawberry and Cream:



Olipo

chromogenex.

1.1 Optical Safety

- Clearly identify the treatment room.
- The area around a laser light source must be secure to protect other persons while treatment is in progress. Entry must be restricted while the system is in use.
- Allow access to the treatment room only to personnel essential to the procedure and well trained in the required safety procedures.
- Ensure that all treatment room personnel are familiar with the system controls and know how to shut it down instantly.
- Appropriate protective evewear must be worn by everyone within the controlled area whenever there is a risk of exposure to the light radiation.
- Never direct the light output at anything other than the intended treatment site, especially at reflective objects such as metal surgical instruments.
- Never look directly into the light emitted from the treatment pads or lymphatic stimulator probes even when wearing protective evenear.

WARNING

IT IS THE RESPONSIBILITY OF THE USER TO ENSURE THAT THE PATIENTS EYES ARE PROTECTED DURING TREATMENT

Energy Output

The maximum energy output can be up to 38 x 50mW

12 Substantial Equivalence Discussion

I-LIPO USER MANUAL DOC

Page 12.22

4

January 31, 2013

Electromagnetic Compatibility

The Laser Lipo 'Strawberry' and 'Strawberry' & Cream systems Have been independently tested by BWS testing laboratory to Meet the following requirements.

Immunity to EN60601-1-2

- EN6100-3-2: Mains Harmonics
- EN6100-3-3: Voltage Fluctuations
- EN6100-4-2: Electrostatic Discharge
 - EN6100-4-3: Radiated Immunity
- EN6100-4-4: Electrical fast transients
 - EN6100-4-5: Surge Immunity
 - EN6100-4-6: Conducted RF
- EN6100-4-11:Voltage Interuption

Emission to EN60601-1-2

- EN55011 Class B Radiated Emissions •
- EN55011 Class B Conducted Emissions

The above shows that the systems have undergone vigorous harmful interference when the equipment is operated in a testing to show the protection levels maintainable against commercial environment.

Please contact Laser Lipo Limited directly if you require further guidance.

© lipo

chromogenex

1.2 Electromagnetic Compatibility

The iLipoTM has been tested by an independent, accredited testing laboratory and found to meet the requirements of:

Immunity to EN60601-1-2

- EN 61000-3-2: Harmonic Distortion
- EN 61000-3-3: Voltage Fluctuations and Flicker EN 61000-4-2: Electrostatic Discharge (ESD) EN 61000-4-3: Radiated Immunity EN 61000-4-5: High Voltage (HV) Surges EN 61000-4-6: Conducted RF

- EN 61000-4-11: Voltage Interruption

Emission to EN60601-1-2

- EN55011 Class B Radiated Emissions
 EN55011 Class B Conducted Emissions
- generates uses and can radiate radio frequency energy, which, if not installed and used in accordance with the User Manual, may cause interference to other equipment. If in doubt about possible interference to other equipment, please contact your distributor or Chromogenex¹¹ direct equipment is operated in a commercial environment. Like all similar equipment, this product These limits are designed to provide reasonable protection against harmful interference when the or guidance.

Electrical Safety

The Laser Lipo 'Strawberry' and 'Strawberry' & Cream systems Has undergone rigorous testing to prove the safety and efficacy of the machines and are accredited to 60601-1-6

Laser Lipo Limited are proud of their accreditations and ensure that the best safety features are provided. To ensure no risk to the operator the exterior housing should only be removed by a Laser Lipo approved service engineer.

Olipo

chromogenex

1.3 Electrical Safety

Chromogenex^{nu} has made every effort to provide as many safety features as possible to assure personal safety. To be effective, these safety features have to be utilised and not ignored.

Therefore to ensure that there is no risk to the operator, no part of the exterior housing should be removed except by Chromogenex^m approved service engineers.

ONLY MANUFACTURERS RECOMMENDED SPARES AND ACCESSORIES SHOULD BE USED WITH THE SYSTEM. NON STANDARD COMPONENTS AND ACCESSORIES MAY DEGRADE PERFORMANCE AND CREATE SAFETY HAZARDS

12 Substantial Equivalence Discussion



January 31, 2013



chromogenex.

7.5 Laser Machine Symbols and Specification

retok.	Decemption
3	Manufacturer date represented as YOYY-MM
€	Type EF appled perts - provides degree of electrical protection against electric shock with isolated or facebrg apples parts
\triangleleft	Caution - there are specific warnings or precedians associated with the clarice.
	Consult Operating Instructions

Symbols Used

SYMBOL	DESCRIPTION	LOCATION
0	Protective Earth (ground).	Internal
\odot	IEC 471-5007. On (power: connection to the mains).	External – Top Moulding
•C	Off (only for part of equipment).	External – Top Moulding
•	IEC 878-02-02. Type B Applied Part	External – Manufacturer Label
ф	IEC 417-5016. Fuse Location.	External – Manufacturer Label
1	IEC 417-5032. Alternating current	External – Manufacturer Label
\mathbb{V}	IEC 348. Attention, consult accompanying documents.	External – Manufacturer Label
5	IEC 471-5007. On (power: connection to the mains).	External – Rear panel
0	IEC 471-5008. Off (power: disconnection from the mains.	External – Rear panel
T5H500V	Fuse Type. Timed 5 Amp ceramic fuse.	External – Manufacturer Label
3	IEC 878-03-04. Non-ionising radiation	User Manual
\bowtie	EN 50419. Separate collection for safe disposal.	External – Rear Panel
0	Emergency Stop The Emergency Stop is the red button on the front of the iLipo [™] . When pressed it immediately disables the system. This button should be depressed only in the event of an emergency. The system can then be reset by rotating the button in the direction of the arrows.	External – Top Moulding

12 Substantial Equivalence Discussion

HLIPO USER MANUAL DOC

~

January 31, 2013





Page 12.26

12 Substantial Equivalence Discussion

January 31, 2013

chromogenex.

1 Indications for Use

the disruption of adipocyte cells within the fat layer for the release of fat and lipids from these cells for The Low level Laser model Strawberry and Strawberry and Cream can be used for body contouring by non-invasive aesthetic use.

2 Introducti

Olipo

7 Understanding Your System

loss and body contouring. The system emits low levels of laser energy (wavelength 550mn - 650mn - 650mn) to predetermined larget sites resulting in disruption of fat cell membranes. Triglycerides spill out from the trooken cell membranes and are released into the intensitial space where they are slowly transported through the body's natural metabolic functions. The ILipo** is a state of the art laser device, intended to be used for lipolysis in order to achieve inch

7.1 Laser Theory

The term laser is derived from: A mplification by the Light

S timulated

E mission of R adiation

produce a coherent light source at a wavelength of 660nm in the "paddles" which lay on the patient's The Low level Laser model Strawberry and Strawberry and Cream utilities a semi-conductor diode to skin during treatment.

7.2 How It Works.

When the laser paddles are switched on, the laser light peretrates the skin and creates a reaction, see the diagram below:

"Strawberry"Penetration Depth



The Laser Beams have an active penetration of 9mm hur continue deeper up to 13mm but with little or no effect

Adipocyte Fat Colls







Party of the state

12 Substantial Equivalence Discussion

























































































The system is user friendly and utilises a membrane control panel with graphical LCD display and 3 control buttons which are used to acroil through the memus and select various options an detailed

The it poin has been designed with user selectable software, ergonomic fightweight laser treatment

pads and lymphatic stimulator probes.

later in the manual.

2

HLIPO USER MANUAL DOC



























January 31, 2013

6.2 Assembling Your System

- Unpack the contents of the cardboard box and check no items are missing -
- 2. Lay the Velcro straps on a table
- Lay out the laser paddles on the Veicro straps. Make sure the "end paddle" is the last in line; it has only one socket so is easy to identify m
- Connect the long lead between the front larger socket on the left hand side of the base of the unit and the first paddle 4
- Connect the two sets of 5 paddles with the longer lead between paddles 5 and 6. ui ui
 - There is only one connection to the last paddle, paddle 10.



The 2 single diodes connect with dual lead to one plug. This fits into the second, smaller sockets on the left hand side of the base of the unit. ň



- Connect the power lead between the rear of the base of the main unit and a suitable supply. 00
- Insert the key into the lock at the rear of the base. 6
- 10. Give one final check that everything is securely connected and then if you are ready to power up the system for use, turn on the power switch at the rear of the base.

7 Understanding Your System

The term laser is derived from: 7.1 Laser Theony A mplification by the S timulated L ight

Olipo

chromogenex

Rear of Lipo ** System



HUPO USER MANUAL DOC:

2

12 Substantial Equivalence Discussion

Health	
ogical	
Radiolo	
s and	
Devices	
for [
Center	

Technical Specifications

STRAWBERRY UNIT:

ower supply:	100-240 V~
power consumption:	Max 60W Di-
nensions:	275 x 185 X
315 mm Weight	6 kilos X 315
nm LCD Display	20 x 4 LCD

Red Laser Probe 660nn 40mW Visible

A STATE OF A		1
erance of Laser Diodes nent of uncertainty	<±15% Stated Probe Label Output Power 5%	
n measured quantities	0%. Diodes are the main source for potential increases in measured quantities over time. Laser diodes suffer negligible	
	obvious degradation until they fail.	
th	660nm ±15 nm	
tput	40mW	
	120mm	
ergence	9 x 38 degrees Typ	
55	38	
edium	GaAIAs	

Olipo

chromogenex

2.2 Technical Specifications

Laser Type:	Laser Diode
Wavelength:	650mm - 680mm
Energy Output:	Up to 38 x 50mW
Mains Power Output:	100VA
Laser Treatment Pad Size:	15cm x 9.6cm
Lymphatic Stimulator Probe Size:	S0em x 18em
Safety:	On board self-diagnostics
Gassification:	Electrical Class 1, Type B Applied Part
Dimensions:	32cm x 48cm x 44cm
Waight	10kg
Cooling Requirements:	Air Cooled
Electrical Requirements:	240V / 5A / 50Hz - Model A00-1184
Fusos:	SA Ceramic Timed
Mode of Operation:	Continuous
System Status & Fault Diegnostics:	On board computer lased self- diagnostics

10 On board con diagnostics

ingress Rating:

0Xdi

Environmental Requirements:

10°C to 30°C Non-condensing trumidities below 75% RH

Equipment Included:

2 x Lipo Keye 1 x User Manual CD 1 x Theshment Guidelness CD 1 x Mants Lead 4 x Later Treatment Pacts & Cabless 2 x Tohrbotic Stimulatur Probes & Cables 2 x Tohrbotic Ceramic Fuses 1 x Theshment Pack Waist Bett (119cm) 2 x Treatment Pack Waist Bett (119cm) 2 x Treatment Pack Am Seas (47cm) 2 x Treatment Pack Am Seas (47cm) 2 x Treatment Pack Lag Betts (82cm) 1 p Stand

13

TUPO USER MMMULL DOC

Page 12.29

12 Substantial Equivalence Discussion

January 31, 2013

6.2 Assembling Your System

- Unpack the contents of the cardboard box and check no items are missing
- Lay the Velcro straps on a table N
- Lay out the laser paddles on the Velcro straps. Make sure the "end paddle" is the last in line; it has only one socket so is easy to identify m
- Connect the long lead between the front larger socket on the left hand side of the base of the unit and the first paddle 4
- Connect the two sets of 5 paddles with the longer lead between paddles 5 and 6. ui ui
 - There is only one connection to the last paddle, paddle 10.



The 2 single diodes connect with dual lead to one plug. This fits into the second, smaller sockets on the left hand side of the base of the unit. ň



- Connect the power lead between the rear of the base of the main unit and a suitable supply. 00
- Insert the key into the lock at the rear of the base. 6
- 10. Give one final check that everything is securely connected and then if you are ready to power up the system for use, turn on the power switch at the rear of the base.

7 Understanding Your System

The term laser is derived from: 7.1 Laser Theory A mplification by the S timulated Light



chromogenex

3 Transportation and Install

3.1 Transport Damage

delivery, please check that the external packaging is not damaged. It signs of damage are present, please report them to the shipping carrier and your distributions Customer Service Department within 24 His. The LLpo* has been carefully packed and should arrive at its destination in perfect condition. Upon

3.2 Environmental Transport & Storage Conditions

The transport and storage conditions for the iLipo** are as follows:

+5°C to +40°C 35% to 65% Non-Condensing. 500 - 1060 hPa Atmospheric Pressure Relative Humidity Temperature

3.3 Installation

pads and lymphatic stimulator probes by plugging into the relevant connector sockets at the rear of the unit. Both the treatment pads and probes rest in the hotster on the top of the unit when not in Carefully remove the packaging from around the unit and unpack the treatment pads and lymphatic cooling of the internal systems with a verited underside. It should be placed either on a firm surface allowing a passage of air undemeath and a distance of approximately 20cm left dear around the unit stimulator probes. The ILipoTM is a compact portable system and has been designed for optimum to allow adequate vertilation or fitted to the it por stand. (See Section 3.4). Fit the isser treatment use. The system can then be plugged directly into a standard mains accket.

Fitting the Stand 3.4

The 4Lpor* stand is fitted with 4 castors, the front 2 with brakes. Before fitting the ILpor*, ensure that the brakes are applied by pushing the lever downwards.





ILLPO USER MANUAL DOC

2

12 Substantial Equivalence Discussion

January 31, 2013

Olipo

chromogenex.

The Laser Lipo Ltd Strawberry and Strawberry and Cream low level laser systems are designed to be placed on a "table-top" and do not have a special stand. Care must be taken not to strain the paddles main cable.









HUPO USER MANAALIDOG

볋

12 Substantial Equivalence Discussion

8 Instructions for Use

8.1 Strawberry Low Level Laser System

- 1. Set up the system as described in section 6.2
- ei.
- The screen will show the default treatment time of 10 minutes. You may use the up and down Switch on the unit by using key and power switch, see section 6.2 ó
- The display "MENU>" indicates the menu can be accessed by the right arrow on the front panel. Follow the on-screen instructions to adjust the sound setting and total time. arrows on the front panel to alter this if desired. 4
 - Two indicator lights show that the power is connected and the laser is ready for use. wi.
- There is a "\$TOP" button in the middle of the front panel which will stop the unit when pressed at any time. ú

8.2 Strawberry and Cream Low Level Laser System

- Switch on the unit by using key and power switch, see section 6,2 Set up the system as described in section 6.2 ÷ ē.
 - m
 - Adjust the treatment time using the touch screen, see below:



4. Adjust the alarm settings using the touch screen, see below:



Olipo

chromogenex

4 System Opt

4.1 Switch On Procedure

To operate the iLpo** system, the following procedure must be followed

- Connect unit to well supply and switch ON ų,
- Switch mains power switch at the rear of the unit to the 1 (on) position. (See p.12) ÷,
- Ensure the emergency stop button is released (rotate clockwise). (See p. 11) m

- After switch on, the screen will display the iLipoth logo and nun through a self-diagnostics check screen. (See Figures 1 & 2).
- The software also provides the user with the option of an initial System Set Up (See Section 4.3)
- After the well disgrootics, the user will be presented with a treatment trine selection screen. (See Section 4.2)



12 Substantial Equivalence Discussion

January 31, 2013

8 Instructions for Use

8.1 Strawberry Low Level Laser System

- Set up the system as described in section 6.2
- The screen will show the default treatment time of 10 minutes. You may use the up and down Switch on the unit by using key and power switch, see section 6.2 ei.
- The display "MENU>" indicates the menu can be accessed by the right arrow on the front panel. arrows on the front panel to alter this if desired 4
 - Follow the on-screen instructions to adjust the sound setting and total time.
- There is a "5TOP" button in the middle of the front panel which will stop the unit when pressed Two indicator lights show that the power is connected and the laser is ready for use. wi. ø

at any time.

8.2 Strawberry and Cream Low Level Laser System

- Set up the system as described in section 6.2 ÷.
- Switch on the unit by using key and power switch, see section 6.2 Switch on the unit by using key and power switch, see section
 Adjust the treatment time using the touch screen, see below:



4. Adjust the alarm settings using the touch screen, see below



12 Substantial Equivalence Discussion

Page 12.33

HLIPO USER MANUAL DOC

11

chromogenex.

4.2 Setting Treatment Time

Olipo

NOTE: EACH LASER TREATMENT PAD CONTAINS TWO SKIN SENSORS IN THE AREAS SHOWN IN FIG.3. THE ILIDO UNIT MUST BE SWITCHED ON BUT NOT ARMED BEFORE THE PLACEMENT OF THE TREATMENT PADS IN ORDER FOR THE SKIN SENSORS TO SWITCHING ON WHEN THE SYSTEM IS ARMED. IF THIS OCCURS, THE TREATMENT PADS MUST BE REMOVED FROM THE BODY AND RETURNED TO THE HOLSTEN. THE MACHINE MUST BE SWITCHED OFF AT THE KEYSWITCH AND THEN BACK ON. THE TREATMENT PADS CAN THEN BE REPOSITIONED ON THE BODY BEFORE ARMING THE SYSTEM. CALIBRATE CORRECTLY, FAILURE TO DO THIS WILL RESULT IN THE DIODES NOT



- Position the laser treatment pads and the lymphatic stimulator (LS) probes as per treatment guidelines.
- Use the AV arrows to select the required treatment time. Up to a maximum of 55 minutes. (Fig.4.) 2



Fig.4



Olipo

chromogenex

the iLipoTM is armed. This will be followed by a 2 second delay before treatment commences. The treatment time will then count down to zero which is when the laser pads and LS probes switch off. (Fig.5.) Press the
 button to arm the system. An audible alert will then be heard indicating that e



Fig.5.

Treatment may be paused at any time by pressing the
button (Fig.6.) and pressing it again to resume. Pressing either of the A V will terminate the treatment and reset the timer to the selected treatment time. (Fig.7.) 4



•0 When treatment is complete, turn off the iLipo by turning the key anti-clockwise to the position and removing it from the system. 5

Initial System Set - Up & Service Information 4.3

technology, to this end; the following options have been made available to the user. When the system is switched on, an audiale beer will be heard which lasts for 3 seconds. During this time, the operator can enter the system Set-Up Menu by holding down both arrow buttons on the control panel membrane at the same time. NOTE: This option is an initial set-up and is not required each time the iLipo[™] is switched on. The iLipoTM has been designed to be extremely user friendly whilst incorporating the latest

The following menu will be displayed: (Fig.8)

HLIPO USER MANUAL DOC

10

Page 12.34

12 Substantial Equivalence Discussion

THIS PAGE LEFT PURPOSLY BLANK
January 31, 2013	Olipo	SETTINGS SERVICE EXIT
Center for Devices and Radiological Health		

chromogenex

THIS PAGE LEFT PURPOSLY BLANK



Fig.8.

The user can select any of the above options by scrolling up and down using the arrow buttons on the membrane panel. The selections available are as follows:

SETTINGS

- Volume. (User can select volume of the iLipoTM safety buzzer).
- Defaults. Resets factory defaults.
- The user can select either of the above settings by selecting the SAVE & EXIT function and pressing the ENTER button. This will return to the main menu screens as seen in Section 4.2

SERVICE

(In the unlikely event that the iLipoTM develops a fault, the user may be required to provide information from this function to the Service Engineer.)

- Shot Count
 Displays total time in minutes of treatments carried out.
- Temperature Displays the temperature inside the treatment pads and main unit.
- Software Displays the revision number of the software programmed into the it jpo[™].

EXIT

- Selecting Exit will return to the main menu screen as seen in Section 4.2

12 Substantial Equivalence Discussion

Page 12.35

19

HLIPO USER MANUAL DOC

9 Maintenance

9.1 Laser Servicing and Life

The machine should be serviced every 6 months or if only used infrequently maybe serviced annually

Like any light emitting device, the light output from each laser diode will decrease with time/use. It is important to service the equipment regularly so that compensating adjustments can be made. Machines are set when new to achieve the specified output using approximately 60% of the maximum capacity, hence this utilization can be increased to compensate for the time-based reduction in light output. Machine life is thus anticipated to be between 3 and 5 years.

Laser Lipo will finalize service arrangements with their US distributors and include in the final manual.

9.3 Cleaning

We advise that after each use the paddles should be cleaned with a "sterile wipe" and that the probes are sterilized.

9.4 Routine Maintenance

There are no consumables needed and no routine maintenance is required.

January 31, 2013

Olipo

chromogenex

.

5 General Maintenance

5.1 Visual Inspection

The mains, treatment pad and LS probe cables should be inspected for damage and wear and tear on a regular basis.

Do not pull on the cables when removing from the treatment pads and probes, unplug by holding onto the connector. In the unlikely event of damage to the treatment pads or LS probes, do not attempt to rectify the damage. Contact your Distributor for replacement parts.

5.2 Cleaning and Disinfecting

Before cleaning maintenance is carried out, please ensure the unit is disconnected from the mains supply. Note that no disassembly of the illiporm unit or associated parts is required for cleaning and disinfecting.

The illiport unit itself can be cleaned by dusting over with a clean, dry, soft, lint free cloth ENSURING THAT NO FLUIDS ARE USED.

The laser treatment pads and lymphatic stimulator probes should be carefully cleaned with an isopropyl wipe before each treatment and allowed to dry thoroughly before the system is switched on. The treatment pads and probes are non-critical products as per the Association for Professionals in Infection Control and Epidemiology. Inc. (APIC) Guidelines. Non-critical is defined as an item that comes in contact with intact skin but not with mucuous membranes. Therefore cleaning with isopropyl wipes in between each client is and that is required by the APIC. NOTE: DO NOT SUBMERCE THE TREATMENT PADS OR PROBES IN ANY SOLUTION. This may cause a hazardous condition for the operator and possibly the client and may also cause the system to malfunction.

12 Substantial Equivalence Discussion

HLIPO USER MANUAL DOC

8

Page 12.36

January 31, 2013

9.5 Changing the Fuses

9.5.1 Equipment Fuse

Grey Fuse unit located at the back of the machine.



PULL out the grey tuse holder from the machine. As shown.



Then holding the fuse assembly, unscrew the cap in an anti-clockwise direction.





Then pull out the fuse, 240v 1AMP.



Conce you have replaced the tase (spares are suppled with each new machine) Re-assemble the tase holder and usefully cush have an intuitient and an annotation of the two concer pilos, when putting that larged with the component prior to meeting it.



9.5.2 Power Supply Fuse

The power supply fuse is located between the power switch and the power lead socket at the rear of the base unit. Consult a qualified electrician if this fuse has blown.

Olipo

chromogenex

5.3 Changing the Fuses

The it joom is fitted with 2 external traves, which can be easily replaced if required (replacement traves are supplied). The fuse holders are located at the rear of the unit (See page 11.)

Using a small flat head screwdriver, undo the flaxe holder by unscrewing anticlockwise and remove from the recess as seen in Figs. 16.2.





Pull out the spent tuse from the holder (Fig.3) and replace with a new one of the correct rating.



F19.2

Screw the hase holder clockwise into the recess and tighten using the fiel head screwdriver

NOTE: The iLipo¹⁴ will only accept the following fuses: -

154H500V 32mm Farrula Fuse Time Delay, High Breaking Capacity, Caramic Body, (Size 6.) x 32mm).

12 Substantial Equivalence Discussion

HUPO USER MMMUL DOC

ñ

Page 12.37

9.2 Error Messages

"CHECK PADDLE" on Strawberry ILO model LCD Screen only:

- Turn off machine
- Disconnect from power supply
- Check all leads and connections to laser paddles and probes
- Check laser paddles are connected in correct order
- Reconnect to power supply
- Example Turn on machine

No light or messages on screen when unit is switched on:

- check power lead is fully inserted at socket on rear of machine
- check power lead is fully inserted and turned on at wall socket
- 3. check that the power switch is turned on, the "1" position
- If the above does not resolve the problem, turn off and disconnect from power supply then check the fuses, see section 9.5.

Olipo

chromogenex.

6 Fault Diagnostics / Servicing

This section will help the Service Engineer to diagnose faults that may be experienced on the iLipoTM. In the unlikely event that any of the following fault codes appear on screen, the user is advised to contact their Service Provider directly quoting the fault code displayed.

DVER TEMPERATURE
MENT PAD OVERTEMPERATURE
M CORRUPT, COMMUNICATIONS ERROR

NOTE: THER ARE NO USER SERVICABLE PARTS INSIDE THE ILIpo™ SYSTEM. SHOULD ANY FAULT ARISE, PLEASE CONTACT YOUR SERVICE PROVIDER.

REMOVAL OF SECURITY SEALS, OPENING THE SYSTEM, OR THE USE OF LIQUIDS ON THE PRODUCT WILL RENDER THE WARRANTY INVALID.

6.1 Safe Disposal

For safe disposal of the unit, please contact your Distributor.

HLIPO USER MANUAL DOC

Page 12.38

ដ

Center for Devices and Radiological Health

January 31, 2013

Treatment Information

practical certified training is provided by As part of Laser Lipo Ltd's protocols full Approved 'Strawberry' trainers.



chromogenex

7 Treatment I

Treatment probacels and graditical cartified training will be provided by Chromogenes¹¹ approved trainers. Users will also be provided with a copy of the Treatment Guidelines.



Chromogenes^m is a quality compary, developing, manufacturing and distributing medical and light based devices world-aide. We atme to meet global regulatory attantiants and the growing needs of the calciume:

Chromogenex¹ operates under a continuous improvement philosophy and herefore reserves the right to make amentments as appropriate without comprovising quality, safety or functionality.

12 Substantial Equivalence Discussion

Page 12.39

2

ILUPO USER MANAL DOC

Center for Devices and Radiological Health

January 31, 2013



chromogenex.

Distributed by:

HUPO USER MANUAL DOC

3

Page 12.40

Chronogene Tachroidiges Lit ta Chronogene * Units 1 & 2 Heal Fincepri Dare Calen Dare Calen Careuraneans S Wake UK S Wake UK S Make UK S Make UK F +44 (n) 154 TS644 F +44 (n) 154 TS644 F +44 (n) 154 TS644

Manufactured by:

12 Substantial Equivalence Discussion

12.4.11. Software

The Laser Lipo Ltd Strawberry low level laser system, Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd I-Lipo System are based on the same FDA guidance document "Guidance for the Content of Premarket Submissions for Software Contained in Medical Devices and comply the following standards:

- ISO 13485
- ISO 14971
- IEC 62304
- IEC 60601-1-1
- IEC 60601-1-2
- IEC 60601-2-22

13. Proposed Labeling

SECTION 13 TABLE OF CONTENTS

13. Propose	ed Labeling	13.1
13.1.	Draft Labels Attached to the Device	
13.1.1.	Draft Physical labeling	13.2
13.1.2.	Laser warning sign for treatment room door	13.5
13.1.3.	Draft Labeling in Software	13.6
13.2.	Cardboard Presentation Box	13.8
13.3.	Draft User Manual	13.10
13.4.	Training Materials	13.17
13.5.	Promotional Material	13.28
13.6.	Web-site and Other Electronic Media Materials	13.29

13.1. Draft Labels Attached to the Device

13.1.1. Draft Physical labeling

European labeling is shown here which form the draft for US labeling.

LASER LIPO Ltd			
MODEL NAME: POWER:	Strawberry Inch Loss 100 - 240v~ 2.0-1.OA, 50/60Hz		
LASER: SERIAL No.:	Class 3B SB200612080		
M DATE:	20/06/2012.		
MANUFACTURE: 0120 A Til Tel.: 0044 844 980 1820	Laser Lipo Ltd Heath House Edenbridge Kent, TN8 6ST. United Kingdom. strawberry-laser.com MADE IN U.K		

Descriptions will be added to EN 980 sysmbols e.g. "date of manufacture", "manufacturer", "consult users manual".



LASER LIGHT **AVOID EXPOSURE TO BEAM CLASS 3B LASER PRODUCT**

Manufactured by:



Laser Lipo Ltd Heath House Edenbridge Kent. TN8 6ST. United Kingdom. strawberry-laser.com 2011-11

Distributed by:

CLASS 3B LASER PRODUCT VISIBLE LASER RADIATION



AVOID EXPOSURE TO BEAM

6 X 40mW λ = 660nm

3B LASER PRODUCT

CLASS

LASER APERTURE



New US-style warning labels will be developed.



13.1.2. Laser warning sign for treatment room door

Figure 13-1 - Strawberry Laser warning sign for treatment room door

13.1.3. Draft Labeling in Software

Strawberry and Cream touch-screen version



Figure 13-3 - Strawberry & Cream Display Language settings mode

Language options and flags will be amended for the US product e.g. the "Stars and Stripes" will replace the "Union Flag" above "English".



Figure 13-4 - Strawberry & Cream Display Support mode



Figure 13-5 - Strawberry & Cream Display Alarm settings mode

Contact details will be amended to US details and/or US distributors once arrangements have been finalized.

13.2. Cardboard Presentation Box



Figure 13-6 – Packaging of the Strawberry unit



Figure 13-7 - Packaging of the Strawberry paddles, probes and leads

Note: photograph is of European paddles showing position of laser warning labels, but European rather than US labels.

13.3. Draft User Manual

User Manual

For the

- Strawberry low level laser system model ILO, and,
- Strawberry & Cream low level laser system model SC



"STRAWBERRY" Model







USER MANUAL INDEX

Contents

1	Indications for Use		
2	2 Contraindications		
3	Sto	rage	
4	Wa	rnings4	
5	Pre	cautions	
6	Pre	paration5	
	6.1	Receiving and Opening	
	6.2	Assembling Your System	
7	Und	derstanding Your System7	
	7.1	Laser Theory	
	7.2	How It Works	
	7.3	Training9	
	7.4	Laser Warning Signs and Safety9	
	7.5	Laser Machine Symbols and Specification	
8	8 Instructions for Use		
9	9 Maintenance		
	9.1	Laser Servicing and Error Messages	
	9.2	Cleaning	
	9.3	Routine Maintenance	
	9.4	Changing the Fuses	

1 Indications for Use

The Low level Laser model Strawberry and Strawberry and Cream can be used for body contouring by the disruption of adipocyte cells within the fat layer for the release of fat and lipids from these cells for non-invasive aesthetic use.

2 Contraindications

The Low level Laser model Strawberry and Strawberry and Cream are contraindicated for use in the vicinity of:

- open wounds
- metallic implants
- active implants, such as a "pacemaker" or "implanted defibrillator"

Further, the Low level Laser model Strawberry and Strawberry and Cream are contraindicated for patients suffering from:

- epilepsy
- urinary infections
- diabetes
- cancer
- medical edema
- kidney problems
- auto immune disease
- gastric ulcers
- cardiovascular conditions, such as: thrombosis, phlebitis, hypertension, hypotension, heart conditions)
- any other infections, fever or disease

The Low level Laser model Strawberry and Strawberry and Cream are contraindicated for use when pregnant.

The Low level Laser model Strawberry and Strawberry and Cream are contraindicated for minors (under the age of 21).

3 Storage

The Low level Laser model Strawberry and Strawberry and Cream should be stored securely in a dry place where they will not be exposed to extremes of temperature (less than 50°F or more than 90°F) for prolonged periods.

4 Warnings

The only adverse warning is that goggles must be worn throughout the duration of the treatment. As long as protocols are followed there is no danger.

Exposure of a class 3b laser directly to the eye can cause damage to the retina. This is the only Warning.

The following Laser Warning Signs are included with the Low level Laser model Strawberry and Strawberry and Cream:



5 Precautions

- Suitable protective eye-wear should be used at all times when the Low level Laser model Strawberry and Strawberry and Cream is in use
 - suitable goggles are supplied with the Low level Laser model Strawberry and Strawberry and Cream
 - o Suitable "over-goggles" are available for those who wear prescription glasses.
- Keep the spare key secure to prevent unauthorized use
- Remove the operating key and store securely when not in use
- Access to and viewing of the treatment area should be limited while the laser is in operation to those wearing suitable protective eye-wear.
- Always check the device and leads for damage before use and if unsure consult a qualified electrician.

6 Preparation

6.1 Receiving and Opening



Your machine will need to be serviced and the packaging will keep the machine safe during transport. The "Strawberry" tm "Strawberry" & Cream" model low level laser system is delivered in a cardboard box. A top tray contains the "paddles", "diodes", "Leads" and other accessories while the main unit is securely packed underneath.



Contents:

- 1 Strawberry or Strawberry & Cream model low level laser system unit.
- 1 power lead
- 10 paddles (9 standard and 1 end paddle)
- 2 probes
- 10 connecting leads
- 1 long lead for first paddle
- 1 longer lead to connect two sets of paddles
- 8 short connecting leads
- 2 Keys
- 2 Replacement fuses
- 1 pair of goggles
- 2 Small straps
- 2 Medium straps
- 1 Large strap
- 1 user manual

6.2 Assembling Your System

- 1. Unpack the contents of the cardboard box and check no items are missing.
- 2. Lay the Velcro straps on a table
- Lay out the laser paddles on the Velcro straps. Make sure the "end paddle" is the last in line; it
 has only one socket so is easy to identify.
- Connect the long lead between the front larger socket on the left hand side of the base of the unit and the first paddle
- 5. Connect the two sets of 5 paddles with the longer lead between paddles 5 and 6.
- 6. There is only one connection to the last paddle, paddle 10.



The two diodes connect with dual lead to one plug. This fits into the second, smaller sockets on the left hand side of the base of the unit.



- 8. Connect the power lead between the rear of the base of the main unit and a suitable supply.
- 9. Insert the key into the lock at the rear of the base.
- Give one final check that everything is securely connected and then if you are ready to power up the system for use, turn on the power switch at the rear of the base.

7 Understanding Your System

7.1 Laser Theory

The term laser is derived from: L ight A mplification by the S timulated E mission of R adiation

The Low level Laser model Strawberry and Strawberry and Cream utilizes a semi-conductor diode to produce a coherent light source at a wavelength of 660nm in the "paddles" which lay on the patient's skin during treatment.

7.2 How It Works

When the laser paddles are switched on, the laser light penetrates the skin and creates a reaction, see the diagram below:



The Laser Beams have an active penetration of 9mm but continue deeper up to 13mm but with little or no effect.

Adipocyte Fat Cells



Adipocyte Fat Cells After aser Treatment



Pares are formed on its adapt give name dimong form to apid out.





Adipocyte Fat Cells After Laser Treatment



More under (Blast) Free Patty Assis (green) and Oppered (possie) apit aut



Adjustita cella significanti notices in stea vivila lina Mente nassal stage conflictuat

13 Proposed Labeling

7.3 Training

Details to be finalized:

Free training will be available from Laser Lipo or its distributors in the United States, however details have yet to be confirmed.

A telephone support help line with a 1-800 number will also be provided.



7.5 Laser Machine Symbols and Specification

lymbot	Description:
-	Manufacturer date represented as YYYY-MM
¥	Type BF applied parts – provides degree of electrical protection against electric shock with isolated or floating applied parts
	Caution – there are specific warnings or precautions associated with the device.
	T Consult Operating Instructions

Technical Specifications

STRAWBERRY UNIT.

Power supply: Power consumption: mensions 315 mm Weight: mm LCD Display 100-240 V~ Max 60W Di-275 x 185 X 6 kilos X 315 20 x 4 LCD

Red Laser Probe 660nn 40mW Visible

Output tolerance of Laser Diodee	<±15% Stated Probe Label Output Power
Measurement of uncertainty	5%
Increase in measured quantities	0%. Diodes are the main source for potential increases in messured quantities over time. Laser diodes suffer negligible obvious degradation until they fail.
Wavelength	660nm ±15 nm
Power Output	40mW
NOHD	120mm
Beam Divergence	9 x 38 degrees Typ

GaAlAs

Lasing Medium

8 Instructions for Use

8.1 Strawberry Low Level Laser System

- 1. Set up the system as described in section 6.2
- 2. Switch on the unit by using key and power switch, see section 6.2
- The screen will show the default treatment time of 10 minutes. You may use the up and down arrows on the front panel to alter this if desired.
- 4. The display "MENU>" indicates the menu can be accessed by the right arrow on the front panel. Follow the on-screen instructions to adjust the sound setting and total time.
- 5. Two indicator lights show that the power is connected and the laser is ready for use.
- There is a "STOP" button in the middle of the front panel which will stop the unit when pressed at any time.

8.2 Strawberry and Cream Low Level Laser System

- 1. Set up the system as described in section 6.2
- 2. Switch on the unit by using key and power switch, see section 6.2
- 3. Adjust the treatment time using the touch screen, see below:

Main Menu	Sound	Support	Counters
INCH LOS	S	[16r	m 14s
SET TIM	-	Star	t 🔘
O Inch Los Diode	s Paddle	Paus	e 🔘

4. Adjust the alarm settings using the touch screen, see below:



9 Maintenance

9.1 Laser Servicing and Life

The machine should be serviced every 6 months or if only used infrequently maybe serviced annually

Like any light emitting device, the light output from each laser diode will decrease with time/use. It is important to service the equipment regularly so that compensating adjustments can be made.

Machines are set when new to achieve the specified output using approximately 60% of the maximum capacity; hence this utilization can be increased to compensate for the time-based reduction in light output. Machine life is thus anticipated to be between 3 and 5 years.

Laser Lipo will finalize service arrangements with their US distributors and include in the final manual.

9.2 Error Messages

"CHECK PADDLE" on Strawberry ILO model LCD Screen only:

- 1. Turn off machine
- 2. Disconnect from power supply
- 3. Check all leads and connections to laser paddles and probes
- 4. Check laser paddles are connected in correct order
- 5. Reconnect to power supply
- 6. Turn on machine

No light or messages on screen when unit is switched on:

- 1. check power lead is fully inserted at socket on rear of machine
- 2. check power lead is fully inserted and turned on at wall socket
- 3. check that the power switch is turned on, the "1" position
- if the above does not resolve the problem, turn off and disconnect from power supply then check the fuses, see section 9.5.

9.3 Cleaning

We advise that after each use the paddles should be cleaned with a "sterile wipe" and that the probes are sterilized.

9.4 Routine Maintenance

There are no consumables needed and no routine maintenance is required.

9.5 Changing the Fuses

9.5.1 Equipment Fuse



Once you have replaced the fuse (Spares are supplied with each new machine) Re-assemble the fuse holder and carefully push back into the machine. Be careful of the two copper pins, when putting it back together. Align the component prior to inserting it.



9.5.2 Power Supply Fuse

The power supply fuse is located between the power switch and the power lead socket at the rear of the base unit. Consult a qualified electrician if this fuse has blown.

13.4. Training Materials

Laser Lipo Ltd will develop appropriate training materials for the United States with their distributors based on developing the European materials included as drafts in this section.

Photography for Abdomen



You <u>MUST</u> photograph every client before the start of the course and after the final treatment in order to show your client their before and after images.

DO NOT PHOTOGRAPH CLIENT'S FACES.



You will need to take 4 pictures.

Back:

Side:



Always make sure that the client has their hands on their head.

You will take the after pictures in the same way, at the end of the course. Make sure the client is wearing the <u>SAME</u> underwear.

<u>10 Minute Treatment For</u> <u>Lower Abdomen</u>

You must measure **EVERY** client **BEFORE** and **AFTER** every treatment. Below are images on where you measure for the abdomen treatment.

You will measure 3 areas when treating the client's lower abdomen.

You need to measure the **Waist, Belly button & Hips** on the client's abdomen.









<u>10 Treatment for</u> <u>Abdomen</u>

Step 1:

Ensure the machine is connected correctly (refer to user manual). All paddles should be placed on the abdomen strap to cover the area to be treated. Place safety goggles onto the clients eyes and the technicians.



Step 2:

Place all of the 10 paddles on to the front of the clients abdomen. Ensure the timer is set to 10 minutes, and press start.



Running Machines



- 1) Use directly post treatment
- Use the running machine as instructed. For 30 minutes only under the supervision of the technician.
- For Cardio Vascular exercise you would programme the treadmill to the 'up-hill' setting.

Power Plate Machines

These machines may seem very simple but they can be very damaging to a person with existing injuries. It is **VITAL** that the client is made aware of any contra-indications prior to them using this type of device.

The power plate will help to speed up the body's metabolic rate and by doing so stimulate the body's natural lymphatic system. This in turn will stop the body from re-storing the fat that has been released from the laser treatment.

A 30 minute session on a standard setting will suffice.

13.5. Promotional Material

Promotional materials will be developed with US distribution partners based on the style of European materials under Labeling Control to ensure they do not exceed the scope of this 510[k].



13.6. Web-site and Other Electronic Media Materials

Laser Lipo Ltd will develop the domain name that it has purchased www.strawberry-laser.us. This will be a dedicated website to support the Strawberry and Strawberry and Cream low level laser systems in the United States, managed in partnership with its local distributors.

The website will be divided between information for potential users and potential patients and will include appropriate descriptions of the products, technologies, treatments and contact details.
Center for Devices and Radiological Health

January 31, 2013

14. Sterilization and Shelf Life

14.1. Sterilization

The Laser Lipo Ltd Strawberry and Strawberry & Cream is supplied nonsterile and is not designed to be sterilized.

14.2. Shelf Life

The Laser Lipo Ltd 'Strawberry' and 'Strawberry and Cream' low level laser systems use solid state diodes for the generation of laser light. All such diodes suffer degradation with time and use.

Having considered the length of use during treatment, number of treatments per day, the manufacturer's data on the laser diodes and Laser Lipo's own experience with products in the European Market, we conclude:

- if products are used intensively on a regular basis, best practice would be to service the unit every 6 months
 - o counters are included to help the user make this determination
 - it is anticipated that even intensively used devices will be within specification, even if they are only serviced annually, but their performance could be less than optimal
- however for the majority of products, where treatments are few, or carried out infrequently, an annual service is recommended.

These servicing requirements are included in the user manual shipped with every system. At each service, the output of the control unit is reset and the output from the laser diodes measured, to ensure performance is still capable, and within specification. Data can be reviewed in section 18.

All laser diodes have an operational life before they are required to be recalibrated or replaced, hence undertaking output measurements during a routine service by a qualified technician is the industry norm.

Laser Lipo Ltd aims to ensure the 'Strawberry' Laser systems shall be maintainable up to 5 years from the last manufacture, beyond this time availability of components may become difficult.

A shelf-life has been determined for 5 years based on the laser diodes.

Annunciation of failure alarms: if the system was to fail for any reason the system's alarm would sound

System Failure Probability – We have not had any system failure on either product.

14.2.1. Laser Diode Manufacturer's Specification

Environmental Variation Conditions – Environmental Requirements: 10°C to 30°C Non-condensing humidities below 75% RH

Output tolerance of Laser Diodes	<±15% Stated Probe Label Output Power
Measurement of uncertainty	5%
Increase in measured quantities	0%. Diodes are the main source for potential

0%. Diodes are the main source for potential increases in measured quantities over time. Laser diodes suffer negligible obvious degradation until they fail.

660nn 40mW Visible Laser Probe

660nm ±15 nm
40mW
120mm
9 x 20 degrees Typ
3B
GaAlAs

Laser Beam profile and diode accuracy – Spread of 17 degrees, diffused.

Center for Devices and Radiological Health

January 31, 2013

15. Biocompatibility

15 Biocompatibility

Page 15.1

15.1. Biocompatibility Summary

For the Laser Lipo Ltd 'Strawberry' and 'Strawberry' and Cream low level laser systems, an assessment of biocompatibility was made following the FDA recognized consensus standards ISO 14971 and ISO 10993-1 as amplified the FDA "Blue Book Memo #G95-1" and other FDA guidance.

The goggles are the same model from the same manufacturer as the predicate device the Chromogenex i-Lipo, K111501. The only difference is that they are printed on the outside surface with the "Strawberry-Laser" web address.



Figure 15-1 - Goggles showing arm printing

15.2. Assessment of Subject Contact

(Repeated from section 11.4.3)

Part Name	Details	Type of Subject Contact	Material
enclosure for control unit	see section 11.1.2	The clinician touches the display for a few seconds	thermoplastic elastomer
Strap	see section 11.1.15	The straps are pulled taught around the clients body and are kept on for the duration of the treatment time	Nylon
Paddle	see section 11.1.7	Paddles are placed against the selected fatty area, pressed against the skin and held in place with Velcro straps.	thermoplastic elastomer
Probe	see section 11.1.8	Probes are placed against the selected fatty area, pressed against the skin.	thermoplastic elastomer
Goggles	see section 11.1.16	goggles are kept on the clients face for the duration of the treatment	Plastic: Nylon Lenses: Polycarbonate
LCD Display	see section 11.4.2	The clinician touches the display for a few seconds	LCM Module Model AT070TN83 V.1 by InnoLux

(b)(4) Trade Secret Process - Product Specs

(b)(4) Trade Secret Process - Product Specs

15.4. ISO 10993-1 Biocompatibility Assessment

(b)(4) T	rade Secret Process -	BiocompatibilityTe	esting	
	15 Biocompatibility	,		Page 15.10

(b)(4) Trade Secret Process - BiocompatibilityTesting

Center for Devices and Radiological Health

January 31, 2013

(b)(4)	Trade Secret Process - Biocomp	batibility l esting		

(b)(4) Trade Secret Process - Biocompatibility Le	sting	

15 Biocompatibility

16. Software

SECTION 16 TABLE OF CONTENTS

16. Sof	tware		16.1
16.1.	Level	of Concern	16.2
16.1	.1.	Decision	16.2
16.1	.2.	Tabular Analysis of Level of Concern	16.3
16.2.	FDA S	Software in Submissions Requirements	16.6
16.2	.1.	Compliance with IEC and FDA requirements	16.6
16.2	.2.	Explanatory Note on Software Level of Concern	16.6
16.3.	IEC 6	52304 Software Development Report	16.7
16.3	.1.	Contents	16.8
16.3	.2.	Document Revision History	16.9
16.3	.3.	Introduction	16.10
16.3	.4.	Purpose	16.10
16.3	.5.	Scope	16.10
16.3	.6.	Software Classification	16.11
16.3	.7.	Requirements	16.12
16.3	.8.	Device Description	16.12
16.3	.9.	Development Plan	16.14
16.3	.10.	Software Performance and Functional Requirements .	16.17
16.3	.11.	Software Release	16.19
16.3	.12.	Software Modification and Maintenance Plan	16.19
16.3	.13.	Annex A: Software Listing for Strawberry	16.20
16.3	.14.	Annex B: Software Listing for Strawberry & Cream	16.21
16.4.	FDA /	Additional Requirements	16.23
16.4	.1.	Device Hazard Analysis	16.23
16.4	.2.	Software Requirements Specification	16.38
16.4	.3.	Architecture Design Chart	16.39
16.4	.4.	Software Design Specification	16.43
16.4	.5.	Verification and Validation Activities	16.50
16.4	.6.	Revision Level History	16.61
16.4	.7.	Unresolved Anomalies	16.61

(b)(4)	Trade Secret Process	- Software Specs		

(b)(4)	Trade Secret	Process -	Software Specs	

Section 16. Software

Center for Devices and Radiological Health

(b)(4) Trade Secret Process	- Software Specs		

16.2. FDA Software in Submissions Requirements

16.2.1. Compliance with IEC and FDA requirements

There are differences in approach and level of documentation required between IEC 62304 and FDA's "Software on Submissions" Guidance, and a different method of classification (determining level of concern, explained below in Section 16.2.2). Hence, Laser Lipo Limited include first the IEC 62304 Software Development Report (as reviewed by our Notified Body SGS for the European CE Mark) as Section 16.3 and then additional items identified by FDA in Section 16.4.

16.2.2. Explanatory Note on Software Level of Concern

Based on the determination that the software is a moderate level of concern, the FDA Guidance for "software in submissions" has been compared against the requirements of the FDA recognized consensus standard IEC 62304 "Medical Device Software, Software Life-cycle Processes" in section 9 and the IEC 62304 Software Design Report is presented below, see section 16.3.

It may be noted that while the FDA Guidance on determining the level of concern associated with software is hazard-based, the IEC 62304 Life-cycle Standard is risk-based. Following the FDA hazard-based approach we agree the software should be considered a moderate level of concern because there is a hazard posed software control of laser light, however taking the ISO 14971 risk-based approach of IEC 62304 we see the probability of this hazard being realized as exceptionally low and hence conclude the software level of concern as Class A.

We do not see the above as inconsistent but as demonstrating the difference in taking a risk or hazard based approach.

January 31, 2013

16.3. IEC 62304 Software Development Report

Strawberry Laser & Strawberry & Cream

Software Development Report

Section 16. Software
16.4.4. Software Design Specification

16.4.5. Verification and Validation Activities

b)(4) Trade Secret Process - Software Spec

Section 16. Software
17. Electromagnetic Compatibility and Electrical Safety

SECTION 17 TABLE OF CONTENTS

17. Electromac	netic Compatibility and Electrical Safety	
17.1 Elect	romagnetic Compatibility Test Report	
17.1.1	Strawberry low level laser system	
17.1.2	Strawberry & Cream low level laser system	
17.2 Elect	rical Safety Test Report	17.100
17.2.1	Strawberry low level laser system	17.100
17.2.2	Strawberry & Cream low level laser system	
17.2.3	Reference to Usability Report	17.481

17.1. Electromagnetic Compatibility Test Report

17.1.1. Strawberry low level laser system
17.1.2. Strawberry & Cream low level laser system
17.2. Electrical Safety Test Report

17.2.1. Strawberry low level laser system

17.2.2. Strawberry & Cream low level laser system

17.2.3. Reference to Usability Report

A Usability Report has been conducted to IEC 60601-1-6:2008 and can be seen in section 11.2.

18 Performance Testing - Bench

18 Performance Testing - Bench

כ כ)(4)) Trade	Secret	Process	- Perform	nance Testing



18 Performance Testing - Bench

Section 18 page 14

18 Performance Testing - Bench

18 Performance Testing - Bench
January 31, 2013

18 Performance Testing - Bench

18 Performance Testing - Bench

Section 18 page 21

18 Performance Testing - Bench

18 Performance Testing - Bench

January 31, 2013

19. Performance Testing – Animal

No animal testing has been conducted with the Laser Lipo 'Strawberry' and 'Strawberry' and Cream low level laser systems.

20. Performance Testing – Clinical

SECTION 20 TABLE OF CONTENTS

20.	Performance Testing – Clinical	20-1
20.1.	US Clinical Trial Data	20-2
20.2.	European Data	20-2
20.3.	Study objectives and parameters	20-3
20.3.1.	Gender selection:	20-4
20.3.2.	Placebos to be included:	20-4
20.3.3.	Age:	20-4
20.3.4.	Size or BMI:	20-4
20.3.5.	Success criteria:	20-5
20.4.	Treatment protocol	20-5
20.4.1.	Approved Strawberry System operators:	20-5
20.4.2.	Prior to treatments:	20-5
20.4.3.	Contraindications:	20-5
20.4.4.	The Consultation Process:	20-6
20.4.5.	Before and after treatment Photographic records	20-8
20.4.6.	Taking and recording subjects measurements	20-9
20.4.7.	Marking the placement of the tape measure on the skin	with pen.
		20-10
20.4.8.	Measurements taken using STRAWBERRY branded tape	measure.
		20-10
20.4.9.	Weighing the subjects	20-11
20.4.10.	Recording of treatment data	20-11
20.4.11.	Treatment frequency	20-12
20.5.	Clinical study conducted by Caldy Clinic.	20-13

(b)(4) Trade Secret Process - Performance Testing								

January 31, 2013

b)(4) Trade Secret Process - Performance Testing

b)(4) Trade Secret Process - Performance Testing

cret Process - Perform

(b)(1)-T

$(\sim)(1)$		anee reeting		

Center for Devices and Radiological Health (4) Trade Secret Process - Performance Testing

January 31, 2013

b)(4) Trade Secret Process - Performance Testing

Center for Devices and Radiological Health (4) Trade Secret Process - Performance Testing

b)(4) Trade Secret Process - Performance Testing

January 31, 2013

(4) Trade Secret Process - Performance Testing

ecret Pro cess - Performance Testing

January 31, 2013

(b)(4) Trade Secret Process - Performance Testing

. .


(b)(4) Trade Secret Process - Performance Testing

Center for Devices and Radiological Health b)(4) Trade Secret Process - Performance Testing

b)(4) Trade Secret Process - Performance Testing

January 31, 2013

(b)(4) Trade Secret Process - Performance Testing

January 31, 2013

(b)(4) Trade Secret	Process - Performanc	eTesting		

January 31, 2013

b)(4) Trade Secret Process - Performance Testing

(b)(4) ⁻	Trade Secret	Process -	Performance	Testing			

)(4) Trade Secret Process - Performance Testing

b)(4) Trade Secret Process - Performance Testing

January 31, 2013

(b)(4) Trade Secret Process	s - Performance Testing		

b)(4) Trade Secret Process - Performance Testing

January 31, 2013

(b)(4) I rade	e Secret Process	- Performance	lesting		

January 31, 2013

21. Other

Laser Lipo Ltd do not wish to present any further information at this time and feel they have determined substantial equivalence of the 'Strawberry' and 'Strawberry and Cream' low level laser systems to the Chromogenex i-Lipo system K111501 based on the information in the preceding sections.

Barlow, Lenny *

From: Sent: To: Cc: Subject: Attachments:

ŕ

Barlow, Lenny * Friday, September 13, 2013 2:04 PM 'ian@strawberry-laser.com' DCCLetters k130341 Correspondence k130341.pdf

S



From: · Rev	iewer Name	Sankar Basu		
Subject: 510)(k) Number	<u>K130341</u>		
To: The	Record			
Please list CTS deci	sion code: <u>SE - Su</u>	ubstantially Equivalent		
Refused to Acc	ept (Note: this is consi	dered the first review cycle. See <u>screening checklist</u> .)		
Hold (Addition	al Information or Telep	phone Hold)		
Final Decision (SE, SE with Limitation	s, NSE (select code below), Withdrawn, etc.)		
Please complete the	following for a final cl	earance decision (i.e, SE, SE with Limitations, etc.)	YES	NO
Indications for Use P	age (Attach IFU)		X	
510(k) Summary or 5	10(k) Statement (Attac	ch Summary or Statement)	×	
Truthful and Accurat	e Statement (Must be p	present for a Final Decision)	X	
¹ " the device Class III	?			X
firm reference	standards? (If yes, plea	se attach <u>Form 3654</u> .)	×	
Is this a combination	product?			×
Is this a reprocessed for Reprocessed Sing	single use device? (See ite-Use Medical Device	e <u>Guidance for Industry and FDA Staff - MDUFMA - Validation Data in 510(k)s</u> is.)		×.
Is this device intende	d for pediatric use on	ky?		\times
Is this a prescription	device? (If both prescr	iption & OTC, check both boxes.)	×	
Is clinical data necessary to support the review of this 510(k)?				
For United States base Requirements of Clin included or was inco	sed clinical studies onl icalTrials.gov Data Bar mplete, then applican	y, did the application include a completed Form FDA 3674, Certification with hk? (If study was conducted in the United States and Form FDA 3674 was not t must be contacted to obtain completed form.)		
Does this device incl	ude an Animal Tissue S	Source?		×
All Pediatric Patients	age <= 21			×
Neonate/Newborn (I	Birth to 28 days)	· · · · ·		×
Infant (29 days to < 2	years)			×
Child (2 years to <12	years)			X
Adolescent (12 years	to <18 years)			X
Transitional Adolesce 's age >= 21 (dif	ent A (18 years to <21 ferent device design o	years); Special considerations are being given to this group, different from r tesating, different protocol procedures, etc.)		×
sitional Adolesce	ent B (18 years to <21 y	years); No special considerations compared to adults >= 21 years)		×

Nanotechnology	(×
his device subject to the Tracking Regulation? (Medical Device Tracking Guidance)		×

Regulation Number:	21CFR 878.4810				
Class:	II				
Product Code:	OLI				
Additional Product Codes:					

Digital Signature Concurrence Table (Not all signatures may be required)					
Branch Chief Sign-Off	Neil R Ogden 2013.09.06 11:31:01 -04'00'				
Division Sign-Off	Mark N. Melkerson -S 2013.09.0615:59:59-04'00'				

Dr. Sankar Basu, PHD U.S. Food and Drug Administration Center for Devices and Radiological Heath Document Control Center - WO66-0609 10903 New Hampshire Avenue Silver Spring, MD 20993-0002



Ref. K130341.

FDA CDRH DMC

MAR 07 2013

Received

March 01, 2013

Dear Dr. Basu,

In reference to your email dated February 28, 2013 I can confirm that we have addressed the following elements that were identified as missing or inconsistent within the Acceptance checklist for Traditional 510(k)s.

- A. Administrative
 - Section 7 a

Form 3454 Certification: Financial Interests and arrangements of clinical investigators has been completed.

Section 7 b Form 3674 Certification of compliance has been completed.

Section 9

The submission identifies prior submissions for the same device details of this can be found within section 3 of our 510(k) Cover letter, and also in section 2 of our 510(k) application CDRH Premarket Review Submission Cover Sheet within section (F)

Section 9 a

We have identified in Appendix A where we identified previous submission issues and how these where resolved.

Please do let me know if I can be of further assistance.

Yours Sincerely

lan Cobley Director.

... continuing to lead the market with innovation

Laser Lipo Ltd. "Heath House" Crockham Hill Edenbridge Kent. TN8 6ST. United Kingdom. T. 0044 844 980 1820. W. strawberry-laser.com Co. Reg. No. 06308992. VAT Reg. No. 891 7315 01

3. 510(k) Cover Letter

Laser Lipo Ltd Heath House Crockham Hill Edenbridge United Kingdom, TN8 6ST Telephone: +44 844 980 1820 e-mail: ian@strawberry-laser.com



March 05, 2013.

Food and Drug Administration Center for Devices and Radiological Health Document Mail Center - WO66-G609 10903 New Hampshire Avenue Silver Spring Maryland 20993-0002 United States of America

Ref: 510(k) Premarket Notification

Dear Document Control Clerk:

Pursuant to the requirements of section 510 (k) of the Food, Drug and Cosmetic Act, notification is made to manufacture and market the following new medical device:

Classification Name:

Low Level laser system for aesthetic use 79 OLI

Classification Regulation:

21 CFR 878.5400

Common/Usual Name:

Low level laser system for aesthetic use

Page 3.2

Proprietary Name:

Laser Lipo Ltd will manufacture two devices:

- the Strawberry low level laser system model ILO, and,
- the Strawberry & Cream low level laser system model SC

Establishment Registration Number:

The Strawberry low level laser system and Strawberry & Cream low level laser system will be manufactured by:

Laser Lipo Ltd Heath House Crockham Hill Edenbridge Kent TN8 6ST United Kingdom Telephone: 011 44 844 980 1820 Fax: 011 44 844 980 1820 Establishment Registration Number: To be applied for following clearance of this submission.

Classification:

The FDA has classified: Low level laser system for aesthetic use (79 OLI) as a Class II medical device under 21 CFR 878.5400.

Performance Standards:

None established under 514, however FDA has published "Class II Special Controls Guidance Document: Low Level Laser System for Aesthetic Use".

The following international consensus standards, recognized by FDA, have been complied with:

IEC 62304:2006 – software life-cycle IEC 60601-1:2006 – electrical safety IEC 60601-1-2:2007 - electromagnetic compatibility IEC 60601-2-22:1996 – laser products IEC 60825-1:2007 – laser products ISO 14971:2009 – risk management ISO 10993-1:2009 - biocompatibility The following additional standards have also been complied with, although not recognized by the FDA:

ISO 13485: 2003 – quality management system EN 980:2008 – European symbols (explanatory text included in US labeling) EN 1041:2008 – requirements for information (supplemented by FDA guidance listed below)

The following Guidance notes have been used extensively in the preparation of this submission:

- Class II Special Controls Guidance Document: Low Level
 Laser System for Aesthetic Use
- Guidance for the Content of Premarket Submissions for Software Contained in Medical Devices
- Format for Traditional and Abbreviated 510(k)s
- Laser Products Conformance with IEC 60825-1 and IEC 60601-2-22

Labeling and Promotional Materials:

Draft labels, labeling (including the User Manual) and promotional materials for the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are contained in Section 13.

Substantial Equivalence:

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is substantially equivalent in design, use and materials to the:

Chromogenex Technologies Ltd I-Lipo System – K111501

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is made of the similar materials as the Chromogenex Technologies Ltd I-Lipo System – K111501.

The Laser Lipo Ltd Strawberry and Strawberry & Cream low level laser systems has similar indications for use

The Low level Laser model Strawberry and Strawberry and Cream can be used for the non-invasive aesthetic treatment for the temporary reduction in waist circumference.

to the Chromogenex Technologies Ltd I-Lipo System.

The Chromogenex Technologies Limited i-lipoTm Low Level Laser System is indicated for non-invasive aesthetic treatment for the temporary reduction in circumference of the waist.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is made of the same materials as the Chromogenex Technologies Ltd I-Lipo System – K111501.

- thermoplastic control unit enclosure
 - o proprietary components and assemblies used within the unit
- thermoplastic paddles and probes
- nylon covered "velcro" fixed straps
- nylon/polycarbonate goggles

Both the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd i-Lipo System:

- use laser diodes
 - o of approximately 660 nm wavelength, and
 - o less than 50 mW output
- incorporated into
 - various laser diode cluster probes and paddles attached to the subject with special straps
 Strawberry 4 to 10 paddles with 6 diodes each
 i-Lipo 4 paddles with 9 diodes each
 - o and two probes with one diode each
- have a similar control unit
 - LCD subject display
 - o membrane key pad inputs
 - incorporated into display as a touch screen for strawberry and cream
 - o microprocessor controlled through software
 - o multi-range power supply (EU and US voltages and frequencies)
- have a similar range of connecting cables
- have the same two pairs of laser protection goggles
 - o different name printed on outside of frame

The Laser Lipo Ltd Strawberry low level laser system, Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd I-Lipo System have the same use achieved through the same technology, are assembled from similar components made from similar materials, have similar user input and output interfaces and have similar features.

All systems use laser emitting diodes of the similar wavelengths and powers: the Laser Lipo systems have the potential to use more paddles but with less

diodes in each paddle. All systems have two cluster probes with a single laser diode in each. In all systems the paddles and cluster probes are secured to the subject with nylon encased rubber straps secured with "velcro".

All control units have plastic enclosures, discrete power supply, microprocessor controlled electronics, controls, alarms and similar control software.

Laser Lipo Ltd concluded that the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are substantially equivalent to the Chromogenex Technologies Ltd I-Lipo System, K111501

For a demonstration of equivalence, please see Section 12.

Promotional Material giving details of the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and Chromogenex Technologies Ltd I-Lipo System is compared in Section 12.

Previous Submission:

Laser Lipo Limited has previously submitted the Strawberry and Strawberry & Cream Low Level Laser system under K122354 and PRE Submission Q120359. The reasons that the Strawberry and Strawberry & Cream devices were not considered being substantially equivalent to the predicate device can be found in Appendix A - Administrative section with a summary of the corrective action taken.

Description of Product:

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system consists of a control unit, various connection leads up to 10 paddles, 2 cluster probes, various "Velcro" attachment straps and other accessories. (A full list of system contents is included at the end of this section.)

A paddle is a device containing six cold red laser emitting diodes which is designed to be placed on the skin. The system can operate using 4, 6, 8 or 10 paddles that are connected to the control unit.

A probe is a device containing one cold red laser emitting diode which is designed to be placed on the skin to treat specific smaller areas of fat, where a usual flat paddle won't ergonomically fit.

The control unit is an electrically powered unit (100-240v, 50-60Hz autoranging), enabled by a main switch and key switch. Once enabled it is

controlled using a button (Strawberry) or touch screen (Strawberry and Cream) interface. The output of the diode (six per paddle) is limited to 40mW +/- 15% by a power limiting PCB from the central processing unit.

When the laser paddles are placed on the skin, the cold red laser beams penetrate the skin just deep enough to reach the layers of fat. When the light hits the fat cells, a rapid chain of events takes place. Firstly, pores form on the cells causing them to spill out. The water, Glycerol and fatty acids move into the interstitial space beneath the fatty layer in the skin. Then further water, fatty acids and Glycerol spill out. The adipocyte cells are therefore reduced in size.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system contain the same control electronics and control software however the Strawberry and Cream has a menu driven touch screen interface while the Strawberry model has a liquid crystal display and (up, down, left, right and center/enter) push buttons. The outputs and functionality of the Strawberry and Strawberry and Cream models are identical, however displays, display drivers, interface electronics and software and power supply transformed are different. Full details are contained in Section 11.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is shipped in dedicated packaging to prevent damage and protected by a cardboard box.

The system as shipped contains:

- o 1 low level laser system unit, either
 - Strawberry model, or
 - Strawberry & Cream model
- o 1 user manual
- o 1 power lead
- o Paddle leads, for example for a 10 paddle system
 - 1 long paddle lead for connection to the device
 - 8 paddle standard leads
 - 1 long paddle lead
- o 2 cluster Probes
- o 1 Pair of goggles
- o 2 Replacement fuses
- o 2 Keys

variance of the system w	m be available with 4, 0, 0 and 10 paddles.
number of paddles	number of paddle leads
3 standard and 1 end	3 standard
5 standard and 1 end	5 standard
7 standard and 1 end	7 standard
9 standard and 1 end	8 standard and 1 long (to connect two groups of five paddles)
M . 1 1 4 57 5	

Variants of the system will be available with 4, 6, 8 and 10 paddles.

Table 1 - Variants

A full list of the proposed variants and a schematic drawing showing the key design features are included in Section 11.

Engineering Drawings and Photographs of the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system can be found in Section 11.

Photographs, photo drawings and block diagrams showing the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd I-Lipo System are contained in Section 11 and comparison to the predicate Chromogenex Technologies Ltd I-Lipo System can be seen in Section 12.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system will be packaged in a cardboard box together with all components (instructions booklet, cables, paddles), software and accessories (diodes, goggles, fuse, straps and lead) within the scope of the 510(k). A shelf-life has been determined for 5 years based on the laser diodes.

Electrical Safety and Electromagnetic Compatibility is demonstrated through compliance with IEC 60601 part 1, part 1-2 and part 2-22 full test reports from a third party laboratory are included in Section 17.

The Guidance also states that there should be compliance with the IEC 60601-1-4 standard. Laser Lipo Ltd believes that compliance with IEC 62304:2006 life-cycle standard achieves at least the same level of assurance in the quality of the software used in the device.

A usability file detailing application of IEC 60601-1-6:2008 is contained in Section 11.

Table 2 - Product Release Testing Summary

Gives the results of bench testing of the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system which can be seen in section 18 and is confirmed by clinical reports in section 20.

Test	Mean	Quantity	Sample Standard Deviation	Capability, Cp
Laser diode power output: Paddle Laser diode output is set to 40mw +/-15%.	39.35	1140	0.81	2.47
Probe Laser diode output is set to $40 \text{mw} + -15\%$.	40.02	40	0.72	3.49
EU portable appliance testing	100% pass	20		
 Visual Inspection: external components, cables, outer casings and the General condition of all associated parts. 	100% pass	20		100% inspection
Assembly tests	100% pass	20		100% inspection
Functionality tests	100% pass	20		100% inspection

Table 2 - Product Release Testing Summary

Biocompatibility:

Biocompatibility assessment on the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system have been conducted following the ISO 10993-1 (FDA Recognized Consensus Standard), Biological Evaluation of Medical Devices, Part 1: Evaluation and Testing. Please see section 15.

Sterilization:

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are supplied non-sterile.

Cleaning instructions are contained in the user manual included with each system.

Safety and Effectiveness:

A summary of safety and effectiveness information ("510(k) Summary" per 21 CFR 807.92) upon which an equivalence determination has been made is included as Section 5.

Indications for Use Statement:

A single sheet statement of "indications for use" as required by FDA for submissions after January 01, 1996, is included in Section 4.

Truthful and Accurate Statement:

The premarket notification "truthful and accurate" statement (per 21 CFR 807.87(k)), signed by the Group Director Ian Cobley of Laser Lipo Ltd is included as Section 6.

E Copy Statement:

The E Copy provided within our submission is an exact duplicate of the paper copy.

Owner:

This Pre-Market Notification is owned by Laser Lipo Ltd.

Submitter:

This Pre-Market Notification is submitted by Ian Cobley Group Director For and on behalf of: Laser Lipo Ltd

Contact Persons:

The address and telephone number for **all** correspondence is:

Ian Cobley Laser Lipo Ltd Heath House Crockham Hill Edenbridge Kent TN8 6ST United Kingdom Tel: 011 44 844 980 1820 Mobile: 011 44 777 445 9611 Fax: 011 44 844 980 1820 e-mail: ian@strawberry-laser.com

As per your email dated March 4, 2013 please find a copy of our response to the acceptance review notification, for K130341

If I can be of any further assistance, please do not hesitate to contact me.

Sincerely,

. . . . (-

Ian Cobley Group Director Laser Lipo Ltd

ally Cover Letter

Page 3.11



FDA CDRH DMC MAR 4 2013

r. Sankar Basu, PHD J.S. Food and Drug Administration Center for Devices and Radiological Heath Document Control Center - WO66-0609 10903 New Hampshire Avenue Silver Spring, MD 20993-0002

March 01, 2013

Dear Dr. Basu,

In reference to your email dated February 28, 2013 I can confirm that we have addressed the following elements that were identified as missing or inconsistent within the Acceptance checklist for Traditional 510(k)s.

Received

A. Administrative

Section 7 a

Form 3454 Certification: Financial Interests and arrangements of clinical investigators has been completed.

Section 7 b

Form 3674 Certification of compliance has been completed.

Section 9

The submission identifies prior submissions for the same device details of this can be found within section 3 of our 510(k) Cover letter, and also in section 2 of our 510(k) application CDRH Premarket Review Submission Cover Sheet within section (F)

Section 9 a

We have identified in Appendix A where we identified previous submission issues and how these where resolved.

Please do let me know if I can be of further assistance.

Yours Sincerely

Ian Cobley Director.

... continuing to lead the market with innovation

aser Lipo Ltd. "Heath House" Crockham Hill Edenbridge Kent. TN8 6ST. United Kingdom. T. 0044 844 980 1820. W. strawberry-laser.com Co. Reg. No. 06308992. VAT Reg. No. 891 7315 01

1000 - 1000 - 1000 - 1000 1000 - 1000 - 1000 - 1000 1000 - 1000 - 1000 1000 - 1000 - 1000 1000 - 1000 - 1000

en selfono en 2003. Al balla 200-maria en 200 maria (Contra en 2004). En 1910 - Al Reign (Entra en 1915), en 1910 - Entra en 1

Stem Physical difference in the



Dr. Sankar Basu, PHD U.S. Food and Drug Administration Center for Devices and Radiological Heath Document Control Center - WO66-0609 10903 New Hampshire Avenue Silver Spring, MD 20993-0002

March 01, 2013

Dear Dr. Basu,

In reference to your email dated February 28, 2013 I can confirm that we have addressed the following elements that were identified as missing or inconsistent within the Acceptance checklist for Traditional 510(k)s.

A. Administrative

Section 7 a

Form 3454 Certification: Financial Interests and arrangements of clinical investigators has been completed.

Section 7 b Form 3674 Certification of compliance has been completed.

Section 9

The submission identifies prior submissions for the same device details of this can be found within section 3 of our 510(k) Cover letter, and also in section 2 of our 510(k) application CDRH Premarket Review Submission Cover Sheet within section (F)

Section 9 a We have identified in Appendix A where we identified previous submission issues and how these where resolved.

Please do let me know if I can be of further assistance.

Yours Sincerely

lan Cobley Director.

... continuing to lead the market with innovation.

Laser Lipo Ltd. "Heath House" Crockham Hill Edenbridge Kent. TN8 6ST. United Kingdom. T. 0044 844 980 1820. W. strawberry-laser.com Co. Reg. No. 06308992. VAT Reg. No. 891 7315 01

February 28, 2013

3. 510(k) Cover Letter

February 28, 2013

Laser Lipo Ltd Heath House Crockham Hill Edenbridge United Kingdom, TN8 6ST Telephone: +44 844 980 1820 e-mail: ian@strawberry-laser.com



February 28, 2013.

Food and Drug Administration Center for Devices and Radiological Health Document Mail Center - WO66-G609 10903 New Hampshire Avenue Silver Spring Maryland 20993-0002 United States of America

Ref: 510(k) Premarket Notification

Dear Document Control Clerk:

Pursuant to the requirements of section 510 (k) of the Food, Drug and Cosmetic Act, notification is made to manufacture and market the following new medical device:

Classification Name:

Low Level laser system for aesthetic use 79 OLI

Classification Regulation:

21 CFR 878.5400

Common/Usual Name:

Low level laser system for aesthetic use

3 510(k) Cover Letter
Proprietary Name:

- Laser Lipo Ltd will manufacture two devices:
- the Strawberry low level laser system model ILO, and,
- the Strawberry & Cream low level laser system model SC

Establishment Registration Number:

The Strawberry low level laser system and Strawberry & Cream low level laser system will be manufactured by:

Laser Lipo Ltd Heath House Crockham Hill Edenbridge Kent TN8 6ST United Kingdom Telephone: 011 44 844 980 1820 Fax: 011 44 844 980 1820 Establishment Registration Number: To be applied for following clearance of this submission.

Classification:

The FDA has classified: Low level laser system for aesthetic use (79 OLI) as a Class II medical device under 21 CFR 878.5400.

Performance Standards:

None established under 514, however FDA has published "Class II Special Controls Guidance Document: Low Level Laser System for Aesthetic Use".

The following international consensus standards, recognized by FDA, have been complied with:

IEC 62304:2006 – software life-cycle IEC 60601-1:2006 – electrical safety IEC 60601-1-2:2007 - electromagnetic compatibility IEC 60601-2-22:1996 – laser products IEC 60825-1:2007 – laser products ISO 14971:2009 – risk management ISO 10993-1:2009 - biocompatibility

The following additional standards have also been complied with, although not recognized by the FDA:

ISO 13485: 2003 – quality management system EN 980:2008 – European symbols (explanatory text included in US labeling) EN 1041:2008 – requirements for information (supplemented by FDA guidance listed below)

The following Guidance notes have been used extensively in the preparation of this submission:

- Class II Special Controls Guidance Document: Low Level Laser System for Aesthetic Use
- Guidance for the Content of Premarket Submissions for Software Contained in Medical Devices
- Format for Traditional and Abbreviated 510(k)s
- Laser Products Conformance with IEC 60825-1 and IEC 60601-2-22

Labeling and Promotional Materials:

Draft labels, labeling (including the User Manual) and promotional materials for the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are contained in Section 13.

Substantial Equivalence:

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is substantially equivalent in design, use and materials to the:

Chromogenex Technologies Ltd I-Lipo System – K111501

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is made of the similar materials as the Chromogenex Technologies Ltd I-Lipo System – K111501.

The Laser Lipo Ltd Strawberry and Strawberry & Cream low level laser systems has similar indications for use

The Low level Laser model Strawberry and Strawberry and Cream can be used for the non-invasive aesthetic treatment for the temporary reduction in waist circumference.

to the Chromogenex Technologies Ltd I-Lipo System.

3 510(k) Cover Letter

The Chromogenex Technologies Limited i-lipoTm Low Level Laser System is indicated for non-invasive aesthetic treatment for the temporary reduction in circumference of the waist.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is made of the same materials as the Chromogenex Technologies Ltd I-Lipo System – K111501.

- thermoplastic control unit enclosure
 - proprietary components and assemblies used within the unit
- thermoplastic paddles and probes
- nylon covered "velcro" fixed straps
- nylon/polycarbonate goggles

Both the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd i-Lipo System:

- use laser diodes
 - o of approximately 660 nm wavelength, and
 - \circ less than 50 mW output
- incorporated into
 - various laser diode cluster probes and paddles attached to the subject with special straps
 Strawberry 4 to 10 paddles with 6 diodes each
 i-Lipo 4 paddles with 9 diodes each
 - and two probes with one diode each
- have a similar control unit
 - LCD subject display
 - membrane key pad inputs
 - incorporated into display as a touch screen for strawberry and cream
 - o microprocessor controlled through software
 - multi-range power supply (EU and US voltages and frequencies)
- have a similar range of connecting cables
- have the same two pairs of laser protection goggles
 - o different name printed on outside of frame

The Laser Lipo Ltd Strawberry low level laser system, Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd I-Lipo System have the same use achieved through the same technology, are assembled from similar components made from similar materials, have similar user input and output interfaces and have similar features.

All systems use laser emitting diodes of the similar wavelengths and powers: the Laser Lipo systems have the potential to use more paddles but with less

diodes in each paddle. All systems have two cluster probes with a single laser diode in each. In all systems the paddles and cluster probes are secured to the subject with nylon encased rubber straps secured with "velcro".

All control units have plastic enclosures, discrete power supply, microprocessor controlled electronics, controls, alarms and similar control software.

Laser Lipo Ltd concluded that the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are substantially equivalent to the Chromogenex Technologies Ltd I-Lipo System, K111501

For a demonstration of equivalence, please see Section 12.

Promotional Material giving details of the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and Chromogenex Technologies Ltd I-Lipo System is compared in Section 12.

Previous Submission:

Laser Lipo Limited have previously submitted the Strawberry and Strawberry & Cream Low Level Laser system under K122354 and PRE Submission Q120359. The reasons that the Strawberry and Strawberry & Cream devices were not considered to be substantially equivalent to the predicate device these can be found in Appendix A - Administrative section with a brief summary of the corrective action taken.

Description of Product:

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system consists of a control unit, various connection leads up to 10 paddles, 2 cluster probes, various "Velcro" attachment straps and other accessories. (A full list of system contents is included at the end of this section.)

A paddle is a device containing six cold red laser emitting diodes which is designed to be placed on the skin. The system can operate using 4, 6, 8 or 10 paddles that are connected to the control unit.

A probe is a device containing one cold red laser emitting diode which is designed to be placed on the skin to treat specific smaller areas of fat, where a usual flat paddle won't ergonomically fit.

The control unit is an electrically powered unit (100-240v, 50-60Hz autoranging), enabled by a main switch and key switch. Once enabled it is

controlled using a button (Strawberry) or touch screen (Strawberry and Cream) interface. The output of the diode (six per paddle) is limited to 40mW +/- 15% by a power limiting PCB from the central processing unit.

When the laser paddles are placed on the skin, the cold red laser beams penetrate the skin just deep enough to reach the layers of fat. When the light hits the fat cells, a rapid chain of events takes place. Firstly, pores form on the cells causing them to spill out. The water, Glycerol and fatty acids move into the interstitial space beneath the fatty layer in the skin. Then further water, fatty acids and Glycerol spill out. The adipocyte cells are therefore reduced in size.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system contain the same control electronics and control software however the Strawberry and Cream has a menu driven touch screen interface while the Strawberry model has a liquid crystal display and (up, down, left, right and center/enter) push buttons. The outputs and functionality of the Strawberry and Strawberry and Cream models are identical, however displays, display drivers, interface electronics and software and power supply transformed are different. Full details are contained in Section 11.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system is shipped in dedicated packaging to prevent damage and protected by a cardboard box.

The system as shipped contains:

- o 1 low level laser system unit, either
 - Strawberry model, or
 - Strawberry & Cream model
- o 1 user manual
- o 1 power lead
- Paddle leads, for example for a 10 paddle system
 - 1 long paddle lead for connection to the device
 - 8 paddle standard leads
 - 1 long paddle lead
- 2 cluster Probes
- 1 Pair of goggles
- 2 Replacement fuses
- o 2 Keys

number of paddles	number of paddle leads
3 standard and 1 end	3 standard
5 standard and 1 end	5 standard
7 standard and 1 end	7 standard
9 standard and 1 end	8 standard and 1 long (to connect two groups of five paddles)
Table 1 - Variants	

Variants of the system will be available with 4, 6, 8 and 10 paddles.

A full list of the proposed variants and a schematic drawing showing the key design features are included in Section 11.

Engineering Drawings and Photographs of the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system can be found in Section 11.

Photographs, photo drawings and block diagrams showing the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system and the Chromogenex Technologies Ltd I-Lipo System are contained in Section 11 and comparison to the predicate Chromogenex Technologies Ltd I-Lipo System can be seen in Section 12.

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system will be packaged in a cardboard box together with all components (instructions booklet, cables, paddles), software and accessories (diodes, goggles, fuse, straps and lead) within the scope of the 510(k). A shelf-life has been determined for 5 years based on the laser diodes.

Electrical Safety and Electromagnetic Compatibility is demonstrated through compliance with IEC 60601 part 1, part 1-2 and part 2-22 full test reports from a third party laboratory are included in Section 17.

The Guidance also states that there should be compliance with the IEC 60601-1-4 standard. Laser Lipo Ltd believes that compliance with IEC 62304:2006 life-cycle standard achieves at least the same level of assurance in the quality of the software used in the device.

A usability file detailing application of IEC 60601-1-6:2008 is contained in Section 11.

Table 2 - Product Release Testing Summary Gives the results of bench testing of the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system which can be seen in section 18 and is confirmed by clinical reports in section 20.

Test	Mean	Quantity	Sample Standard Deviation	Capability, Cp
Laser diode power output: Paddle Laser diode output is set to 40mw +/-15%.	39.35	1140	0.81	2.47
Probe Laser diode output is set to 40mw +/-15%.	40.02	40	0.72	3.49
EU portable appliance testing	100% pass	20		
 Visual Inspection: external components, cables, outer casings and the General condition of all associated parts. 	100% pass	20		100% inspection
Assembly tests	100% pass	20		100% inspection
Functionality tests	100% pass	20		100% inspection

Table 2 - Product Release Testing Summary

Biocompatibility:

Biocompatibility assessment on the Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system have been conducted following the ISO 10993-1 (FDA Recognized Consensus Standard), Biological Evaluation of Medical Devices, Part 1: Evaluation and Testing. Please see section 15.

Sterilization:

The Laser Lipo Ltd Strawberry low level laser system and Strawberry & Cream low level laser system are supplied non-sterile.

Cleaning instructions are contained in the user manual included with each system.

Safety and Effectiveness:

A summary of safety and effectiveness information ("510(k) Summary" per 21 CFR 807.92) upon which an equivalence determination has been made is included as Section 5.

Indications for Use Statement:

A single sheet statement of "indications for use" as required by FDA for submissions after January 01, 1996, is included in Section 4.

Truthful and Accurate Statement:

3 510(k) Cover Letter

The premarket notification "truthful and accurate" statement (per 21 CFR 807.87(k)), signed by the Group Director Ian Cobley of Laser Lipo Ltd is included as Section 6.

E Copy Statement:

The E Copy provided within our submission is an exact duplicate of the paper copy.

Owner:

This Pre-Market Notification is owned by Laser Lipo Ltd.

Submitter:

This Pre-Market Notification is submitted by Ian Cobley Group Director For and on behalf of: Laser Lipo Ltd

Contact Persons:

The address and telephone number for **<u>all</u>** correspondence is:

Ian Cobley Laser Lipo Ltd Heath House Crockham Hill Edenbridge Kent TN8 6ST United Kingdom Tel: 011 44 844 980 1820 Mobile: 011 44 777 445 9611 Fax: 011 44 844 980 1820 e-mail: ian@strawberry-laser.com

As per your letter dated February 14, 2013 please find a replacement E copy of our submission.

If I can be of any further assistance, please do not hesitate to contact me.

Sincerely,

Ian Cobley Group Director Laser Lipo Ltd February 28, 2013

Appendix A

A. Administrative section

- 9) Laser Lipo Ltd identified our prior submission K122354 within our new application K130341 within 'Section 2 CDRH Pre Market Review Coversheet'.
 - a) The reasons that the Strawberry and Strawberry & Cream devices were not considered to be substantially equivalent to the predicate device are listed below with the action taken to address these issues.
 - The original application contained data in two studies using two lymph probes for causing lymphatic drainage. The removal of these lymphatic probes required new study data to be supplied.

A new study is contained within Section 20 Performance Testing -Clinical within our new submission K130341.

ii) The previous submission did not include a cardiovascular exercise program post-laser treatment.

This can be found Within the treatment protocol and the clinical studies in Section 20 Performance Testing - Clinical in our new submission K130341.

iii) The Indication for Use (IFU) statement submitted originally included generic fat reduction and body contouring without specifying the anatomical area.

The new submission clarifies that the IFU is for the temporary reduction in waist circumference. This is listed within K130341 in Section 4, Indications for use statement.

ix) The labeling did not make it clear as to the number of laser paddles (containing 6 laser diodes) should be used on each patient.

Direct comparisons between the Strawberry laser paddles and the predicate laser paddles, including the number of paddles to be used and the amount of energy provided is contained within Section 11 Device Description of K130341

2. CDRH Premarket Review Submission Cover Sheet

		TMENT OF HEALTH AN FOOD AND DRUG AD	ND HUMAN SERVI MINISTRATION	CES Form Approval OMB No. 9010 Expiration Date See OMB Stat			val 10-0120 ate: December 31, 2013. tatement on page 5	
Date of Submission		User Fee Payment ID	Number	COVER	FD	A Submiss	sion Document	t Number (if known)
17 December 2	012	(b)(4) Trade Secr	et Process					
SECTION A			TYPE OF S	UBMISSIO	N			
PMA	PMA	& HDE Supplement	PDP		_	510(k)		Meeting
Original Submission		gular (180 day)	Original PDP		🛛 🖂 Origir	nal Submis	sion:	Pre-510(K) Meeting
Premarket Report		ecial	Notice of Cor	npletion		raditional		Pre-IDE Meeting
Modular Submission	Pa	nel Track (PMA Only)		to PDP		pecial		Pre-PMA Meeting
		day Supplement				DDreviated	(Complete	Pre-PDP Meeting
Report Amondmont		day Nouce			Additi	onal Inforn	nation	Agreement Meeting
		al time Review				Party	i du on	Determination Meeting
		endment to PMA						Other (specify)
	&H	DE Supplement						
	Oth	ner						
IDE	Hui	manitarian Device	Class II Exempt	ion Petition	Evalu	ation of A	utomatic	Other Submission
					Cia	(De Nov	ghation o)	542(-)
		iginal Submission		nission	Origin	nal Submis	sion	513(g)
	-	Amenament		ornation	Addit	ional Inform	nation	(describe submission):
		Report						,
		Report Amendment						
Have you used or cited Sta	ndards i	n your submission?	Yes [No (lf	Yes, please	complete	Section I, Pag	e 5)
SECTION B		SUB		CANTORS	PONSOR	-		
Company / Institution Name	2			Establishme	ont Registra	tion Numb	er (if known)	
Lacer Line Limit				To be an	plied fo	r followi	ing cleara	nce of this 510[k]
	cu				pheare		ing cloara	
Division Name (if applicable)				Phone Numb	per (includin	g area code		
				011 44	844 9	80 182	20	
Street Address				FAX Number	r (including a	area code)		
Heath House, Ci	rockh	iam Hill		011 44	844 9	80 182	20	
City				State / Provi	nce	ZIP/Pos	tal Code	Country
Edenbridge				Kent		TN8	6ST	U.K.
Contact Name								
Ian Cobley								
Contact Title				Contact E-m	ail Address			
Group Director				ian@st	rawber	ry-las	er.com	
SECTION C	A		SPONDENT (e.	a consulta	nt. if diffe	erent fron	n above)	
Company / Institution Name	2			Establishm	ont Rogistry	ation Numb	or (if known)	
Company / institution Name				Latabilatiti	chi registre			
Division Name (if applicable)				Phone Numb	per (including	g area code)	
Street Address				FAX Numbe	r (including a	area code)		
City				State / Provi	nce	ZIP/Pos	tal Code	Country
Contact Name				1				1
Contact Title				Contact E-m	ail Address			

SECTION D1 RE	ASON FOR APPLICATION - PMA, PDP, OR H	IDE
 New Device Withdrawal Additional or Expanded Indications Request for Extension Post-approval Study Protocol Request for Applicant Hold Request for Removal of Applicant Hold Request to Remove or Add Manufacturing Site 	 Change in design, component, or specification: Software / Hardware Color Additive Material Specifications Other (specify below) 	Location change: Manufacturer Sterilizer Packager
Process change: Manufacturing Sterilization Packaging Other (specify below) Response to FDA correspondence:	Labeling change: Indications Instructions Performance Shelf Life Trade Name Other (specify below)	
Other Reason (specify)		
SECTION D2	REASON FOR APPLICATION - IDE	Repose to FDA Letter Concerning:
 New Indication Addition of Institution Expansion / Extension of Study IRB Certification Termination of Study Withdrawal of Application Unanticipated Adverse Effect Notification of Emergency Use Compassionate Use Request Treatment IDE Continued Access 	Correspondent / Applicant Design / Device Informed Consent Manufacturer Manufacturing Process Protocol - Feasibility Protocol - Other Sponsor Report submission: Current Investigator Annual Progress Report Site Waiver Report Final	 Conditional Approval Deemed Approved Deficient Final Report Deficient Progress Report Deficient Investigator Report Disapproval Request Extension of Time to Respond to FDA Request Meeting Request Hearing
Other Reason (specify):		
SECTION D3	REASON FOR SUBMISSION - 510(k)	
New Device	Additional or Expanded Indications	Change in Technology
Other Reason (specify):	1	

S	ECTION E	to w	/hich	ADD substantial equivale		DNAI e is cla	- INFC	ORMATIO	ON ON 51	0(K) SUB	MIS	SSION	IS	9	Summarv of.	or statement concerning.
1	OLI		2		1100	3	3			4					S	afety and ef	fectiveness information
5		╉	6			7	,			8						∑ 510 (k) :	summary attached
Int	formation on devices to	whit	° ch ci	ubstantial equivalence	o ic	claim	od (if k	now(n)		Ŭ						510 (k)	statement
	510/	•) N/			e is	Claim		lo or Proprie	otony or Ma	dol	Namo					Man	pufacturor
	570(A) 14		<i>er</i>			Trau				vanie					Wan	lulaciurer
1	K111501				1	I-L	_ipo	Syster	m				1	Chro	or	nogene	ex
							·	,						lecr	٦r	lologie	s Ltd
2					2								2				
3					3								3				
4					4								4				
5					5								5				
6					6								6				
6				PRODUCT											0		
S		r els	accifi	PRODUCT	INI		VIATIC	DN - APPI	LICATIO		J ALL	Al	PLIC	ATION	5		
I		r		stem for ae	ctl	hoti	ic us	-									
-			зу.	scent for de	30	iicu	c us										
	Trade or Proprietary o	Mo	del	Name for This Devic	e								Mode	Number	r		
1	Strawberry											1	ILC)			
2	Strawberry 8	<u>ک</u> (Cre	eam								2	SC				
3												3					
4												4					
5												5					
	A desument sumbers of	fall	nric	r related submission	0 (0	ogord	loop of	outcome				Ŭ					
г. 1	DA document numbers o	2	piic		3	eyaru	1955 01 0	outcomej	4				5				6
K	(122354																
7		8			9				10				11				12
Da	ata Included in Submissi	on			ory	Testin	ng		Animal Tria	s		[Hur	man Trial	s		
S	ECTION G			PRODUCT	LA	SSIF	ICAT	ION - API	PLICATIO	DN.	TO AL	LA	PPLI	CATIO	NS	;	
Pr	roduct Code	C.F	F.R. 1 (Section (if applicable	∍) ∩∩	•					Devic	e Cl	ass				
	Assification Panel	2	1 (CIR 070.34	00	,						lass	1		las	ss II	
G	General and p	las	sti	c surgery d	ev	ices	5					Clas	s III	L] U	Jnc	lassified	
In	dications (from labeling)																
Т	he Low level	La	ise	er model Sti	٦a	wbe	erry	and St	rawbe	err	y an	d	Crea	am ca	an	be us	ed for the
n	on-invasive t	en	np	orary reduc	tio	on i	n wa	aist cir	cumfe	re	nce	by	' the	e disr	u	ption o	f adipocyte
С	ells within the	e f	at	layer for th	le	rele	ease	e of fat	and li	pi	ds fr	on	n th	ese c	e	lls for ı	non-invasive
а	esthetic use.																

<i>Note:</i> Submission of this or 2891a Device Establis	information does not affect the nee hment Registration form.	ed to submit a 2891	FDA Document Number (if known)					
SECTION H	MANUFACTURING / PACK	AGING / STERILIZ	ATION SITES RELATING TO	A SUBMISSION				
Original	FDA Establishment Registration To be applied for follo clearance of this 510[Number Dwing [k]	Manufacturer] Contract Sterilizer] Repackager / Relabeler				
Laser Lipo Lin	nited		Establishment Registration Number To be applied for following clearance of this 510[k]					
Division Name (if applica	ble)		Phone Number (including area cod 011 44 844 980 18	^{le)} 20				
Street Address Heath House,	Crockham Hill		FAX Number (including area code) 011 44 1732 866 231					
^{City} Edenbridge			State / Province Kent	ZIP/Postal CodeCountryTN8 6STU.K.				
Contact Name Ian Cobley		Contact Title Group Direc	tor	Contact E-mail Address ian@strawberry- laser.com				
	EDA Establishment Registration	Number						
Original	T DA Establionment Registration	i di i boi	Manufacturer	Contract Sterilizer				
Add Delete			Contract Manufacturer	_ Repackager / Relabeler				
Company / Institution Na	ne		Establishment Registration Numbe	r				
Division Name (if applica	ble)		Phone Number (including area cod	le)				
Street Address			FAX Number (including area code)					
City			State / Province	ZIP/Postal Code Country				
Contact Name		Contact Title		Contact E-mail Address				
	EDA Establishment Devictor	Number						
Original	T DA ESTADIISHIITIENT REGISTRATION	NUTIDEI	Manufacturer	Contract Sterilizer				
Add Delete			Contract Manufacturer	Repackager / Relabeler				
Company / Institution Na	me		Establishment Registration Numbe	r				
Division Name (if applica	ble)		Phone Number (including area cod	le)				
Street Address			FAX Number (including area code)					
City			State / Province	ZIP/Postal Code Country				
Contact Name		Contact Title		Contact E-mail Address				

SEC	TIONI		UTILIZATION OF STANDARDS		
Note: state	Complete this sectior ment.	n if your application or	submission cites standards or includes a "Declaration of Confo	rmity to a Recognized	Standard"
	Standards No.	Standards Organization	Standards Title	Version	Date
1	62304	IEC	Medical Device Software, Software Life-cycle Processes		2006
	Standards No.	Standards Organization	Standards Title	Version	Date
2	60601	IEC	Medical Electrical Equipment, General Requirements for Safety	part 1	2006
	Standards No.	Standards Organization	Standards Title	Version	Date
3	60601	IEC	Medical Electrical Equipment, General Requirements for safety and essential performance, Collateral standard, Electromagnetic compatibility and tests	parts 1-2	2007
	Standards No.	Standards Organization	Standards Title	Version	Date
4	60601	IEC	Medical electrical equipment, Particular Requirements for safety. Specification for Diagnosis and Therapeutic Laser Equipment.	parts 2-22	1996
	Standards No.	Standards Organization	Standards Title	Version	Date
5	60825	IEC	Safety of laser products, Equipment classification and Requirements	part 1	2007
	Standards No.	Standards Organization	Standards Title	Version	Date
6	14971	ISO	Application of Risk Management to Medical Devices		2009

Department of Health and Human Services Food and Drug Administration Office of the Chief Information Officer 1350 Piccard Drive, Room 400. Rockville, MD 20850

Biological evaluation of Medical

Devices, Evaluation and Testing

Please include any additional standards to be cited on a separate page.

Public reporting burden for this collection of information is estimated to average 0.5 hour per response, including the time for reviewing instructions, searching existing data sources, gathering and maintaining the data needed, and completing reviewing the collection of information. Send comments regarding this burden estimate or any

An agency may not conduct or sponsor, and a person is not required to respond to, a collection of information unless it displays a currently valid OMB control number.

other aspect of this collection of information, including suggestions for reducing this burden to:

Standards

ISO

Organization

Standards Title

Standards No.

10993

7

Date

2009

Version

part 1

UTILIZATION OF STANDARDS

Note state	: Complete this section ment.	on if your application	or submission cites standards or includes a "Declaration of Co	onformity to a Recog	nized Standard"
	Standards No.	Standards Organization	Standards Title	Version	Date
8	13485	ISO	Quality Management Systems, Requirements for Regulatory Purposes		2003
9	Standards No.	Standards Organization	Standards Title	Version	Date
	Standards No.	Standards Organization	Standards Title	Version	Date
10					
11	Standards No.	Standards Organization	Standards Title	Version	Date
11					
	Standards No.	Standards Organization	Standards Title	Version	Date
12					
	Standards No.	Standards Organization	Standards Title	Version	Date
13					
	Standards No.	Standards Organization	Standards Title	Version	Date
14					
		Please	include any additional standards to be cited on a separa	te page.	
Publi data other	ic reporting burden for sources, gathering and aspect of this collectio	or this collection of inf maintaining the data n n of information, include	ormation is estimated to average 0.5 hour per response, including t eeded, and completing reviewing the collection of information. So ling suggestions for reducing this burden to:	he time for reviewing i end comments regardin	nstructions, searching existing g this burden estimate or any
1			Department of Health and Human Services		

Department of Health and Human Services Food and Drug Administration Office of the Chief Information Officer 1350 Piccard Drive, Room 400. Rockville, MD 20850

An agency may not conduct or sponsor, and a person is not required to respond to, a collection of information unless it displays a currently valid OMB control number.

DEPARTMENT OF HEALTH AND HUMAN SERVICES	Form Approved: OMB No. 0910-0396 Expiration Date: August 31, 2012
CERTIFICATION: FINANCIAL INTERESTS AND	
ARRANGEMENTS OF CLINICAL INVESTIGATORS	
TO BE COMPLETED BY APPLICANT	
With respect to all covered clinical studies (or specific clinical studies liste support of this application, I certify to one of the statements below a certification is made in compliance with 21 CFR part 54 and that for the investigator includes the spouse and each dependent child of the investiga	ed below (if appropriate)) submitted in s appropriate. I understand that this purposes of this statement, a clinical ator as defined in 21 CFR 54.2(d).
Please mark the applicable checkbox.	
(1) As the sponsor of the submitted studies, I certify that I have not environment with the listed clinical investigators (enter names of clinical investigator construction) whereby the value of compensation to the investigator construction and the sponsor whether the investigator had a proprietary interest in the sponsor as defined in 21 CFR 54.2(a) did not disclose any solution is the sponsor was the recipient of significant payments of other sponsor of the sponsor was the recipient of significant payments of the sponsor of the sponsor was the recipient of significant payments of the sponsor was the recipient of significant payments of the sponsor of the sponsor was the recipient of significant payments of the sponsor of the sponsor was the recipient of significant payments of the sponsor of the sponsor was the recipient of significant payments of the sponsor of the sponsor was the recipient of significant payments of the sponsor of the sponsor was the recipient of significant payments of the sponsor of the sponsor was the recipient of significant payments of the sponsor of the sponsor was the recipient of significant payments of the sponsor was the recipient of significant payments of the sponsor	antered into any financial arrangement gators below or attach list of names to buld be affected by the outcome of the linical investigator required to disclose in this product or a significant equity in such interests. I further certify that no r sorts as defined in 21 CFR 54.2(f).
Clinical Investigators	
(2) As the applicant who is submitting a study or studies sponsor applicant, I certify that based on information obtained from the investigators, the listed clinical investigators (attach list of names financial arrangement with the sponsor of a covered study where investigator for conducting the study could be affected by the ou CFR 54.2(a)); had no proprietary interest in this product or signif the covered study (as defined in 21 CFR 54.2(b)); and was not th other sorts (as defined in 21 CFR 54.2(f)).	ed by a firm or party other than the sponsor or from participating clinical to this form) did not participate in any eby the value of compensation to the stcome of the study (as defined in 21 icant equity interest in the sponsor of the recipient of significant payments of
(3) As the applicant who is submitting a study or studies sponsore applicant, I certify that I have acted with due diligence to obtain (attach list of names) or from the sponsor the information required do so. The reason why this information could not be obtained is attached.	ed by a firm or party other than the n from the listed clinical investigators under 54.4 and it was not possible to tached.
NAME	
IAN COBLEY DIREC	ror
FIRM/ORGANIZATION	
LASER LIPOLTD	
SIGNATURE	DATE (mm/dd/yyyy) 28/02/2013
L 4. '	
Paperwork Reduction Act Statement An agency may not conduct or sponsor, and a person is not required to respond to, a collection or information unless it displays a currently valid OMB control number. Public reporting burden for thi collection of information is estimated to average 1 hour per response, including time for reviewing instructions, searching existing data sources, gathering and maintaining the necessary data, and completing and reviewing the collection of information. Send comments regarding this burden estimat or any other aspect of this collection of information to the address to the right:	f Department of Health and Human Services Food and Drug Administration Office of Chief Information Officer 1350 Piccard Drive, 420A Rockville, MD 20850

Department of Plach IT AND HUMAN SERVICES Food and Drug Administration Food and Drug Administration Certification of Compliance, under 42 U.S.C. § 282(j)(£ Requirements of ClinicalTrials.gov Data Bank (42 U.S. (For submission with an application/submission, including amendments, supplements, and resubmissions, Federal Food, Drug, and Cosmetic Act or § 351 of the Public Health Service Act.) SPONSOR / APPLICANT / SUBMITTER INFORMATIC SPONSOR / APPLICANT / SUBMITTER INFORMATIC 1. NAME OF SPONSOR/APPLICANT/SUBMITTER LASER LIPO LTD 2. DA U.A ADDRESS (Number, Street, State, and ZIP Code) HEATH HOUSE, CROCKMAM HIM, EDANDIDGS. HEATH HOUSE, CROCKMAM HIM, EDANDIDGS. Street, State, and ZIP Code) HEATH HOUSE, CROCKMAM HIM, EDANDIDGS. HENTIC NOT DNSCH / APPLICANT / SUBMITTER LASER LIPO LIPCOMENTING STRAWBERRY MA CREAM STRAWBERRY MA CREAM	5)(B), with C. § 282(j)) under §§ 505, 515, 520(m), or 510(k) of the N TE OF THE APPLICATION/SUBMISSION HICH THIS CERTIFICATION ACCOMPANIES AJJA44 51, 2013 LEPHONE AND FAX NUMBERS clude Area Code) el.) 0044 844 980 /820 ax) Blood/Cellular/Gene Therapy Product Name(s) me(s) and/or Model Number(s)
Certification of Compliance, under 42 U.S.C. § 282(j)(4 Requirements of ClinicalTrials.gov Data Bank (42 U.S. (For submission with an application/submission, including amendments, supplements, and resubmissions, Federal Food, Drug, and Cosmetic Act or § 351 of the Public Health Service Act.) SPONSOR / APPLICANT / SUBMITTER INFORMATIC NAME OF SPONSOR/APPLICANT/SUBMITTER LASEEL LIPO LTD 3. ADDRESS (Number, Street, State, and ZIP Code) HEATH HOUSE, CROCK NAM HIM, EDANDIDGO, HEATH HOUSE, CROCK NAM HIM, EDANDIDGO, OPODUCT INFORMATION STEAWBERRY and CREAM STEAWBERRY and CREAM STEAWBERRY and CREAM	5)(B), with C. § 282(j)) under §§ 505, 515, 520(m), or 510(k) of the N TE OF THE APPLICATION/SUBMISSION HICH THIS CERTIFICATION ACCOMPANIES ALL ALL ALL ALL ALL ALL ALL ALL ALL ALL
(For submission with an application/submission, including amendments, supplements, and resubmissions, Federal Food, Drug, and Cosmetic Act or § 351 of the Public Health Service Act.) SPONSOR / APPLICANT / SUBMITTER INFORMATIC 1. NAME OF SPONSOR/APPLICANT/SUBMITTER 2. DA LASER LIPO LTD 3. 3. ADDRESS (Number, Street, State, and ZIP Code) 4. TE HENTH HOUSE, CROCKHAM HILL, EDANDOGO, TNN & 65T. U.K. (In PRODUCT INFORMATION (In 5. FOR DRUGS/BIOLOGICS: Include Any/All Available Established, Proprietary and/or Chemical/Biochemical/E For Devices: Include Any/All Common or Usual Name(s), Classification, Trade or Proprietary or Model Nam (Attach extra pages as necessary) STRAWBERRY JLO	under §§ 505, 515, 520(m), or 510(k) of the N TE OF THE APPLICATION/SUBMISSION HICH THIS CERTIFICATION ACCOMPANIES AJARY 31, 2013 LEPHONE AND FAX NUMBERS clude Area Code) el.) 0044 844 980 /820 ax) Blood/Cellular/Gene Therapy Product Name(s) me(s) and/or Model Number(s)
SPONSOR / APPLICANT / SUBMITTER INFORMATIC 1. NAME OF SPONSOR/APPLICANT/SUBMITTER 2. DA LASER LIPO LTD 3. 3. ADDRESS (Number, Street, State, and ZIP Code) 4. TE HEATH HOUSE, CROCKHAM HILL, EDANDIDGO. 4. TE (In (In KENT. TN8 6ST. U.K. (T PRODUCT INFORMATION 5. 5. FOR DRUGS/BIOLOGICS: Include Any/All Available Established. Proprietary and/or Chemical/Biochemical/E FOR DEVICES: Include Any/All Common or Usual Name(s), Classification, Trade or Proprietary or Model Nam (Attach extra pages as necessary) STRAUBERRY AD CREAM JLD	DN TE OF THE APPLICATION/SUBMISSION HICH THIS CERTIFICATION ACCOMPANIES AJA2431, 2013 LEPHONE AND FAX NUMBERS clude Area Code) el.) 0044844980 /820 ax) Blood/Cellular/Gene Therapy Product Name(s) ne(s) and/or Model Number(s)
1. NAME OF SPONSOR/APPLICANT/SUBMITTER 2. DA LASER LIPO LTD J 3. ADDRESS (Number, Street, State, and ZIP Code) 4. TE HEATH HOJSE, CROCKHAM HILL, EDANDIDGO. 4. TE KENTIL TONSE, CROCKHAM HILL, EDANDIDGO. (In FOR DRUGS/BIOLOGICS: Include Any/All Available Established, Proprietary and/or Chemical/Biochemical/	TE OF THE APPLICATION/SUBMISSION HICH THIS CERTIFICATION ACCOMPANIES AJJA24 31, 2013 LEPHONE AND FAX NUMBERS clude Area Code) el.) 0044 844 980 /820 ax) Blood/Cellular/Gene Therapy Product Name(s) me(s) and/or Model Number(s)
LASER LIPO LTD 3. ADDRESS (Number, Street, State, and ZIP Code) HEATH HOJSE, CROCKHAM HIM, EDENDIDGE. KENT. TNB 6ST. U.Y. (In PRODUCT INFORMATION 5. FOR DRUGS/BIOLOGICS: Include Any/All Available Established, Proprietary and/or Chemical/Biochemical/Bio	ALJARY 31, 2013 LEPHONE AND FAX NUMBERS clude Area Code) el.) 0044 844 980 /820 ax) Blood/Cellular/Gene Therapy Product Name(s) me(s) and/or Model Number(s)
3. ADDRESS (Number, Street, State, and ZIP Code) HEATH HOUSE, CROCKHAM HILL, EDANDIDGA. KENT. TNB 6ST. U.K. (In (In (In (In (In (In (In (In	LEPHONE AND FAX NUMBERS clude Area Code) el.) 0044 844 980 /820 ax) Blood/Cellular/Gene Therapy Product Name(s) ne(s) and/or Model Number(s)
HEATH HOUSE, CROCKHAM HILL, EDENDIDGE. (In KENTI TNS 6ST. U.K. (I (I (I (F PRODUCT INFORMATION 5. FOR DRUGS/BIOLOGICS: Include Any/All Available Established, Proprietary and/or Chemical/Biochemical/E (II FOR DEVICES: Include Any/All Common or Usual Name(s), Classification, Trade or Proprietary or Model Name (Attach extra pages as necessary) ILO STRAUBERRY and CREAM SC	clude Area Code) el.) 0044 844 980 /820 ax) Blood/Cellular/Gene Therapy Product Name(s) ne(s) and/or Model Number(s)
KENT. TNB 6ST. U.Y. (T (F PRODUCT INFORMATION 5. FOR DRUGS/BIOLOGICS: Include Any/All Available Established, Proprietary and/or Chemical/Biochemical/E For DEVICES: Include Any/All Common or Usual Name(s), Classification, Trade or Proprietary or Model Nar (Attach extra pages as necessary) STRAWBERRY ILO STRAWBERRY SC	el.) 0044 844 980 /820 ax) Blood/Cellular/Gene Therapy Product Name(s) me(s) and/or Model Number(s)
(F PRODUCT INFORMATION 5. FOR DRUGS/BIOLOGICS: Include Any/All Available Established, Proprietary and/or Chemical/Biochemical/E FOR DEVICES: Include Any/All Common or Usual Name(s), Classification, Trade or Proprietary or Model Nar (Attach extra pages as necessary) STRAWBERRY and CREAM SC	ax) Blood/Cellular/Gene Therapy Product Name(s) ne(s) and/or Model Number(s)
PRODUCT INFORMATION 5. FOR DRUGS/BIOLOGICS: Include Any/All Available Established, Proprietary and/or Chemical/Biochemical/E FOR DRUGS/BIOLOGICS: Include Any/All Available Established, Proprietary and/or Chemical/Biochemical/E FOR DRUGS/BIOLOGICS: Include Any/All Common or Usual Name(s), Classification, Trade or Proprietary or Model Nar (Attach extra pages as necessary) STRAWBERRY STRAWBERRY and CREAM SC	Blood/Cellular/Gene Therapy Product Name(s) ne(s) and/or Model Number(s)
5. FOR DRUGS/BIOLOGICS: Include Any/All Available Established, Proprietary and/or Chemical/Biochemical/E FOR DEVICES: Include Any/All Common or Usual Name(s), Classification, Trade or Proprietary or Model Nar (Attach extra pages as necessary) STRAWBERRY STRAWBERRY STRAWBERRY SC	Blood/Cellular/Gene Therapy Product Name(s) ne(s) and/or Model Number(s)
STRAWBERRY and CREAM SC	
STRAWBERRY and CREAM SC	
APPLICATION / SUBMISSION INFORMATION	
IND NDA ANDA BLA PMA HDE	() PDP Other
7. INCLUDE IND/NDA/ANDA/BLA/PMA/HDE/510(k)/PDP/OTHER NUMBER (If number previously assigned)	
K130341 K122354	
8. SERIAL NUMBER ASSIGNED TO APPLICATION/SUBMISSION WHICH THIS CERTIFICATION ACCOMPAN	NIES
CERTIFICATION STATEMENT / INFORMATION	
9. CHECK ONLY ONE OF THE FOLLOWING BOXES (See instructions for additional information and explanation	on)
A. I certify that the requirements of 42 U.S.C. § 282(j), Section 402(j) of the Public Health Service 110-85, do not apply because the application/submission which this certification accompanies of	loes not reference any clinical trial.
B. I certify that the requirements of 42 U.S.C. § 282(j), Section 402(j) of the Public Health Service	ce Act, enacted by 121 Stat. 823, Public Law
110-85, do not apply to any clinical trial referenced in the application/submission which this cert	ification accompanies.
110-85, apply to one or more of the clinical trials referenced in the application/submission v	which this certification accompanies and that
those requirements have been met.	Colored Colore
 IF YOU CHECKED BOX C, IN NUMBER 9, PROVIDE THE NATIONAL CLINICAL TRIAL (NCT) NUMBER(S) UNDER 42 U.S.C. § 282(i)(1)(A)(i), SECTION 402(i)(1)(A)(i) OF THE PUBLIC HEALTH SERVICE / 	ACT, REFERENCED IN THE APPLICATION/
SUBMISSION WHICH THIS CERTIFICATION ACCOMPANIES (Attach extra pages as necessary)	
NCT Number(s):	
The undersigned declares, to the best of her/his knowledge, that this is an accurate, true, and complete su	bmission of information. I understand that the
of a false certification under such section are prohibited acts under 21 U.S.C. § 331, section 301 of the Fed	eral Food, Drug, and Cosmetic Act.
Warning: A willfully and knowingly false statement is a criminal offense, U.S. Code, title 18, section 1001.	
11. SIGNATURE OF SPONSOR/APPLICANT/SUBMITTER OF AN 12. NAME AND TITLE OF THE AUTHORIZED REPRESENTATIVE (SIGN)	E PERSON WHO SIGNED IN NO. 11
(Name) IAN C	20869
(an totaling	
(Title) DiRECT	01C
13. ADDRESS (Number, Street, State, and ZIP Code) (of person identified in Nos. 11 and 12) 14. TELEPHONE AND FAX N (Include Area Code)	UMBERS 15. DATE OF CERTIFICATION
HERTH HOUSE, CROCKHAM HILL, (Tel.) 0044 844	+ 980 132 28/02/2013
EDENBRIDGE. NENI . NO OSI. OT	

Form FDA 3674 (3/12) (FRONT)

PSC Publishing Services (301) 443-6740 EF

Laser Lipo Ltd Heath House

Crockham Hill Edenbridge

United Kingdom, TN8 6ST Telephone: +44 844 980 1820 July 26, 2013

FDA CDRH DMC

JUL 2 9 2013

Received



July 26, 2013

e-mail: ian@strawberry-laser.com Food and Drug Administration Center for Devices and Radiological Health

Document Mail Center - WO66-G609 10903 New Hampshire Avenue Silver Spring Maryland 20993-0002 United States of America

Ref: Additional Information - K130341/S1

Dear Document Control Clerk:

In reference to our ongoing application for a 510k we would like to provide additional information to our application K130341/S1

Additional information comprises of our study protocol.

E Copy Statement:

The E Copy provided within our submission is an exact duplicate of the paper copy.

Owner:

This **Additional Information - K130341/S1** is owned by Laser Lipo Ltd.

Submitter:

ſ

This Additional Information is submitted by Ian Cobley Group Director For and on behalf of: Laser Lipo Ltd

Contact Persons:

The address and telephone number for **<u>all</u>** correspondence is:

Ian Cobley Laser Lipo Ltd Heath House Crockham Hill Edenbridge Kent TN8 6ST United Kingdom Tel: 011 44 844 980 1820 Mobile: 011 44 777 445 9611 e-mail: ian@strawberry-laser.com

Please find enclosed our E copy submission, along with one paper copy.

If I can be of any further assistance, please do not hesitate to contact me.

Sincerely B.

Ian Cobley Group Director Laser Lipo Ltd

3

1. Additional Information (S1) Cover Letter

Laser Lipo Ltd Heath House Crockham Hill Edenbridge United Kingdom, TN8 6ST Telephone: +44 844 980 1820 e-mail: ian@strawberry-laser.com



July 26, 2013

Food and Drug Administration Center for Devices and Radiological Health Document Mail Center - WO66-G609 10903 New Hampshire Avenue Silver Spring Maryland 20993-0002 United States of America

Ref: Additional Information - K130341/S1

Dear Document Control Clerk:

In reference to our ongoing application for a 510k we would like to provide additional information to our application K130341/S1

Additional information comprises of our study protocol.

E Copy Statement:

The E Copy provided within our submission is an exact duplicate of the paper copy.

Owner:

This **Additional Information - K130341/S1** is owned by Laser Lipo Ltd.

Submitter:

This Additional Information is submitted by Ian Cobley Group Director For and on behalf of: Laser Lipo Ltd

Contact Persons:

The address and telephone number for <u>all</u> correspondence is:

Ian Cobley Laser Lipo Ltd Heath House Crockham Hill Edenbridge Kent TN8 6ST United Kingdom Tel: 011 44 844 980 1820 Mobile: 011 44 777 445 9611 e-mail: ian@strawberry-laser.com

Please find enclosed our E copy submission, along with one paper copy.

If I can be of any further assistance, please do not hesitate to contact me.

Sincerely,

Ian Cobley Group Director Laser Lipo Ltd Dr Sankar Basu, Ph.D Senior Physicist Food and Drug Administration Center for Devices and Radiological Health Document Mail Center - WO66-G609 10903 New Hampshire Avenue Silver Spring Maryland 20993-0002



Re: Response for K130341/S1

26th July 2013.

Dear Dr Basu

b)(4) Trade Secret Process - Testing

Yours sincerely

lan B. Cobley Director.

... continuing to lead the market with innovation.

Laser Lipo Ltd. "Heath House" Crockham Hill Edenbridge Kent. TN8 6ST. United Kingdom. T. 0044 844 980 1820. W. strawberry-laser.com Co. Reg. No. 06308992. VAT Reg. No. 891 7315 01
